Solar Geometry in Performance of the Built Environment

An Integrated Computational Design Method for High-Performance Building Massing Based on Attribute Point Cloud Information

Miktha Farid Alkadri

Solar Geometry in Performance of the Built Environment

An Integrated Computational Design Method for High-Performance Building Massing Based on Attribute Point Cloud Information

Miktha Farid Alkadri

A+BE | Architecture and the Built Environment | TU Delft BK

21#13

Design | Sirene Ontwerpers, Véro Crickx

ISBN 978-94-6366-421-9 ISSN 2212-3202

© 2021 Miktha Farid Alkadri

This dissertation is open access at https://doi.org/10.7480/abe.2021.13

Attribution 4.0 International (CC BY 4.0)

This is a human-readable summary of (and not a substitute for) the license that you'll find at: https://creativecommons.org/licenses/by/4.0/

You are free to: Share — copy and redistribute the material in any medium or format Adapt — remix, transform, and build upon the material for any purpose, even commercially. This license is acceptable for Free Cultural Works. The licensor cannot revoke these freedoms as long as you follow the license terms.

Under the following terms:

Attribution — You must give appropriate credit, provide a link to the license, and indicate if changes were made. You may do so in any reasonable manner, but not in any way that suggests the licensor endorses you or your use.

Unless otherwise specified, all the photographs in this thesis were taken by the author. For the use of illustrations effort has been made to ask permission for the legal owners as far as possible. We apologize for those cases in which we did not succeed. These legal owners are kindly requested to contact the author.

Solar Geometry in Performance of the Built Environment

An Integrated Computational Design Method for High-Performance Building Massing Based on Attribute Point Cloud Information

Dissertation

for the purpose of obtaining the degree of doctor at Delft University of Technology, by the authority of the Rector Magnificus Prof.dr.ir. T.H.J.J. van der Hagen, chair of the Board for Doctorates to be defended publicly on Monday, 14 June 2021 at 15.00 o'clock

by

Miktha Farid ALKADRI Master of Architecture, University of Indonesia, Indonesia born in Sinjai, Indonesia This dissertation has been approved by the promotors.

Composition of the doctoral committee:

Rector Magnificus,	chairperson
Prof. dr. ir. arch. I.S. Sariyildiz	Delft University of Technology, promotor
Dr. M. Turrin	Delft University of Technology, copromotor

Independent members:

Prof. dr. ir. P.J.M van Oosterom	Delft University of Technology
Prof. dr. G. Vrachliotis	Delft University of Technology
Dr. ir. M.J. Tenpierik	Delft University of Technology
Prof. drIng. ir. Widjaja Martokusumo	Bandung Institute of Technology
Dr. F. De Luca	Tallinn University of Technology,
	other member

This research was funded by Indonesia Endowment Fund for Education (LPDP), PK-54

Not having learned it is not as good as having learned it; having learned it is not as good as having seen it carried out; having seen it is not as good as understanding it; understanding it is not as good as doing it.

Xunzi

TOC

Acknowledgements

Bismillahi Rahmani Rahim.

This thesis represents not only a milestone of my scientific work but also the lessons learned from an amazing experience in the Department of Architecture Engineering and Technology, TU Delft. Over the past five years, I have received enormous supports from many remarkable individuals including families, friends, colleagues, supervisors, and also specifically from the Indonesia Endowment Fund for Education (LPDP) which has provided me with full PhD funding. This thesis would never have been possible without their generous help.

First and foremost, I would like to express my sincere gratitude to my academic supervisors, Prof.dr.ir.arch. I.S. Sariyildiz and dr. Michela Turrin, MSc., for giving me an exciting opportunity to be part of the Design Informatics Research Group. For my promotor, Prof.dr.ir.arch. Sevil Sariyildiz, I greatly appreciate your trust, time, energy, moral support, and the freedom you have given me to explore new ideas during my PhD studies. You have created an invaluable space for me to do this research and develop myself as a PhD researcher in the best possible way. Your concern for work-life balance always encourages me to enjoy the journey of the PhD process. For my copromotor, Dr. M. Turrin, I am deeply thankful for your continuous guidance and heart-warming support since day one of my PhD research. Your productivity and outstanding attitude in research and education will always motivate me in academia.

The same sincere thanks also goes to Dr. Francesco de Luca, for showing remarkable open-mindedness and excellent collaboration. I truly hope that we will be given the opportunity to continue our research cooperation in the future. I also want to take a moment to thank the other defense committee members, Prof.dr.ir.P.J.M van Oosterom, Prof.dr. G. Vrachliotis, Dr.ir. M.J. Tenpierik, and Prof.dr.-Ing.ir Widjaja Martokusumo, for investing time to read my PhD thesis.

Special thanks to Valentini Vanhecke from 4Visualization, Marnix van der Wolk from Faro Benelux B.V., Dr. Irwan Gumilar and his teams from Geodesy and Geomatics Engineering, ITB, and Ir. Haidar from PT Asaba, who has helped me compile the dataset used in this PhD study. In parallel, my heartfelt thanks to Dr.(cand) Daliana and Dr. Florian from SHAU, and Dr. Hendrajaya Isnaeni from PT Arkitekton for providing me access to use their project as case studies in this PhD Thesis. Some people of outstanding importance during my PhD journey. First, my fellow PhDs at the Design Informatics Chair, Berk, Frank, Ding, and Cemre who share not only the same struggles and enthusiasms about PhD life, but also great discussions, research debates, and exchange of ideas during coffee breaks. Thanks to Frank for being such a good partner both in the workspace and badminton, as well as Berk who always inspires me about hard work, and Serdar who is always positive with new research collaboration ideas. Second, my PhD colleagues in the AET Department who create a pleasant work environment: Dadi, Tiantian, Tatiana, Nick, Marco, Phan, Zoheir, Minyoung, Lia, Nan Bai, Michiel and other great colleagues. Also, thanks to Ahmed for always inviting me to your beautiful home and Yu-Chou Chiang for assisting me with technical details during the dataset processing. Third, Osman and James from the Urbanism Editing Team who help me to proofread the journal manuscripts. Fourth, special thanks to all AET secretaries for always helping me when missing some crucial information during my PhD studies.

And then of course my Indonesian colleagues and friends. I will begin to thank to Daniel who has been my partner-in-crime since the LPDP training camp – PK54. You have been such a good friend and thank you for working with me on creating the Nulis Jurnal channel. Next are my beloved neighbors, Mas Nabriz and Tika. It is a wonderful and fortunate experience to have you on our rooftop, not only because of your support but also your generosity. I would also like to thank other Indonesian communities such as Indonesian PhDs in Delft, Mas Nasikun, Mas Sebri, and other KMD (Keluarga Muslim Delft) members, friends from ISR (Indonesische Stichting Rotterdam) and PPMR (Persatuan Pelajar Muslim Rotterdam), Krantenbezorger colleagues, and Bu Wiwi from Liveable Cities Taskforce for Indonesian Diaspora Networks. Thank you all for bringing happiness and colorful memories during my stay in Rotterdam and Delft.

Finally, I dedicate this last paragraph to thank my lovely family – my parents, Ibu Nurhayati and Bapak Amiruddin, in-laws, Ibu Sri widyastuti and Bapak Iskandar, my brothers, Muh. Fauzan Azhari and Dedi Permadi, and in-laws, Mas Arif and Fachri. Your unconditional love, prayers, and support have truly made me alive and stronger in every step of my PhD life. Also, deepest thanks to the rest of my family, my grandparents, uncles, aunts, and cousins, your support always shines through on my PhD days. Last but not least, a very special thanks to my lovely wife, Noor Fajrina Farah Istiani and my precious daughter, Nouza Kieana Atka. Thank you for always being by my side, supporting me in any condition, healing my anxieties, and saying my name in your every prayers. I am nothing without you.

Alhamdulillah.

Thank you – Dank u wel – Terima kasih.

TOC

Contents

List of Tables 16 List of Figures 17 Summary 19 Samenvatting 23

1 Introduction 27

- 1.1 General Background 27
- 1.2 **Problem Statement** 29
- 1.3 Research questions, aims and objectives 30
- 1.3.1 Research questions 30
- 1.3.2 Aim 31
- 1.3.3 Objectives 31

1.4 **Research output** 32

1.5 Research method 33

1.6 **Research relevance** 37

- 1.6.1 Scientific relevance 37
- 1.6.2 Societal relevance 37

2 Review Of Computational Solar Envelopes 41

- 2.1 Introduction 43
- 2.2 Scope and Method of the Review 45
- 2.3 Review findings: Computational methods and parameters of solar envelopes 47
- 2.3.1 Design methods 48
- 2.3.2 Design parameters 50
- 2.3.3 Comparative analysis of design parameters in relation to design methods 50

2.4 Discussion: digital simulation tools and case studies 59

2.4.1 Digital tools 62

2.4.2 Case studies 65

- 2.5 Knowledge gaps and new directions 67
- 2.6 Conclusion 69

3 Preliminary Simulation Of Solar Envelopes Based On Geometric Properties Of Point Cloud 79

- 3.1 Introduction 82
- 3.1.1 General background 82
- 3.1.2 Solar envelopes 83
- 3.1.3 Point cloud data 85
- 3.2 Proposed procedure: From point clouds to the solar envelopes and design analysis 87
- 3.2.1 Data collection 88
- 3.2.2 Generation of the solar envelopes 88
- 3.2.3 Design analysis 89
- 3.3 Case study: The Kruisplein area in Rotterdam, The Netherlands 89

3.4 Data collection 90

- 3.4.1 Data conversion 91
- 3.4.2 Computing environment 92
 - 3.5 Development of the design framework for solar envelopes 92

3.6 Results and discussion 95

- 3.6.1 Simulation of the existing contexts 96
- 3.6.2 Simulation of the proposed design 99
- 3.7 Concluding remarks 101

4 Exploring Potential Applications Of Geometric And Radiometric Information Stored In Point Cloud Data 107

4.1 Introduction 110

4.2 Theoretical background 112

- 4.2.1 Point cloud data 112
- 4.2.2 Correction of the raw dataset 114
- 4.2.3 Material properties 115
- 4.2.4 Solar radiation analysis 118

4.3 Development of the computational design method 119

- 4.3.1 Correction of the raw dataset 121
 - 4.4 **Dataset collection** 129

4.5 Study findings and discussion 130

- 4.5.1 Intensity correction 130
- 4.5.2 Surface distribution of material properties 134
- 4.5.3 Simulation of solar radiation 136
- 4.5.4 Material selection with integrated insolation values 137
 - 4.6 Concluding remarks and future perspectives 141

5 Computational Models For Constructing Solar Geometry Based On Attribute Point Cloud Information 149

5.1 Introduction 154

5.2 Theoretical background 157

- 5.2.1 Basic principle of solar envelopes 157
- 5.2.2 Point cloud data 160
- 5.2.3 Subtractive solar envelopes 162
- 5.3 A proposed method for new subtractive solar envelopes 164

- 5.4 The implementation of the case study 167
- 5.4.1 Dataset collection 167
- 5.4.2 The computational design process 169
- 5.4.2.1 Step 01 169
- 5.4.2.2 Step 02 174
- 5.4.2.3 Step 03 174
- 5.4.2.4 Step 04 176

5.5 Results and Discussion 179

- 5.5.1 The final geometry of solar envelopes 179
- 5.5.2 Insolation analysis for solar envelopes 181
- 5.5.3 Glare analysis simulation 182
- 5.5.4 The geometric configuration of solar envelopes based on glare simulation 184

5.6 **Conclusion and Future Recommendations** 186

5.7 Introduction 194

- 5.7.1 General background 194
- 5.7.2 Related works 196
- 5.7.2.1 Solar envelopes 196
- 5.7.2.2 Shading envelopes 197
- 5.7.2.3 Subtractive solar envelopes 199
- 5.7.2.4 Subtractive solar envelopes based on point cloud data (SOLEN) 200

5.8 Proposed methods for subtractive shading envelopes 202

- 5.8.1 Stage A Input 203
- 5.8.1.1 Step A Preparation of Input Parameters 203
- 5.8.2 Stage B Simulation Process 204
- 5.8.2.1 Step B Dataset preparation 205
- 5.8.2.2 Step C Selection criteria 208
- 5.8.2.3 Step D Form generation process of shading envelopes 209
- 5.8.3 Stage C Output 212
- 5.8.3.1 Step E Form Generation Process of Self-Shading Envelopes 212

5.9 Dataset collection 214

5.10 Results and discussion 217

- 5.10.1 Step A Input parameters 217
- 5.10.2 Step B Dataset preparation 218
- 5.10.3 Step C Selection criteria 222
- 5.10.4 Step D Form generation process of shading envelopes 224
- 5.10.5 Step E Form generation process of self-shading envelopes 226

5.11 Conclusion 229

6 Conclusions 235

- 6.1 Introduction 235
- 6.2 Reflections from design practitioners 236
- 6.3 Answer to the research questions 238
- 6.3.1 Sub-research questions 238
- 6.3.2 Main research-question 242
 - 6.4 Limitation of the research 244
 - 6.5 Future recommendations 245

Curriculum Vitae 253 List of Publications 255

List of Tables

- 2.1 Database of design parameters and computational method of solar envelopes based on the selected references 52
- 2.2 Database of the computational environment parameters of solar envelopes based on the selected references 60
- 2.3 Knowledge gaps and new directions for solar envelopes 67
- 3.1 Data processing workflow of 3D point cloud data 91
- 3.2 Simulation results based on the existing context 97
- 3.3 Simulation results of the proposed design 100
- 4.1 Material database for the outdoor building context (collected from various sources) 125
- 5.1 Technical specifications of the tools [99] 168
- 5.2 Detailed specifications of 3D scanner [71] 215

List of Figures

- 1.1 Overview of research structure and methods 34
- 2.1 General framework of the review 47
- 2.2 Distribution of design parameters according to corresponding computational methods and selected references 54
- 2.3 Categorization of design parameters and corresponding methods based on usage frequency 55
- 2.4 Specific distribution of design parameters (geographic and climatic properties) based on the group usage frequency and corresponding methods 57
- 2.5 Distribution of computational environment parameters according to the corresponding computational methods 62
- 3.1 Design mechanism of solar envelopes, generated from: A. daily time limits and B. annual time limits [6] 84
- 3.2 Attribute information of point cloud data [40] quality. 86
- 3.3 Overview of the proposed procedure 87
- 3.4 Geometric representation of the selected site in different 3D models 90
- 3.5 Selected parameters for solar envelope 94
- 3.6 Computational workflow of solar envelope frequency 94
- 3.7 Comparative indicators of the simulation context 96
- 3.8 Comparison of the solar point vectors 98
- 4.1 Overview of the computational workflow 120

- 4.2 Correction of the raw dataset 3D models 121
- 4.3 Material exploration of point cloud data 123
- 4.4 Solar radiation simulation workflow 126
- 4.5 Calculation of solar irradiance (INC: angle of incidence, G: global irradiance) [87] 127
- 4.6 Integrated simulation workflow 128
- 4.7 The selected 3D colour point cloud dataset 129
- 4.8 Distribution of points according to the angle of incidents 131
- 4.9 Distribution of points according to the standard deviations of different cos values 131
- 4.10 Comparison of the (A) Original dataset and(B) Corrected intensity 132
- 4.11 Surface distribution catalogue of material properties 135
- 4.12 Simulation of solar radiation 136
- 4.13 Material selection according to optical and thermal properties 138
- 4.14 Material selection according to insolation value simulation results 140
- 5.1 A design mechanism for establishing solar envelopes, generated from A. Daily time limits and B. Annual time limits. These diagrams are elaborated from a book of Sun, Rhythm, and Form, authored by Knowles [5]. It specifically demonstrates the time setting before 9 am, and after 3 pm at 34° north (Los Angeles, USA). 158

- 5.2 The design mechanism of existing subtractive solar envelopes. These diagrams are elaborated from the paper of solar form finding, authored by De Luca [9] 163
- 5.3 The computational workflow used in the proposed method of generating new subtractive solar envelopes 165
- 5.4 Dataset collection and the different views captured in relation to the scanning position 168
- 5.5 Dataset preparation A. Outlier removal B. Clipping box 169
- 5.6 The distribution of points on each data scan according to A. Angles of incidence and B. The standard deviation of different cos values 171
- 5.7 Intensity correction on each data scan 172
- 5.8 3D polyhedra of a proposed building 174
- 5.9 Hit-or-miss analysis for each data scan 175
- 5.10 The surface distribution of material properties in Scan_003 178
- 5.11 The final geometry of solar envelopes 180
- 5.12 The insolation analysis of the final solar envelopes 182
- 5.13 A comparison of material properties before and after the sun visibility selection 183
- 5.14 The geometric configuration of solar envelopes according to glare simulation analysis 185
- 5.15 An overview of the proposed computational workflow. 202
- 5.16 Preparation of input parameters. 204
- 5.17 Detailed procedures for the dataset preparation. 206
- 5.18 Selection procedures based on the criteria of sun visibility and material properties. 209
- 5.19 Detailed procedures for the design simulation. 211

- 5.20 Detailed procedures for generating the final output of self-shading envelopes. 213
- 5.21 Dataset collection with different views captured in relation to the scanner position 216
- 5.22 Selected dataset and 3D polyhedra for a case study 218
- 5.23 Dataset correction (A). Point distributions based on cosine values of each incident angle (B). The standard deviation of cosine values based on a group sample of points (C). The standard deviation of the whole dataset population. 219
- 5.24 Intensity correction on each data scan. 221
- 5.25 Transformation of the dataset attributes. 222
- 5.26 The resulting points after performing the selection criteria 223
- 5.27 The geometric configuration of subtractive shading envelopes based on sequential steps of design simulation. 225
- 5.28 Geometric configuration of subtractive shading envelopes 227

Summary

In recent years, urban population growth has increased rapidly in parallel with the escalation of energy consumption from the building sector. The UN SDGs (United Nations Sustainable Development Goals) play an important role in driving strategies such as green building technology and high-performance envelope to manage and calculate the efficiency of energy intensity in buildings. Although these approaches promote promising technical solutions, attempts to improve the environmental quality of the built environments have often been neglected, especially with regard to passive design strategies in the conceptual design phase. This is crucial for architects not only to prevent unexpected failures after a new building is built in a new context but also to understand comprehensively the microclimatic conditions from the design context. Through the computational development of solar geometric models, this thesis explored a new method for designing and analyzing existing environments by making use of attributes point cloud information, and solar and shading envelopes. The developed method not only helps further architects to construct interdependencies between the new building and the local context but also to make informed-design decisions towards high-performed building massing.

However, a state-of-the-art computational method for generating solar geometries poses a major challenge in understanding site characteristics of existing environments. Existing methods predominantly construct 3D digital contextual models based on basic architectural geometric shapes (i.e., solid modelling platforms), currently isolated from properties around the local context (i.e., vegetation, temporal site elements, materials). In addition, current methods also require labor and time-intensive to cover detailed site properties of complex projects, especially those located in remote and congested areas. Thus, it is clear that this condition may result in a fragmented understanding of local contexts during the design and simulation process.

Thanks to the potential application of attribute point cloud information, relevant features such as insolation analysis and material properties of the existing context can be used to address the aforementioned issues during the conceptual design stage. These features are essentials for architects not only to identify the specific characteristics and performances of the existing context but also to develop an integrated design method for simulating solar geometry based on real contextual datasets. Furthermore, this research conducted a mixed-methods study using both deductive and inductive reasoning with gualitative and guantitative analysis. The research analysis was carried out based on the actions performed in three parts categorized into the research structure, namely theoretical framework, methodological frameworks, and case studies. In parallel, each of these frameworks contains a different research objective that corresponds to each chapter in this thesis. First, theoretical frameworks specifically addressed two main aspects of the research, namely solar envelopes and point cloud data. The first step was conducting a systematic review on computational solar envelopes based on design methods, parameters, digital tools, and design implementation (chapter 2). This chapter aims to review state-of-the-art computational solar envelopes and identify knowledge gaps as well as future directions of the study. Recommendations from the review study were followed up with a preliminary simulation of solar envelopes based on geometric properties of ALS (Aerial Laser Scanning) point cloud data (chapter 3). The goal of this preliminary simulation is not only to test the feasibility of integrating point clouds in the solar envelope simulation, but also to gain critical feedback on relevant and potential attribute point cloud information. For example, exploring radiometric information of the TLS (Terrestrial Laser Scanning) dataset allows one to compute material properties (i.e., thermal and optical) and perform insolation analysis on the surface of the contextual dataset (chapter 4).

Second, the methodological framework focuses on the development of computational workflows for the two proposed models, namely subtractive-solar envelopes (SOLEN) (chapter 5-Part A) and subtractive-shading envelopes (SHADEN) (chapter 5-part B). In this regard, the SOLEN model aims to generate solar envelopes based on a specific amount of direct sunlight from the surrounding context while the SOLEN model was developed to generate an appropriate building mass based on the shading performance criteria. Although these models were established based on the integration of the environmental performance features from the point cloud and subtractive design principles of solar envelopes, both models used these features differently in computational workflows. In SOLEN model, environmental features were used to assess the potential and impact of the final geometry of subtractive solar envelopes.

The last part is case studies, in which the SOLEN and SHADEN model were applied in different contextual and climatic settings. The SOLEN model was applied in Delft, The Netherlands using a hypothetical case study and temperate climate settings while the SHADEN model was applied in Bandung, Indonesia, incorporated with an architectural firm using tropical climatic settings. As an exploratory study, this research aimed at demonstrating the proposed computational method into design practices by conducting an interview with design practitioners (i.e., architects, developer, local government) (chapter 6). This is important not only to identify the feasibility and practicality aspect of the models developed into the existing architectural design scheme but also to gain technical and conceptual feedback comprehensively for future implementation. The interviews began with a brief presentation from the researcher about the background of the study, then continued with a booklet containing procedural steps for computational workflows. As a result of this interview, three main aspects such as computational environments, project cost, and local regulations require further attention to architectural design method implementation.

To sum up, the research developed an integrated computational method for establishing solar geometry based on attribute information stored in point cloud data, and solar and shading envelopes. The method presented in this thesis contributes to assisting architects not only in understanding the existing context comprehensively before starting the design exploration but also in supporting better design decisions during the conceptual design stage. In addition, this thesis also encourages a collaborative and interdisciplinary approach between the research field of remote sensing and architectural design. As for further considerations, this thesis acknowledges some critical aspects such as the requirement of a prerequisite knowledge for the dataset processing, especially for organizing and selecting relevant information from large datasets, the cost of tool affordability that is currently still expensive for small architectural design firms, and improvements to computational workflows to address factors such as dataset correction parameters, simulations from dense datasets, and design implementation using a variety of urban settings.

Samenvatting

In de afgelopen jaren is de groei van de stedelijke bevolking snel toegenomen, parallel met de escalatie van het energieverbruik van de bouwsector. De UN SDG's (United Nations Sustainable Development Goals) spelen een belangrijke rol bij het stimuleren van strategieën zoals groene bouwtechnologie en highperformance envelop om de efficiëntie van energie-intensiteit in gebouwen te beheren en te berekenen. Hoewel deze benaderingen veelbelovende technische oplossingen bevorderen, zijn pogingen om de milieukwaliteit van de gebouwde omgevingen te verbeteren vaak verwaarloosd, vooral met betrekking tot passieve ontwerpstrategieën in de conceptuele ontwerpfase. Dit is cruciaal voor architecten om niet alleen onverwachte storingen te voorkomen nadat een nieuw gebouw in een nieuwe context is gebouwd, maar ook om de microklimatologische omstandigheden vanuit de ontwerpcontext volledig te begrijpen. Door de computationele ontwikkeling van geometrische modellen voor zonne-energie, onderzocht dit proefschrift een nieuwe methode voor het ontwerpen en analyseren van bestaande omgevingen door gebruik te maken van attributen puntwolkeninformatie en zonne- en schaduwenveloppen. De ontwikkelde methode helpt niet alleen andere architecten om onderlinge afhankelijkheden tussen het nieuwe gebouw en de lokale context te construeren, maar ook om geïnformeerde ontwerpbeslissingen te nemen voor een hoog presterende gebouwmassa.

Een state-of-the-art computationele methode voor het genereren van zonnegeometrieën vormt echter een grote uitdaging bij het begrijpen van locatiekenmerken van bestaande omgevingen. Bestaande methoden construeren voornamelijk 3D digitale contextuele modellen op basis van architecturale geometrische basisvormen (d.w.z. solide modelleerplatforms), momenteel geïsoleerd van eigenschappen rond de lokale context (d.w.z. vegetatie, tijdelijke site-elementen, materialen). Bovendien vergen de huidige methoden ook arbeids- en tijdintensief om gedetailleerde locatieeigenschappen van complexe projecten te dekken, vooral die in afgelegen en drukke gebieden. Het is dus duidelijk dat deze voorwaarde kan resulteren in een gefragmenteerd begrip van lokale contexten tijdens het ontwerp- en simulatieproces.

Dankzij de mogelijke toepassing van attribuut-puntenwolkeninformatie kunnen relevante kenmerken zoals analyse van de instraling en materiaaleigenschappen van de bestaande context worden gebruikt om de bovengenoemde problemen aan te pakken tijdens de conceptuele ontwerpfase. Deze kenmerken zijn essentieel voor architecten om niet alleen de specifieke kenmerken en prestaties van de bestaande context te identificeren, maar ook om een geïntegreerde ontwerpmethode te ontwikkelen voor het simuleren van zonnegeometrie op basis van echte contextuele datasets.

Bovendien heeft dit onderzoek een studie met gemengde methoden uitgevoerd met behulp van zowel deductief als inductief redeneren met kwalitatieve en kwantitatieve analyse. De onderzoeksanalyse werd uitgevoerd op basis van de acties die werden uitgevoerd in drie delen onderverdeeld in de onderzoeksstructuur. namelijk theoretisch kader, methodologische kaders en casestudy's. Tegelijkertijd bevat elk van deze raamwerken een andere onderzoeksdoelstelling die overeenkomt met elk hoofdstuk in dit proefschrift. Ten eerste behandelden theoretische kaders specifiek twee hoofdaspecten van het onderzoek, namelijk zonne-enveloppen en puntenwolken. De eerste stap was het uitvoeren van een systematische review van computationele zonne-enveloppen op basis van ontwerpmethoden. parameters, digitale tools en ontwerpimplementatie (hoofdstuk 2). Dit hoofdstuk beoogt een overzicht van de modernste computationele zonne-enveloppen en het identificeren van hiaten in de kennis en toekomstige richtingen van het onderzoek. Aanbevelingen uit de reviewstudie werden opgevolgd met een voorlopige simulatie van zonne-enveloppen op basis van geometrische eigenschappen van ALS (Aerial Laser Scanning) puntenwolkgegevens (hoofdstuk 3). Het doel van deze voorlopige simulatie is niet alleen om de haalbaarheid te testen van het integreren van puntenwolken in de simulatie van de zonne-envelop, maar ook om kritische feedback te krijgen over relevante en potentiële attribuut-puntenwolkeninformatie. Door bijvoorbeeld radiometrische informatie van de TLS-dataset (Terrestrial Laser Scanning) te onderzoeken, kan men materiaaleigenschappen (d.w.z. thermisch en optisch) berekenen en een analyse van de instraling uitvoeren op het oppervlak van de contextuele dataset (hoofdstuk 4).

Ten tweede richt het methodologische raamwerk zich op de ontwikkeling van computationele workflows voor de twee voorgestelde modellen, namelijk subtractieve zonne-enveloppen (SOLEN) (hoofdstuk 5-deel A) en subtractieveschaduw-enveloppen (SHADEN) (hoofdstuk 5-deel B). In dit opzicht beoogt het SOLEN-model zonneschermen te genereren op basis van een specifieke hoeveelheid direct zonlicht uit de omringende context, terwijl het SOLEN-model is ontwikkeld om een geschikte bouwmassa te genereren op basis van de prestatiecriteria voor zonwering. Hoewel deze modellen werden opgesteld op basis van de integratie van de milieuprestatiekenmerken van de puntenwolk en subtractieve ontwerpprincipes van zonne-enveloppen, gebruikten beide modellen deze functies op verschillende manieren in computationele workflows. In het SOLEN-model werden omgevingskenmerken gebruikt om het potentieel en de impact van de uiteindelijke geometrie van subtractieve zonne-enveloppen te beoordelen, terwijl het SHADENmodel het gebruikte om de uiteindelijke geometrie van subtractieve schaduwenveloppen te genereren.

Het laatste deel is casestudy's, waarin het SOLEN- en SHADEN-model werden toegepast in verschillende contextuele en klimatologische omstandigheden. Het SOLEN-model werd toegepast in Delft, Nederland met behulp van een hypothetische casestudy en gematigde klimaatomgevingen, terwijl het SHADEN-model werd toegepast in Bandung, Indonesië, opgenomen met een architectenbureau dat gebruik maakte van tropische klimatologische omgevingen.

Als verkennend onderzoek had dit onderzoek tot doel de voorgestelde computationele methode in ontwerppraktijken aan te tonen door een interview te houden met ontwerpdeskundigen (d.w.z. architecten, ontwikkelaar, lokale overheid) (hoofdstuk 6). Dit is niet alleen belangrijk om het haalbaarheids- en praktische aspect van de modellen die zijn ontwikkeld in het bestaande architectonische ontwerpschema te identificeren, maar ook om technische en conceptuele feedback te krijgen voor een uitgebreide implementatie voor toekomstige implementatie. De interviews begonnen met een korte presentatie van de onderzoeker over de achtergrond van het onderzoek, en vervolgden met een boekje met procedurele stappen voor computationele workflows. Als resultaat van dit interview vereisen drie hoofdaspecten, zoals computeromgevingen, projectkosten en lokale voorschriften, verdere aandacht voor architectonische ontwerppraktijken bij de implementatie van ontwerpmethoden.

Samenvattend, het onderzoek ontwikkelde een geïntegreerde rekenmethode voor het vaststellen van zonnegeometrie op basis van attribuutinformatie opgeslagen in puntenwolkgegevens en zon- en schaduwomhullingen. De methode die in dit proefschrift wordt gepresenteerd, helpt architecten niet alleen om de bestaande context volledig te begrijpen voordat ze met de ontwerpverkenning beginnen, maar ook om betere ontwerpbeslissingen te ondersteunen tijdens de conceptuele ontwerpfase. Daarnaast moedigt dit proefschrift ook een collaboratieve en interdisciplinaire benadering aan tussen het onderzoeksveld van teledetectie en architectonisch ontwerp. Wat verdere overwegingen betreft, erkent dit proefschrift enkele kritische aspecten, zoals de vereiste van een vereiste kennis voor de verwerking van de dataset, met name voor het organiseren en selecteren van relevante informatie uit grote datasets, de kosten van betaalbaarheid van tools die momenteel nog steeds duur zijn voor kleine architectenbureaus., en verbeteringen aan computationele workflows om factoren aan te pakken zoals correctieparameters voor datasets, simulaties van dichte datasets en ontwerpimplementatie met behulp van een verscheidenheid aan stedelijke omgevingen.



1 Introduction

1.1 General Background

The United Nations (UN) Population Division predicts a significant increase in world population growth of around 3.2 billion from 2019 to 2100 [1]. This trend simultaneously underlines the rapid urban growth which presents many challenges and opportunities in the future. One of them is the implementation of an urban development agenda which must meet the criteria of 11 Sustainable Development Goal (SDG) programs, namely, to make cities inclusive, safe, resilient and sustainable [2]. The integration between renewable energy and passive design strategies plays a crucial part for the development of future urban planning, not only to reduce the increase of annual energy consumption of the building sectors [3] but also to improve the quality of the built environment.

In order to maintain a reciprocal relationship between a new building and the local context, architects are responsible for ensuring that environmental performance aspects have been prioritized during the design process. It is an essential task during the conceptual design stage, where the most significant design decisions are taken [4]. Therefore, architects are expected to be able to avoid potential failures before new construction is carried out, especially related to the unexpected microclimatic impacts caused by a building and imposed on the surrounding context (and vice versa).

In the past, the concept of vernacular buildings played a crucial role when dealing with environmentally sustainable designs, especially those related to passive solar designs. They have effectively granted massive reductions in the unsustainable use of energy resources by considering relevant aspects such as building orientation, location, geometry, and material choices. To date, continuous improvements to passive solar design principles have attracted increasing attention, such as solar-form finding [5] [6], solar energy simulation for building facades [7] [8], and solar potential for the urban areas [9] [10]. However, these existing studies predominantly

focus on context-oriented buildings and energy quantities that unfortunately lack a contextual analysis. In most cases, this issue occurs in architectural design practice due to some practical constraints such as project deadlines, computational skills, limited budgets, and outdated regulations [20].

On the other hand, one concept in architectural design practice that specifically considers solar performance aspects in a local context is related to solar envelopes. Initially introduced by Knowles [11], solar envelopes conceptually provide relevance in addressing solar accessibility between new buildings and existing contexts. It specifically results in a maximum volumetric shape or building massing that guarantees direct solar access to the surrounding buildings for a predefined period [12]. In other words, it is a solar access protection approach that accommodates the development potential of urban forms based on solar irradiation needs.

The basic concept of solar envelopes has significantly encouraged architects to explore a variety of envelope generation methods. Over the last few decades, several developments in computational solar envelopes have been explored, resulting in various advanced workflows, design parameters, digital tools, and a broad range of case studies [13]. These developments allow architects to understand specific computational methods and parameters that are relevant and applied to different contextual urban settings. Nevertheless, existing methods predominantly do not support an understanding of the site characteristics of an existing environment, especially in relation to the geometric properties of surrounding contexts (i.e., vegetation, surface properties, and materials) [14] [15]. The absence of these properties can lead to enormous discrepancies found during an environmental performance analysis such as misinterpretation of simulation results and so forth.

Thanks to major advances in technological developments, computational solar envelopes can now be implemented in various urban settings. In that respect, climate consideration becomes one of the pivotal aspects to pay attention to given that most of the existing methods merely focus on the temperate zone in countries found in the Southern and Northern Hemisphere which have different climatic conditions during the four-seasons. Therefore, when it comes to tropical countries or those located on the Equator such as Indonesia, the suitability of this context becomes less appropriate because the design objectives and parameters are different and require particular adjustments to the local climate [16]. For example, since buildings in Indonesia are typically designed to block direct sun access due to hot daytime temperatures, the goal of solar envelopes in guaranteeing sun access ultimately becomes less applicable. Furthermore, with the advancement of 3D laser scanning technology in capturing complex information from real contexts, potential features stored in the point cloud such as geometric and radiometric information may include making information relevant to the aforementioned issues available. As a basis for creating an informed design context, the attribute information of point cloud (i.e., position – XYZ, color – RGB, and reflection intensity – I) enables architects not only to deliver a rapid and realistic contextual modeling but also to conduct environmental performance assessments that support the simulation of solar envelopes both as a performance evaluator and form generator.

Following up on relevant aspects from the potential application of point cloud data and further consideration of current solar envelopes methods, this research specifically explores an integrated computational design method for solar geometry based on attribute information stored in point cloud data. Similarly to Szokolay [17], Bruce [18], and Alread and Leslie [19], the term solar geometry in this research not only covers geometric relationships from thermal aspects but also includes a comprehensive discussion of the desired lighting conditions. Accordingly, this term will regularly be used to describe the concept of solar and shading envelopes as the main models of the research.

Ultimately, this research should be of interest to practitioners (e.g., architects, urban planners, building engineers), decision makers (e.g., local municipalities), and academics in the areas of architectural engineering, design simulation, energy sustainability, and remote sensing.

1.2 **Problem Statement**

As part of the passive design strategy, the development of computational solar envelopes plays a major role not only to improve the quality of the built environment but also to construct a cooperative performance exchange between new buildings and their local contexts. However, a state-of-the-art computational method of solar envelopes poses a great challenge in understanding site characteristics from the given context. Existing methods predominantly construct 3D context models based on basic architectural geometric shapes (i.e., solid modelling platforms [21]), which are currently isolated from the surrounding properties of local contexts (i.e., vegetation, temporal site elements [14] [15], materials [22]). In addition, computational methods of solar envelopes are currently a tedious and time-consuming task especially in covering the detailed site properties of complex projects located in isolated and congested areas. It is clear that this condition may result in a fragmented understanding of the local context during the design and simulation process.

With the potential application of attribute information of point cloud data, it is necessary to consider relevant parameters such as the surface and material properties of existing contexts during the simulation of solar geometries, which are currently absent in computational frameworks, especially to address the aforementioned issues in the conceptual design stage. As such, the new method is required to enable architects not only to measure specific performances of the local context but also to identify vulnerable areas that may affect the proposed design.

1.3 **Research questions, aims and objectives**

1.3.1 **Research questions**

Considering the discussion in the previous section, this research addresses the following research questions:

How can we develop a computational design method of solar geometries as a decision-making support for architects based on solar and shading envelopes and attribute point cloud information?

In order to answer this overarching question, the research investigates further five sub-questions in association with its objectives, which are addressed in different chapters of this thesis:

1 What is a state-of-the-art review of computational design methods for establishing solar envelopes? (Chapter 2) – A comprehensive review of computational solar envelopes.

- 2 How do geometric properties of point cloud data contribute to the simulation of solar envelopes? (*Chapter 3*) *Preliminary simulation of solar envelopes based on geometric properties of the point cloud.*
- ³ What are potential features that can be developed further from attribute information of point cloud data? (*Chapter 4*) *Exploring the potential applications of geometric and radiometric information stored in point cloud data.*
- 4 How can new computational models for constructing solar geometry in temperate and tropical urban contexts integrate attribute information stored in point cloud data (i.e., position-XYZ, color-RGB, and reflection intensity-I)? (Chapter 5) – Computational models for constructing solar geometry based on attribute point cloud information.

1.3.2 **Aim**

The aim of the research is to develop an integrated computational method to analyze and design solar geometry in the built environment during the conceptual design stage. The ultimate goal is to allow architects to make informed-design decisions towards high-performance building massing based on solar and shading performance criteria, as well as geometric and radiometric information from point cloud data.

1.3.3 **Objectives**

In association with the aforementioned aim, the research has formulated the following objectives:

- 1 To investigate a state-of-the-art review of solar envelopes in order to identify variations in computational methods, tools, and design applications.
- 2 To conduct a preliminary study that focuses on the use of point cloud geometric properties in the existing solar envelopes method.
- 3 To investigate the potential application of attribute information stored in point cloud data through material properties and solar radiation analysis of the existing context.
- ⁴ To develop and demonstrate a computational model for subtractive solar envelopes and subtractive shading envelopes based on geometric and radiometric information of point cloud data, considering the temperate and tropical climate, respectively.

1.4 Research output

As a final result, this research produces an integrated computational method for establishing solar geometries based on attribute information stored in point cloud data. This method consists of subtractive solar envelopes and subtractive shading envelopes that refer to the SOLEN and SHADEN model, respectively. Each model contains a series of computational workflows that are developed to generate geometric envelopes based on its design objectives, contextual settings, and climatic parameters. The SOLEN model aims to produce solar envelopes based on the specific amount of direct sunlight obtained from ray tracing analysis between the 3D polyhedra of the new building and the existing site's point cloud dataset. In this regard, radiometric point cloud information is used to conduct insolation analysis and to detect material behavior of the existing environment as part of the environmental performance assessment of the solar envelope's final geometry. The SOLEN model is particularly applied to temperate climates with selected case study in Delft, the Netherlands. Meanwhile, the SHADEN model is developed to generate the appropriate mass based on shading performance criteria obtained from the results of ray tracing analysis between the 3D polyhedra of the new building and the point cloud dataset from the existing site. In this part, radiometric point cloud information is used to investigate material properties and sun visibility of the given context as part of form generation process for establishing the final geometry of shading envelopes. This model is specifically applied to tropical climates with a case study in Bandung, Indonesia.

1.5 Research method

In order to answer the research questions and to address the main aims and objectives of the thesis, the research follows a mixed-methods study using both deductive and inductive reasoning with qualitative and quantitative analysis, depending on the action taken in each section. In particular, the research structure is formulated into three main parts (see Figure 1.1).

The first part contains theoretical frameworks (i.e., design parameters of solar envelopes, attribute information stored in point cloud data) incorporated with preliminary simulations of solar envelopes based on geometric properties of point cloud data. This part also simultaneously supplies the fundamental concept and relevant elements for the following sections. The second part deals with the methodological framework to develop the computational design workflow of proposed models (i.e., SOLEN, SHADEN). The third part focuses on design implementation, in which case studies are applied to each proposed model. Although the computational workflows presented for chapter 5-Part B were predominantly developed from chapter 5-Part A, the sequence of these parts does not need to be linear. This is because some tasks have the same input and output which can be performed in parallel in later steps. For example, the computational design workflows for both chapters 5-Part A and 5-Part B received similar input regarding material properties and design parameters from chapter 3 and 4. Nevertheless, the simulation process and the final output remain different due to the different design workflows and case studies.

Furthermore, the aforementioned parts contain different tasks that correspond to each research question, as follows:



FIG. 1.1 Overview of research structure and methods
Part 1 – Theoretical framework

The first part specifically constructs a theoretical framework for the two main aspects of the research, namely solar envelopes and point cloud data. This framework is generated by first conducting a systematic review of computational solar envelopes based on design methods, parameters, digital tools, and case studies. It aims to map state-of-the-art methods for solar envelopes and identify knowledge gaps from existing studies. Recommendations from the review studies were followed up with a preliminary simulation to test the feasibility and relevance of point cloud data addressing with the current simulation of solar envelopes. This simulation allows for gaining critical feedback that is used to define *firstly*, further potential application of attribute information of point cloud data (i.e., radiometric information) by exploring the material properties of existing contexts and *secondly*, the design parameters for the subtractive method of solar envelopes.

This part has been published as follows:

- [13] M. F. Alkadri, F. De Luca and M. Turrin and S. Sariyildiz, "Understanding Computational Methods for Solar Envelopes Based on Design Parameters, Tools, and Case Studies: A Review," *Energies*, vol. 13, no. 13, pp. 3302-3326, 2020.
- [23] M. F. Alkadri, M. Turrin and S. Sariyildiz, "The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts," *Advances in Computational Design*, vol. 3, no. 4, pp. 319-338, 2018.
- [24] M. F. Alkadri, M. Turrin and S. Sariyildiz, "Identifying the surface material of the existing environment through point cloud data attributes," in *Symposium on simulation for architecture and urban design (SimAUD 2018)*, Delft, 2018.
- [25] M. F. Alkadri, M. Turrin and S. Sariyildiz, "Toward an Environmental Database
 Exploring the material properties from the point cloud data of the existing environment," in *Conference: eCAADe 2018 - Computing for a better tomorrow*, Lodz, 2018.
- [26] M. F. Alkadri, M. Turrin and S. Sariyildiz, "A computational workflow to analyse material properties and solar radiation of existing contexts from attribute information of point cloud data," *Building and Environment*, vol. 155, no. -, pp. 268-282, 2019.

Part 2 – Methodological framework

The second part focuses on developing computational design method for the two proposed models, namely subtractive-solar envelopes (SOLEN) and subtractive-shading envelopes (SHADEN). Although these models address a similar workflow for calculating the material properties of existing contexts, the SHADEN model was principally established based on the further development of the SOLEN model. In particular, the SOLEN model was developed considering material properties as evaluation criteria for the resulting envelope, while the SHADEN model employed material properties as part of the generation criteria to establish the final envelope.

Part 3 – Case studies

The third part discusses the design implementation of SOLEN and SHADEN model. These models were implemented in selected case studies with different contextual and climatic settings. The SOLEN model was applied in Delft, The Netherlands using a hypothetical case study and a temperate climate setting while the SHADEN model was applied in Bandung, Indonesia, incorporated with an architectural firm using a tropical climatic setting. In parallel, reviews with design practitioners (i.e., architects) were conducted to gain feedback on the relevance of implementing the proposed models in their design workflows.

This part has been published as follows:

- [27] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "Making use of point cloud for generating subtractive solar envelopes," in *eCAADe SIGraDi 2019: Architecture in the Age of the 4th Industrial Revolution*, Porto, 2019.
- [28] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "An integrated approach to subtractive solar envelopes based on attribute information from point cloud data," *Renewable and Sustainable Energy Reviews*, vol. 123, no. -, p. 109742, 2020.
- [29] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "A Computational Workflow for Generating a Voxel-based Design Approach Based on Subtractive Shading Envelopes and Attribute Information of Point Cloud Data," *Remote Sensing*, vol. 12, no. 16, p. 2561, 2020.

1.6 Research relevance

1.6.1 Scientific relevance

The research scientifically expands on knowledge about passive solar design, especially as related to the new computational method of solar geometries in the conceptual design stage of the architectural design process. A systematic review of computational solar envelopes contributes not only to highlighting current possibilities and gaps for design methods, digital tools, and simulation parameters but also to further advocate for the implementation of solar envelopes in different contextual and climatic settings.

Furthermore, the exploration of attribute information stored in point cloud data enables the expansion of functional properties available in 3D scanning technology. In particular, radiometric information of point cloud (e.g., position, color, reflection intensity) makes it possible to identify material properties of an existing environment that are simultaneously relevant for performing environmental analysis from a design context.

With the integrated computational workflow between point cloud data and subtractive solar envelopes in chapter 5-A and subtractive shading envelopes in chapter 5-B, the method proposed in this research has confirmed the feasibility and contribution of new potential applications of 3D scanning into the design process of architectural projects. In other words, the research presents a multidisciplinary approach between architectural design and engineering that enlarges the scopes of development of remote sensing, computational design, and renewable energy.

1.6.2 Societal relevance

The research proposes a computational method for a passive solar design strategy that supports architects in making design decisions during the conceptual design stage. This method plays a key role not only to conduct environmental performance assessments between new buildings and existing contexts but also to understand comprehensively microclimatic conditions from the design context. Therefore, architects can avert the potential for unforeseeable failures that might occur before placing a new building into a real context.

The method proposed in this research also introduces new potential areas of 3D scanning technology for architects in design practice. For this reason, the community and industrial markets for 3D scanners are gradually being adapted and relevant for day-to-day architectural practices in terms of availability and cost-effectiveness. Therefore, further development of this technology can be affordable for architectural firms of all scales in the long term.

Lastly, the research allows decision makers (e.g., local municipalities, national government) and practitioners (e.g., architects, urban planners, building engineers, sustainability experts) to reflect upon and evaluate current regulatory parameters, especially those related to shadow fences, solar radiation analysis for the building facades, and solar access criteria. It is important to provide holistic guidance for architects regarding sustainable and green building policies that are currently missing in the conceptual design stage, especially in Indonesia. Most often, environmental performance assessments are carried out after design decisions have been taken, thus neglecting many crucial parameters in the early phase of design.

References

- H. Ritchie and M. Roser, "Urbanization," 2020. [Online]. Available: https://ourworldindata.org/ urbanization. [Accessed 30 1 2020].
- U. Nations and P. D. Department of Economic and Social Affairs, "World population proscpects 2019: Highlight," Unied Nations, New York, 2019.
- [3] IEO, "International Energy Outlook 2019 with projections to 2050," U.S. Energy Information Administration Office of Energy Analysis, U.S. Department of Energy, Washington, 2019.
- S. Sariyildiz, Conceptual Design by Means of Islamic- Geometric-Patterns within a CAADEnvironment., Delft: PhD-thesis, Faculty of Architecture, TU Delft, 1991.
- [5] M. Leidi and A. Schlüter, "Exploring urban space Volumetric site analysis for conceptual design in the urban context," *International journal of architectural computing*, pp. 157-182, 2013.
- [6] J. Da Veiga and P. La Roche, "A Computer Solar Analysis Tool for the Design and Manufacturing of Complex Architectural Envelopes: EvSurf," in *Proceedings of the 6th Iberoamerican Congress of Digital Graphics* [SIGraDi 2002], Caracas, 2012.
- [7] B. S. Obaidat, "A Simulation Model for Defining Three Different Solar Accesses in Site Planning," *Simulation*, vol. 65, no. 6, pp. 357-371, 1995.
- [8] A. Jain, K. Kensek and D. Noble, "Interactive Web-based teaching tool for simplified 3D analysis of solar rhythms," *Automation in Construction*, vol. 8, no. 2, pp. 181-194, 1998.
- [9] S. Freita, C. M. Catita, P. Redweik and M. Brito, "Modelling solar potential in the urban environment: State-ofthe-art-review," *Renewable and sustainable energy reviews*, vol. 41, no. -, pp. 915-931, 2015.
- [10] G. Lobaccaro, F. Frontini, G. Masera and T. Poli, "SolarPW: A new solar design tool to exploit solar potential in existing urban areas," *Energy Procedia*, vol. 30, pp. 1173-1183, 2012.
- [11] R. L. Knowles, Energy and form: An ecological approach to urban growth, Cambridge: The MIT Press, 1974.
- [12] R. L. Knowles and R. D. Berry, Solar envelope concepts; Moderate density Building applications, Springfield: National technical information service U.S. Department of Commerce, 1980.

- [13] M. F. Alkadri, F. De Luca and M. S. S. Turrin, "Understanding Computational Methods for Solar Envelopes Based on Design Parameters, Tools, and Case Studies: A Review," *Energies*, vol. 13, no. 13, pp. 3302-3326, 2020.
- [14] Y. Arayici, A. Hamilton and P. Gamito, "Modelling 3D Scanned Data to Visualize and Analyse the Built Environment for Regeneration," *Surveying and Built Environment*, vol. 17, no. 2, pp. 7-28, 2006.
- [15] Y. Arayici, "Towards Building Information Modelling for Existing Structures," *Structural Survey*, vol. 26, no. 3, pp. 210-222, 2008.
- [16] B. Paramita and M. Koerniawan, "Solar envelope assessment in tropical region building case study: vertical settlement in Bandung, Indonesia," in *The 3rd International conference on sustainable future for human security SUSTAIN 2012*, Kyoto, 2013.
- [17] S. V. Szokolay, Solar geometry, Brisbane: PLEA: Passive and Low Energy Architeture International in association with the University of Queensland, Department of Architechture, 1996.
- [18] G. Bruce, "High density, Low energy: achieving useful solar access for Dublin's multi-storey apartment developments," in *PLEA*, Dublin, 2008.
- [19] J. Alread and T. Leslie, Design-Tech: Building Sciences for Architects, Oxford: Architectural Press, ElsevierLtd., 2007.
- [20] U. Komac, "Why architectural design and research are not more relevant in the real world," Lusofona Journal of Architecture and Education, vol. 11, no. 11, pp. 573-578, 2014.
- [21] N. N. Staneva, "Approaches for generating 3D solid models in AutoCAD and solid works," *Journal of Engineering*, vol. VI, no. 3, pp. 28-31, 2008.
- [22] I. Yüksek and T. T. Karadayi, "Energy-Efficient Building Design in the Context of Building Life Cycle," in Sustainable Development, London, IntechOpen, 2017, pp. 93-123.
- [23] M. F. Alkadri, M. Turrin and S. Sariyildiz, "The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts," *Advances in Computational Design*, vol. 3, no. 4, pp. 319-338, 2018.
- [24] M. F. Alkadri, M. Turrin and S. Sariyildiz, "Identifying the surface material of the existing environment through point cloud data attributes," in *Symposium on simulation for architecture and urban design (SimAUD 2018)*, Delft, 2018.
- [25] M. F. Alkadri, M. Turrin and S. Sariyildiz, "Toward an Environmental Database Exploring the material properties from the point cloud data of the existing environment," in *Conference: eCAADe 2018 - Computing* for a better tomorrow, Lodz, 2018.
- [26] M. F. Alkadri, M. Turrin and S. Sariyildiz, "A computational workflow to analyse material properties and solar radiation of existing contexts from attribute information of point cloud data," *Building and Environment*, vol. 155, no. -, pp. 268-282, 2019.
- [27] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "Making use of point cloud for generating subtractive solar envelopes," in *eCAADe SIGraDi 2019: Architecture in the Age of the 4th Industrial Revolution*, Porto, 2019.
- [28] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "An integrated approach to subtractive solar envelopes based on attribute information from point cloud data," *Renewable and Sustainable Energy Reviews*, vol. 123, no. -, p. 109742, 2020.
- [29] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "A Computational Workflow for Generating a Voxel-based Design Approach Based on Subtractive Shading Envelopes and Attribute Information of Point Cloud Data," *Remote Sensing*, vol. 12, no. 16, p. 2561, 2020.



2 Review Of Computational Solar Envelopes

This chapter has been published as: Alkadri, M. F., De Luca, F., Turrin, M. and Sariyildiz, S., 2020. Understanding computational design methods of solar envelopes based on design parameters, tools, and case studies. *Energies*, 13(13), pp. 3302. The layout has been adjusted to fit the template of this thesis.

This chapter deals with a state-of-the-art review of computational solar envelope that specifically caters to the background study of the main topic in this research. In particular, a comprehensive investigation of the available knowledge of solar envelopes regarding computational methods, parameters, tools and design applications is presented. This review study aims to identify the different characteristics and levels of complexity for each categorized design method of solar envelopes based on 58 selected references. Under the three selected literature databases, namely, Web of Sciences, Scopus, and Google Scholar, specific topics are determined to stipulate the scope of reviews, which are conceptual themes (i.e., solar architecture, solar envelopes, and solar access), design workflow (i.e., computational design, solar design, and solar simulation), and contextual settings (i.e., urban planning, urban design, and architectural design). In addition, knowledge gaps and future developments of computational solar envelopes are discussed extensively to provide architects with an inclusive understanding of conceptual frameworks of solar envelopes as a passive design strategy. Ultimately, this literature review serves as a part of the theoretical foundation for the following section of the research presented. In particular, knowledge gaps and future directions of the study address potential features that will be considered for the development of new computational method of solar envelopes.

Understanding Computational Method of Solar Envelopes Based on Design Parameters, Tools, and Case Studies: A Review

The increasing population density in urban areas simultaneously impacts the trend of ABSTRACT energy consumption in building sectors and UHI (urban heat island) effects of urban infrastructure. Accordingly, passive design strategies to create sustainable buildings play a major role in addressing these issues, while solar envelopes prove to be a relevant concept that specifically considers the environmental performance aspects of a proposed building given their local contexts. As significant advances have been made over the past decades regarding the development and implementation of computational solar envelopes, this study presents a comprehensive review of solar envelopes while specifically taking into account design parameters, digital tools, and the implementation of case studies in various contextual settings. This extensive review is conducted in several stages. First, an investigation of the scope and procedural steps of the review is conducted to frame the boundary of the topic to be analyzed within the conceptual framework of solar envelopes. Second, comparative analyses between categorized design methods in parallel with a database of design parameters are conducted, followed by an in-depth discussion of the criteria for the digital tools and case studies extracted from the selected references. Third, knowledge gaps are identified, and the future development of solar envelopes is discussed to complete the review. This study ultimately provides an inclusive understanding for designers and architects regarding the progressive methods of the development of solar envelopes during the conceptual design stage.

KEYWORDS solar envelopes; passive design strategies; computational design methods

2.1 Introduction

By 2050, the UN (united nations) estimates that the world's population in urban areas will increase by approximately 68 % [1] with urban dwellers around 6.7 billion [2]. This trend is simultaneously followed by a continuous increase in energy consumption from the building sector, which will share 1.3 % of annual increment and reach 22 % in 2050 [3]. This means that a future urban planning requires sustainable strategies to deal with the energy use and building emissions [4]. Some strategies have been proposed to tackle these issues using the NDCs (nationally determined contributions) and UN SDG (sustainable development goal) programs. For example, many researchers have actively developed specific methods and tools to provide more practical guidance regarding sustainable buildings and construction technologies, such as the adoption of green building technologies [5] [6] and the enhancement of building energy efficiency [7] [8]. However, past surveys did not discuss the conceptual domain of passive design strategies but rather focused predominantly on the technical building operations such as, HVAC (heating, ventilation, and air conditioning) system [9] [10]. Consequently, a knowledge gap exists when addressing the environmental performances of building design especially at the conceptual design stage. In this case, this paper contributes by increasing the knowledge on passive design strategies. Specifically, it comprehensively reviews and examines computational methods, parameters, tools, and case studies related to solar envelopes.

This review is relevant in that it addresses a contextual design approach in which solar envelopes play a significant role in enhancing the guality of the built environment. Furthermore, this review integrates the environmental performance of a new building with the existing context and contributes to the most crucial design decisions made during the early design phase. In this respect, the concept of solar envelopes has made a relevant contribution by addressing the solar accessibility of new buildings and their existing contexts. By definition, solar envelopes are composed of the maximum volumetric container as determined by the amount of desirable or required sun access without considering the shadowing of adjacent buildings [6]. Accordingly, the envelope of proposed designs can be maximized without compromising the solar rights of surrounding buildings during the critical period. During the conceptual design process, this concept is useful for architects, as they seek to avert potential failures once a new building has been constructed. especially with respect to negative microclimatic impacts. In design practices, this approach has successfully been implemented by the Dutch architectural and urban design firm MVRDV through the project of P15 Ravel Plot, which is located in the Zuidas district, 1082 LC, Amsterdam [11] and the Grotius Tower II, which is

located in the area of the Prince Bernard Viaduct, Den Haag [12]. These projects have similarly addressed the idea of solar-oriented design by integrating the optimal sightline for each housing unit with the terraces and greenery landscape. In so doing, proposed buildings have successfully presented high performing envelopes that fulfil both geometric and environmental performance quality.

Since the inception of the solar envelope, several methods for its determination have been developed. For example, Topaloglu [13] describes three simple techniques for establishing solar envelope, namely, the descriptive, profile angle, and 2D orthographic projection techniques. The descriptive technique adopts the initial solar envelopes concept introduced by Knowles [14]. As such, it intersects the vertical planes plotted on the selected site by using the trigonometric principles of the solar azimuth, altitude (α) and, cut-off times (i.e., daily and annual time limits). For example, given a full day setting, the morning sun governs the envelope's boundary of its western limits, while the afternoon sun establishes the envelope's shape of its eastern limits. This same mechanism applies to the annual time setting by calculating the sun's position during the winter and summer months. The profile angle technique consists of an intersection between inclined planes that are generated on each edge of the plot according to minimum solar angles as determined by a different orientation. In general, the profile angle is also employed to determine the geometric positions of the shading devices, the penetration of the sun's rays into a room, and the shading line on the building's facades. The orthographic projection technique employs a mechanism similar to that of the profile angle techniques but only applies to rectangular sites with two elevation planes within a two-dimensional projection. While these methods are valuable and convenient, further consideration of several aspects is required including the simulation time, range of input parameters, and accuracy of the 3D visualization, especially with respect to complex architectural forms [15]. In contrast to the above, this review investigates computational methods that offer several effective ways to address these challenges. Hence, ultimately, this study advances the work on the sustainable design approach by providing an overview on the current state of the computational environment of solar envelopes and exposing critical gaps for future consideration.

Having introduced the relevance and basic principles of solar envelopes, this paper is organized as follows. Section 2 describes the scope and procedure of the review, and Section 3 presents the results of the review and discusses the computational design methods for solar envelopes and related design parameters are presented. For each design method described in Section 3, Section 4 focuses on the aspects of the computational environment in parallel with the digital tools and the implementation of the case studies. Section 5 then addresses the knowledge gaps and new directions for future research on solar envelopes, and Section 6 presents the conclusions.

This review addresses the main question that frames the survey of existing computational solar envelopes, i.e., What are the performance criteria and related computational methods for generating solar envelopes? This question simultaneously leads us to explore specific tasks and features of various design parameters, tools, and the implementation of case studies found in each design method of solar envelopes. The ultimate goals are to identify and understand the most basic and advanced parameters and computational methods for constructing solar envelopes and to analyze relevant factors that affect the complexity and flexibility of solar envelope methods from the perspective of the user. Explicit awareness of these issues is crucial for the comprehensive understanding of the current use of solar envelopes.

The articles for review are accessed through three selected literature databases, namely, *Web of Science (WoS), Scopus,* and *Google Scholar (GS)*. Although there are many database searching that can be used for a literature review, this study focuses on employing Scopus, WoS, and GS because they are part of wide citation and bibliographic searching platform that have been designed to support scientific and research environment globally. To be more specific, Scopus and WoS similarly present substantial factual information that includes a number of peer-reviewed literature and article indexed through Elsevier and ISI citation database, respectively. On the other hand, GS has also been extensively utilized for a large interdisciplinary field coverage with a wide unique type of materials (i.e., PDF files, word docs, and technical reports), including indexed and non-indexed articles, especially for Master and PhD thesis that contains relevant information for the topic of solar envelopes [16] [17]. Accordingly, it can complement other databases such as Scopus and WoS that are predominantly based on indexed publication.

Furthermore, a new direction for the further development of solar envelopes is identified as a result of this review, which considers three main topics and sub-topics, namely, conceptual themes (i.e., solar architecture, solar envelopes, and solar access), design workflow (i.e., computational design, solar design, and solar simulation), and contextual settings (urban planning, urban design, and architectural design). The scientific findings of the reference databases are narrowed by setting the timespan to range between 1960 and 2019, given that the term "solar envelopes" was initially introduced in the scientific literature by Knowles around 1970 [18] although some authors, such as Giacomo [19], Galton [20], and

Atkinson [21] implicitly discussed related topics regarding solar access prior to Knowles' work. The detailed parameters used for reference databases are shown in Table A.1 in Appendix 2A.

Given that the resulting search remains extremely broad, the screening process is then weighted on references that only investigate the concept of solar envelopes within the domain of passive design strategy. To do so, some irrelevant references resulting from a discrepancy in research objectives are eliminated. For example, studies related to solar-form finding that focuses only on a generative architectural form [22] [23] [24], solar performance simulation that merely addresses the quantification of solar energy for existing building facades, 2D plans and new development areas [25] [26] [27], and studies related to solar radiation analysis that examine solar potential for the urban contexts [28] [29]. These references are consequently not further included in the main discussion of the review. In addition, this study identifies one previous review of solar envelope properties authored by Stasinopoulos [30]. However, this recent study focuses on the construction of solar envelopes based on solid modeling techniques [31], an area that can benefit from further consideration in our review that systematically addresses (1) a comprehensive study of computational solar envelopes based on various methods, tools, and contextual settings of various case studies, (2) an in-depth investigation of design parameters that correspond to different computational methods and the identification of relevant parameters for each method, and (3) several criteria that enable the further analysis of current gaps and future directions of the study. Therefore, the present review can be a very useful companion to the researchers involved in studying new methods and computational tools for generating solar envelopes. Due to the presented potentialities of solar envelopes, the expected impact of this review at large, is that to help in developing new design tools to increase sustainability, resource efficiency, and livability of our buildings and cities.

After having selected the relevant references that meet the screening criteria, this study proposes a conceptual framework for the review as presented in Figure 2.1.



FIG. 2.1 General framework of the review

2.3 Review findings: Computational methods and parameters of solar envelopes

This section presents the results of the review by investigating the computational methods and the parameters for solar envelopes obtained from the 58 selected references. The computational methods based on design procedures when establishing solar envelopes are first categorized. This is followed by extracting and mapping the parameters from the selected references into the database and then, drawing a comparative analysis among the methods and parameters based on their frequency of use. These actions facilitate the identification of additional characteristics of the parameters and the performance aspects of each design method.

2.3.1 **Design methods**

By examining the basic computational procedures of each collected reference, this study identifies three methods of computational solar envelopes, namely, descriptive geometry (DG), solar obstruction angle (SOA), and constructive solid geometry (CSG).

Descriptive geometry (DG)

As a derivative approach from the descriptive mechanism, the DG method employs basic parameters such as latitude, a closed boundary of selected site, sun vectors, cut-off-times, and obstruction geometry of surrounding buildings, to generate solar envelopes. However, the use of these parameters is highly dependent on the type and objective of the projects. When simulating, for example, no-obstruction solar envelopes, such as parks and open spaces, the relevant parameters only require a closed curve of the site, the latitude and/or sun vectors, and the maximum height of the intended envelope given that the aim is to maximize solar access of the proposed envelope within a specific period. Moreover, solar envelopes that incorporate the surrounding context must consider the geometric obstacles from the existing context. This is pertinent when developing solar rights and solar collection, as the proposed envelopes cannot violate the solar access of neighboring buildings (solar rights) and the surrounding buildings cannot obstruct the sun access of the proposed envelopes (solar collection) [32].

Solar obstruction angle (SOA)

The SOA method simulates the projection of the minimum profile angle on each side of the plot through shadow lines and shadow angles. In this approach, the reference lines refer to the horizontal property lines of the North and the vertical axis, i.e., where the existing building stands in relation to the Cartesian space [33]. Due to daylight and the different orientation on each side of the plot, the resulting geometry of the SOA varies. Higher settings regarding the angle of solar obstruction produce shorter periods of shading and minimum shading conditions. Therefore, each side of the plot should be treated differently because of the shading conditions unique to each side. For example, it is important to minimize solar obstruction in designated areas within a range of \pm 30° for southern facades due to passive solar gains while taking into account latitude position in the UK [24].

Constructive solid geometry (CSG)

The CSG method refers to a solid modeling approach [31] based on the Boolean operations of subtraction and intersection between the 3D plot and the initial shapes of new buildings. These initial shapes are usually generated from two settings of solar vectors. For example, the first solar vector, which may correspond to a solar access constraint or sun position is calculated by using latitude, longitude, and cut-off times. Furthermore, the second solar vector refers to the vectors that are generated from the sun path diagram [34]. In principle, these settings can be used in either a full [35] [36] [37] or a partial way [38] [39] during the construction of solar envelopes depending on the design concept and the complexity of the surrounding contexts. The CSG method can also be complemented with a solar fan to adjust the height and width of the floors of surrounding buildings.

In general, these computational methods have successfully addressed various advanced simulation techniques for establishing solar envelopes. However, these same methods also create dissimilar computational behaviors during the construction of solar envelopes. For example, among the three methods, CSG exhibits the highest rate of generating an unpredictable model in terms of the final configuration of geometric solar envelopes. This is because the CSG method primarily involves the volumetric intersection between different shapes, which means that the probability of yielding unstructured envelopes is also moderately high. Furthermore, but consistent with this, the geometric configuration of CSG requires a high computational performance, especially with respect to the mesh modeling of the contoured plot in larger urban scales. In contrast to the CSG method, however, the DG and SOA methods establish solar envelopes based on the plane intersections that are generated from shadow fences or the obstacle curves of surrounding buildings. Accordingly, solar envelopes that are constructed based on DG and SOA are not generally computationally intensive.

To better understand the characteristics of each computational method when considering the design practices, it is also important to examine the specific elements relevant during the simulation process, such as the basic and advanced parameters of each method, the interrelations between design parameters, and the contextual settings of each method of each specific project. The following section will discuss these aspects based on the collected references.

2.3.2 **Design parameters**

Design parameters include various aspects that determine the geometric and spatial properties of solar envelopes during the simulation process. Hence, this section focuses primarily on identifying each type of parameter included in the selected references. According to its functional typologies on space-time constraints [14], this study further subdivides the parameters into geographic and climatic properties. Geographic properties includes a series of elements that constitute a spatial relationship between a selected plot and the surrounding environment whereas **climatic properties** refer to the non-geometric characteristics that are primarily considered when establishing the time for the construction of the geometric model of the solar envelopes. In total, this study identifies 18 parameters from the collected references; among these parameters, 11 are regarded as geographic properties (i.e., longitude, latitude, orientation, courtyard, surrounding facades, sidewalk, surrounding building's height, FAR (floor area ratio), setback, shadow fences, and street), and seven are regarded as climatic properties (i.e., profile angle, cut-off times, dry bulb temperature, sun path, solar azimuth, solar altitude, and sun access duration). Similar to space and time, these properties also involve an inverse construction mechanism when assessing the volumetric size of solar envelopes. For example, the greater the time interval (cut-off-time) of the solar envelopes, the less the space that is produced inside the envelope. Similarly, the more sun access duration there are, the less the volume of the constructed solar envelope. Regarding the site orientation, the long sides of building with an in EW (east-west) orientation are larger and have higher ridges in the geometric envelopes than the long-sided areas with a NS (north-south) orientation. In contrast, the widths of streets and courtyards on the EW (east-west) sides generate the minimum height for solar envelopes [39].

To draw an in-depth analysis among parameters and methods of solar envelopes, it is necessary to examine the distribution level of each parameter for each corresponding method with respect to quantity and priority.

2.3.3 Comparative analysis of design parameters in relation to design methods

This comprehensive analysis between the design parameters and computational methods of solar envelopes consists of three major tasks. First, each parameter and its corresponding references based on computational methods must be plotted (see Figure 2.2) to provide an overview of the distribution pattern regarding the method and the total number of references of each parameter. Second, the usage frequency

between parameters and their corresponding methods must be plotted (see Figure 2.3). This facilitates the categorization of the parameters into three different groups, namely, high, medium, and low, according to their usage level during the construction of the solar envelopes. Third, the total number of parameters, i.e., geographic and climatic properties, registered for each computational methods are calculated based on the predefined categories discussed (see Figure 2.4). This allows us to identify the most basic and the most advanced parameters as well as the methods used to establish solar envelopes.

To perform these three tasks, Table 2.1 provides the necessary preliminary information regarding the distribution of design parameters and corresponding methods based on the selected references.

Table 1 displays a list of the selected references on computational solar envelopes that are plotted based on their design methods. Each of these references is then further investigated by marking its contextual settings and parameters through a binary operation. For example, the "True" condition is indicated with a bullet point, whereas the "False" condition remains empty. The results illustrate that some references nearly fulfill all of the listed parameters (e.g., [39] [67] [68]), while others require only six parameters ([30][34][71][72]) to construct solar envelopes. This is because each reference is affected not only by the type of applied computational method but also by the design concept of the projects that often involve more parameters. With respect to the contextual settings, the plot condition for solar envelopes is divided into the inclusion and exclusion conditions of surrounding site properties (e.g., vegetation, adjacent buildings, open spaces, and other relevant elements). Accordingly, the solar envelope simulations that include site properties create spatial negotiations between the proposed building and the existing context, while the exclusion condition focuses on the given land parcel, and hence, the context implementations primarily correspond to new development areas. In general, all categorized methods contain both inclusion and exclusion conditions during the construction of solar envelopes. As the most-used method, DG considers site properties more often than SOA and CSG. However, the DG method cannot guarantee the number of involved parameters during the simulation because the input parameters of solar envelopes rely on design concept and project complexity. For example, Camporeale [58] and Saleh and Al-Hagla [55] employ more variations of design parameters than with Machacova et al. [44] and Martin and Keeffe [45], who take into account the surrounding site properties.

Having established the preliminary database regarding design parameters and solar envelope methods presented in Table 1, a comparative analysis of the two can now be performed.

		Input Parameters																	
		Geo	graph	ic pro	pertie	S							Clim	atic p	roper	ties			
Literature	Context*	Longitude	Latitude	Orientation	Court yard	Surrounding facade	Sidewalk	SBH**	FAR	Setback	Shadow fences	Street	Profile angle	Cut-off-times	DBT***	Sun path	Solar azimuth	Solar altitude	Sun access duration
No.		1	2			5	6			9	10	11	12	13	14	15		17	18
Descriptive Geometry (DG)																			
[40][41][42]	+	•	•	•		٠	•	•	•			•		•			٠	•	•
[43]	+	•	•	•		•		•					•	•	•	•	•	•	•
[44]	+		•			•	•	•			•	•		•			•	•	
[45][46]	+		•	•		•		•			•			•			•	•	•
[47]	+	•	•	٠			•		•	•			•	•		•	•	•	
[48]	+	•	•	•		•		•			•			•		•	•	•	•
[49]	+	•	•	•		•		•			•			•		•	•	•	•
[50]	+	•	•	•		٠		•						•			٠	•	٠
[51]	+	•	•	•		٠		•						•			٠	•	٠
[52]	+	•	•	•		٠	•					•		•			٠	•	•
[53]	-		•	•					•		•			•			•	•	•
[54] [55]	-		•	•				•	•					•		•	•	•	•
[56]	-		•	•	•			•		•	•			•			•	•	•
[57] [58]	-		•	•				•	•	•		•		•			•	•	•
[59]	-		•	•							•	•		•			•	•	•
[60]	-		•	•					•					•			•	•	•
[61]	-	•	•	•				•	•		•	•	•	٠		•	•	•	•
[62]	-	•	•	•							•			٠		•	•	•	•
Solar Obstruction Angle (SOA)																			
[63]	+		•	•			•	•	•			•	•	•			•		•
[33]	-	•	•	•		•		•		•			•		•		•	•	•
[64][65][66]	-			•		•		•		•		•	•	•	•	•			•
[67] [68]	+	•	•	•		•	•	•	•	•	•		•	•	•	•	•	•	•
[69]	+		•	•		•		•					•	•	•	•			•
[70]	-	•	•	•				•	•		•		•	•			•	•	•
[71] [72]	+		•	•							•		•	•					•
	/-																		
[73][74]	-					•		•	•	•	•		•	•		•	•	•	
[75][76]	-					•	•	•	•	•	•		•	•	•	•	•	•	•
																			>>>

TABLE 2.1 Database of design parameters and computational method of solar envelopes based on the selected references

		Inpu	t Para	amete	rs														
		Geog	graph	ic pro	pertie	s							Clim	atic p	roper	ties			
Literature	Context*	Longitude	Latitude	Orientation	Court yard	Surrounding facade	Sidewalk	SBH**	FAR	Setback	Shadow fences	Street	Profile angle	Cut-off-times	DBT***	Sun path	Solar azimuth	Solar altitude	Sun access duration
No.		1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18
Constructive Solid Geometry (CSG)																			
[35] [77] [78]	+/-		•					•	•	•		•	•	•			•	٠	•
[34][30]	-		•										•	•			٠	٠	•
[36]	-		•	•				•		•	•		•	•			٠	٠	•
[39]	-		•	•	•	•	•	•		•	•	•	•	•		•	٠	٠	•
[37] [79]	-		•	•				•					•	•			•	•	
[38][80]	+		•	•			•					•	•	•					•
[81]	+	•	•	•		•		•			•			•		•	٠	٠	•
[82]	+	•	•	•		•		•			•			•		•	•	•	•
[83]	+	•	•	•		•		•			•			•		•	٠	٠	•
[84][85]	+	•	•	•		•		•			•	•	•	•			٠	٠	•
[86][15]	+		•	•		•		•			•	•	•	•		•	٠	٠	•
[87]	-		•	•							•			•		•	•	٠	•
[88]	+	•	•	٠	•	٠				•				•		•	٠	٠	•

IABLE 2.1 Database of design parameters and computational method of solar envelopes based on the selected reference	TABLE 2.1	Database of design	parameters and computat	tional method of solar	envelopes based on	the selected reference
---	-----------	--------------------	-------------------------	------------------------	--------------------	------------------------

* Context = (+) include site properties, (-) exclude site properties; **S.B.H = Surrounding building's height; ***D.B.T = Dry Bulb Temperature



Task 1 – Design parameters, methods, and total references

FIG. 2.2 Distribution of design parameters according to corresponding computational methods and selected references

Figure 2.2 illustrates the pattern of computational solar envelopes based on the registered parameters and the number of references for each method. In general, the trend indicates that, with respect to using the parameters of solar envelopes, DG is referenced in more studies that the other two methods. Specifically, DG is referenced in 23 studies, followed by CSG with 20 and SOA with 15. This provides an early indication that DG is the most-used technique for constructing solar envelopes. Additionally, DG includes four parameters, i.e., latitude, orientation, cut-off times, and solar altitudes, thus nearly satisfying all selected references, while SOA and CSG consist of one (profile angle) and two (latitude and cut-off times) parameters, respectively. Moreover, an interesting pattern is observed regarding parameters 4 (courtyard) and 14 (dry bulb temperature). Specifically, these

parameters are similarly registered only in two methods for parameter 4, i.e., DG and CSG, and two methods for parameter 14, i.e., DG and SOA. Hence, it can be argued that courtyard and dry bulb temperature are rarely used parameters and are thus irrelevant properties for SOA and CSG, respectively. This condition also indicates the relevance of parameters that may only be employed in certain context during the construction of solar envelopes.



Task 2 – Usage frequency of parameters and corresponding methods

FIG. 2.3 Categorization of design parameters and corresponding methods based on usage frequency

After quantifying the total references for each parameter and corresponding method in Figure 2.2, the usage frequency of the parameters is divided by the total references for each method into three groups with each group comprising a certain range of references that indicate the usage frequency level of the parameters. Accordingly, the higher the number of references is for one parameter, the higher the frequency of that parameter's use during the construction of solar envelopes. For example, DG contains 23 references that correspond to 18 parameters.

parameter may consist of a different number of references depending on the type of parameters and how many studies that use that parameter. It is worth to note that some parameters may also look identical such as "solar angle" and "solar altitude". In this regard, "solar angle" is not an independent parameter. It is retrieved based on the projection of solar altitude perpendicularly to the border of the plot. Thus, some references may not mention explicitly certain parameters they used. Furthermore, these 23 references are then divided into three range groups, whereby each group represents a different level of frequency; e.g., the high category consists of nine references, followed by the medium and low categories with eight and six references, respectively. To identify the category for each parameter, the specific ranges of total references for each category should be defined first. Specifically, the high category ranges from 15 to 23 references, while the medium and low categories range from 7 to 14 and from 1 to 6 references, respectively. Hence, a specific category for each parameter can be identified and similar approach can now be applied to other methods.

As illustrated in Figure 2.3, the general trend reveals a fluctuating pattern in the percentages for each category of each method. For example, the greatest percentages for each category are identified by different methods. For example, the high category is fulfilled by CSG with 45 %, whereas SOA and DG dominate the medium and low categories with 45% and 33%, respectively. These highest numbers simultaneously represent the priority usage of parameters when constructing solar envelopes, which means that the parameters listed within the high and medium categories serve as basic parameters for establishing solar envelopes. Moreover, the low category, due to its lowest usage values, especially for SOA and SCG is a list of the advanced and complementary parameters, and accordingly, only a few parameters can be used in this category such as longitude and street for SOA and courtyard, sidewalk, and FAR for CSG. An exception applies to DG because it contains more assigned parameters and references, i.e., nearly 18 % above the other values. Based on these data, it is concluded that DG is the most-used method, as evidenced by its wide range of options for complementary parameters. Moreover, the small discrepancies in the values between each category in DG results in greater flexibility with respect to switching parameters when establishing solar envelopes.





FIG. 2.4 Specific distribution of design parameters (geographic and climatic properties) based on the group usage frequency and corresponding methods

The plot of the parameters illustrated in Figure 2.4 requires further explanation. First, denoting the highest usage frequency, the high category confirms the greatest numbers of parameters that are incorporated in all methods. This category consists of four parameters (i.e., site orientation, surrounding building height, cut-off times, and sun access duration), whereas the remaining parameter, shadow fences, is found only in the medium category. Based on the similarity of the three methods, these shared parameters can be further defined as global parameters. However, there are several parameters, such as surrounding facades, set back, latitude, sidewalk, dry bulb temperature, solar azimuth, solar altitude, longitude, and street, that are included only in particular methods. As these parameters specifically correspond to SOA, they are defined as local parameters. To some extent, these local parameters act as basic parameters when they are located in the high-frequency use category (red dashed box) while at the same time, they are considered nonstandard parameters when implemented in other methods. For example, setback is a regular parameter for SOA, but it is regarded as an advanced or nonstandard parameter for DG. This is because SOA requires different daylight conditions on each side of the plot and is therefore influenced by the setback and plot orientation, whereas DG treats the same condition based on the whole boundary of the plot. Another interesting trend is the DG's local parameters (i.e., setback, solar angle, dry bulb temperature, and sun path), which simultaneously become advanced parameters because they are in the low-frequency use category. As a the most frequentlyused method, this trend indicates that DG exhibits a higher degree of complexity, especially when comparing the quantity of DG's local parameters to those of the other two methods. Thus, it is worth noting that the specific parameter of a solar envelope plays a great role in determining the computational workflow of the simulation method.

Second, after plotting the parameters based on the usage frequency, the number of geographic and climatic properties for each method can be determined. The general trend indicates that only the parameters from both categories are found in the high category while only parameters that satisfy the geographic properties are assigned to the other two methods. Specifically, climatic properties are absent from the medium (DG) and low categories (SOA and CSG). This is not only because the total number of assigned parameters in the high category is greater than that in the others but also because the medium and low categories are populated with rarelyused parameters.

2.4 Discussion: digital simulation tools and case studies

An investigation of the computational environment of solar envelopes focuses on two qualities, namely, digital simulation tools and implementation of case studies (see Table 2.2), which are specifically investigated based on computational criteria and typologies of the projects, respectively. To conduct this investigation, each selected reference is evaluated by using a similar binary operation, as in the previous section, and the evaluation is conducted based on predefined computational parameters. For example, the digital tools of each selected reference are first investigated, and the evaluation criteria are then established based on four relevant factors i.e., self-developed tools, dynamic-parameter input, modeling environment, and integrated environmental simulation. The project implementations (case studies) are then specified according to two aspects, namely, architectural scales (i.e., urban, open space, and single building) and functional utilities (i.e., housing, offices, and commercial). The distributions of these qualities are subsequently plotted onto the bar graph, as illustrated in Figure 2.5. The purpose is to identify the performance of each computational method in relation to the number of corresponding references for each criterion. In so doing, a better overview regarding the computational workflow of solar envelopes is developed, especially with respect to the context of design practices available for architectural use.

Literature	Digital tools	Con	nputa	tiona		Case studies								
		crite	eria			Arc	hitect	ural	Fun	ctiona	al util	ities		
						scal	es		Hou	sing				
		А	В	С	D	E	F	G	н	I	J	К	L	М
Descriptive Geon	netry (DG)													
[40][41][42]	SustArc	•		•	•	•		•			•			
[43]	Rhino, Grasshopper (Ladybug)		•	•	•			•				•	٠	
[44]	-			•		•		•			•	•		
[45][46]	CAD			•		•		•				•	٠	•
[47]	-						•					•		
[48]	Rhino, Grasshopper (Ladybug, Octopus)		•	•	•			•		•				
[49]	Rhino, Grasshopper (Ladybug)		•	•	•			•		•				
[50]	T4SU, Sketchup, GIS		•	•				•					•	
[51]	AutoCAD			•				•	•				•	•
[52]	Heliodon, Ecotect		•	•	•	•		•	•				•	•
[53]	TAS (EDSL v. 9.09)		•	•	•			•		•				
[54][55]	Rhino, Grasshopper, Diva, Ecotect, Vasari		•	•	•	•					•			
[56]	CalcSolar (Autolisp)- Autocad			•			•	•			•			
[57] [58]	Rhino, Grasshopper, Ecotect, Galapagos		•	•	•	•					•			
[59]	-				•	•		•		•				
[60]	AutoCAD, Sketchup, 3D Max		•	•				•	•					
[61]	-				•		•	•	•				•	•
[62]	Autodesk			•		•					•			
Solar Obstruction	n Angle (SOA)													
[63]	-	•			•	•	•			•			•	•
[33]	The Obstrucao 1.0	•		•	•			•	•					
[64][65][66]	MascaraW	•		•	•	•		•			•			
[67][68]	CityZoom (Block magic 3D)	•		•	•	•		•				•		•
[69]	-					•		•					٠	
[70]	Envi-met (thermal analysis), PMV		•	•	•		•	•	•					
[71] [72]	CAD-Microstation			•	•		•	•			•			
[73][74]	BRADA	•		•	•	•		•			•			
[75] [76]	City SHADOWS, Envi-met	• • •				•		•		•				

TABLE 2.2 Database of the computational environment parameters of solar envelopes based on the selected references

>>>

Literature	Digital tools	Computational				Case studies									
		criteria			Arcl	nitect	ural	Functional utilities							
						scal	es		Hou	sing					
		Α	В	С	D	Е	F	G	н	Ι	J	K	L	М	
Constructive Solar Geometry (CSG)															
[35] [77] [78]	Solar envelopes tools + BSK			•	•	•		•			•				
[34][30]	AutoCAD 2000			•		•		•				•		٠	
[36]	CalcSolar (Autolisp)-Autocad	•		•				•				•			
[39]	GIS, EnergyPlus 8.1			•	•	•		•		•					
[37]	Form.Z			•				•				•			
[79]	SolCAD	•		•			•	•				•			
[38] [80]	Rhino, Grasshopper, EnergyPlus		•	•	•			•				•	•		
[81]	Rhino, Grasshopper, Ladybug		•	•	•			•	•				•		
[82]	Rhino, Grasshopper, Ladybug		•	•	•			٠	•			•	•		
[83]	Rhino, Grasshopper, Ladybug		•	•	•			•	•				•		
[84] [85]	PIRAMIDA	•		•	•			•		•					
[86][15]	Autodesk's 3dsmax™			•	•	•		•				•	•		
[87]	AutoCAD			•	•			•		•					
[88]	Rhino, Grasshopper, Ladybug		•	•	•			•	•				•	•	

TABLE 2.2 Database of the computational environment parameters of solar envelopes based on the selected references

A: Self-developed; B: Dynamic parameter input; C: Parametric modeling environment; D: Integrated environmental simulation; E: Urban; F: Open space; G: Single building; H: Discontinued collective; I: Continued collective; J: Dense individual; K: Dispersed individual; L: Offices; M: Commercial.



FIG. 2.5 Distribution of computational environment parameters according to the corresponding computational methods

2.4.1 Digital tools

The simulation tools in each selected reference to construct solar envelopes are examined. According to Anderson [89], in *Design Energy Simulation for Architects*, the minimum requirement for design tools to run performance simulations consists of three components, specifically, a user interface, three-dimensional modelers, and an engine. Accordingly, several criteria are established to identify a general pattern of simulation platforms in relation to the computational methods of solar envelopes. simulation method.

Self-developed tools

Self-developed refers to the ease of access of the tools during the creation of solar envelopes. These tools are then defined according to three specific criteria. The first category of tools includes those preferred by authors who use popular CAD-related platforms to run the simulations of solar envelopes (e.g., [45] [46] [51] [60] [71] [72] [34] [30] [87]). The second category consists of a custom-built module that is generated from a particular function of existing digital tools such as T4SU in SketchUp [50], SolCAD [79] and Calcsolar [56] in Autocad, solar envelope tools in Revit [36], and the Solar Toolbox plugin in Grasshopper [90]. Third, a tailor-made supporting tools that perform specific tasks related to solar envelopes. Among these tools are MascaraW [66], SustArc [40], CityZoom [68], Calcsolar-AutoLISP [56], The Obstrucao 1.0 [33], CitySHADOWS [94,95], form Z [37], and PIRAMIDA [84] [85]. Note that some references are found to be incomplete due to limited information during the review process, i.e., [44] [47] [59] [61] [63] [69].

When these criteria are plotted with their corresponding methods and references in Figure 2.5 (see point A), it is evident that SOA exhibits the higher number of references, followed by DG and CSG given that SOA consists of the greatest number of local parameters. As previously elucidated in Figure 2.4, local parameters represent a series of parameters that only attach to a particular method due to their scarcity and complexity of use. Accordingly, SOA requires particular tools to formulate the appropriate parameters when constructing solar envelopes.

Dynamic parameter input

This criterion emphasizes the flexibility between the fixed and adaptable-parameter algorithms. The fixed-parameter algorithm often includes both a static and a limited number of parameters due to the default system of design tools. Consequently, the end user of the tool can only follow the simulation procedure and input the dataset on the basis of the given parameters [36]. Some design tools, however, consist of adaptable or dynamic parameters that permit additional tasks, such as the reduction in the number of and generation of relevant parameters. These tasks provide a direct interaction between the users and the tools when developing a solar envelope simulation.

To create a legible representation, Figure 2.5 (see point B) specifically illustrates a bar graph of references that corresponds to the dynamic parameter inputs. The graph indicates that DG has the highest number of dynamic parameter input references, followed by CSG and SOA. This trend is relevant to all methods, however, as DG simultaneously also consists of the largest quantity of references. As the most frequently-used and flexible method (see Figure 2.3), DG provides great accessibility for using the existing digital tools during the construction of solar envelopes. Table 2.2 (see the digital tools) illustrates that DG predominantly uses a well-known tool with a wide input of parameters.

Modelling environment

According to the geometric representation, the modeling environment of the selected references is predominantly generated based on NURBS (non-uniform rational B-spline) models that range from the organic free-form surface to the 3D solid model [31]. The NURBS models are further divided into parametric and direct modeling approach models. These approaches not only differ with respect to design rule and process but also in the complexity of geometrical parameters. Accordingly, the geometric configuration of solar envelopes is dependent on the applied algorithm of the modeling approach. For example, the surface representation of the 3D model can geometrically vary when generated from the TIN (triangulation irregular network) of the point cloud compared to one that is manually created based on the CAD platform [91].

According to Figure 2.5 (see point C), the total number of references for the DG and CSG methods are equally proportioned and outnumber the studies the reference for the SOA method. This trend represents the total selected references for all methods, except some references for unidentified tools. This is because all methods use design tools with parametric functions during the construction of solar envelopes. Nonetheless, further research is required to identify the geometrical behaviors generated by the interaction of parameters.

Integrated environmental performances

Interoperability plays an important role during the design process, especially when dealing with various simulation tools and multiple dataset sources. While this can create, to some extent, a computational issue due to different algorithmic operations, a comprehensive analysis for optimal design solution can be achieved. With an integrated environmental simulation, the computational functions of certain design tools, such as solar thermal exposure [61], wind analysis [42], daylight availability, solar photovoltaic exposure, ventilation enhancement, and water surface catchment and flow [92], can be extended during the construction of solar envelopes.

In general, the trend in Figure 2.5 (see point D) illustrates high percentages regarding the use of environmental performances in all methods. This results means that most of the selected studies performed one or more environmental simulation during the construction of solar envelopes. According to the stage of use during the simulation process, these environmental performances can be categorized into three functions, namely, generator, evaluator, and generator, as well as evaluators that are operated in the same workflow. For example, first, the performance generator is used to support the main parameter to establish the final geometry of solar envelopes. Some example of performances can be observed in DG that includes direct sun access duration, temperature [43], annual space heating demands, daylight, thermal performances, solar renewables [53], wind analysis [59], visual assessment, street network [55] [54], and solar irradiation [58] [57], while in SOA urban heat islands [71] [72] and daylight [73] [74], and CSG consists of sun hours availability [88]. Second, the performance evaluator is employed to assess the final geometry of solar envelopes. In other words, this process measures the environmental impacts of new envelopes and compares those impacts to previous and existing conditions. Examples of these criteria are found in DG with performances that involve urban density, direct sun access duration [49], and solar irradiation [52], SOA with energy consumption [33], temperature, wind, albedo, thermal comfort [70], urban heat island, and daily direct solar radiation [76] [75], CSG with aeshthetics, solar access, lighting, ventilation, public safety [35] [78] [77], solar access hours, annual energy consumption, cost, CO2e [38] [80], urban density [15] [86], and sunlight and shading simulation [87]. Third, the performance generator and evaluator consist of a combination of two types of performances that are operated simultaneously in one workflow. For example, some studies on DG use performances such as urban density, energy consumption [32] [41] [42], temperature, wind, climate, and energy [61], while references in the SOA category consider insolation hours, urban density [63], comfort issues and wind flows [67] [68], and those in CSG involve more performances related to air temperatures, global radiation, passive solar gains, heating loads, insolation values [39], sun access duration [81], building density [82], and sphere view factors [83].

2.4.2 Case studies

To identify the contextual settings of computational solar envelopes in design practices, this study separates and plots the selected references in two functions, namely, architectural scales and functional utilities. According to Figure 2.5 (see points E-G), case studies of solar envelopes are predominantly implemented in single buildings since the building-oriented context requires fewer geographic parameters

than do urban scales. For example, the CSG method has the greatest number of references with a single building context (see Figure 2.5 point G), but it receives the lowest rate for implementation in urban contexts (see Figure 2.5 point E) because the modeling construction of CSG is more appropriate for building contexts than for urban-scale contexts due to the high cost of computational issues. The context of open space, however, has received less attention. In fact, open space only seems to play an essential role during the construction of solar envelopes, particularly in highly dense areas such as metropolitan cities.

The functional utilities of the projects are divided into three types, namely, housing, offices, and commercial. The housing category includes those typologies proposed by Maizea et al [93] and thus, consists of the *discontinued collective, continued collective, dense individual*, and *dispersed individual*. In general, the trend of this housing category suggests that the 'individual' groups are referenced more frequently than the 'collective' groups due to the complexity of the projects and scale of the plots. Accordingly, the DG and CSG method are referenced more often in the dense individual and dispersed individual groups, respectively.

With respect to the comparison of housing group and other functions, such as offices and commercial properties, Figure 2.5 illustrates that of all the functional utilities, housing is the one most referenced, even though most people who live in big cities or dense areas spend far more time in offices during the day. Accordingly, solar envelopes are crucial to providing sunlight penetration to the working space to reduce energy consumption during working hours. Consistent with this fact, solar envelopes also play an important role in determining specific conditions of commercial areas. For example, as direct sunlight can affect food and product durability issues, especially when located in ground floor level storefronts, shading becomes a critical factor. However, as some references do not include a specific function for the implementation of a case study, it is challenging to understand the relationship between the contextual settings of these references and performance criteria of the solar envelopes.

2.5 Knowledge gaps and new directions

Based on the understanding of the computational methods of solar envelopes presented in the previous sections, this study identifies several gaps that may drive further research for new approaches to the generation of solar envelopes (see Table 2.3). These gaps are formulated into three aspects, namely, 3D contextual model, climatic properties, and geometric configuration. The proposed directions are also discussed in relation to each gap.

TABLE 2.3 Know	ledge gaps and new	directions for solar envelopes	
No.	Qualities	Knowledge gaps	Future directions
1	3D contextual model	Limited discussion on covering contextual geometries	DEM (digital elevation modeling) Point cloud data
		Limited understanding of site characteristics information	
2.	Climatic properties	Predominantly based on four-season countries	Tropical countries
		The objective is to collect direct sunlight	The objective is to avoid direct sun access
		Predefined period only relies on cut-off times	Consider sun visibility on each period
3.	Geometric configuration	Limited results on final geometry of solar envelopes	Multi objective optimization
		Limited performance criteria	Integrate multi performance criteria (e.g. material)
		Focuses only on 3D mass of solar envelopes	Explore performance configuration of the layout of the building's interior.

3D contextual model

As previously described, most of the current methods employ solid modeling as a platform of the 3D contextual model. Most important when considering this approach is the challenge to comprehensively understand the characteristics of the existing contexts, especially when dealing with complex sites. The current approach to 3D site modeling often not only fails to preserve geometric aspects of existing context but also fails to sufficiently address the surrounding site properties, such as vegetation or other temporal site elements that may be relevant for further analyses of solar envelopes. Moreover, the surface characteristics of the existing environment, such as

the material of surrounding facades, have also received less attention to date. That said, it is argued that the calculations of solar energy within solar envelopes should take into account the surface characteristics of the surrounding environment.

An alternative to the aforementioned issues is DEM (digital elevation modeling). In comparison with other solar envelope methods that are created primarily by CAD drawings, the DEM platform employs image processing techniques to obtain and quantify a solar exposure map by means of shadow volumes ([94][95]). This approach includes iso-solar rights and iso-solar collection surfaces to implement energy-oriented shapes in urban environments. As the current DEM method predominantly focuses on the urban scale, it remains challenging to identify and calculate specific geometric parameters, such as building scales. Another consideration is the 3D laser scanning technologies that offer opportunities to capture the physical properties of the environment. As a product of laser scanner, potential applications of point cloud data may counterbalance relevant information within the surrounding context using geometric and radiometric properties.

Climatic properties

With respect to the climatic parameters found within the collected literature, the existing studies are based primarily on four-season countries. This means that their objectives focus on minimizing sun access duration during summer while maximizing it during winter so that the sunlight can penetrate the main activity room. In fact, these objectives differ significantly from those of tropical countries, especially for those countries located on the equator. Since tropical countries consist of annual wet and dry seasons, these climatic factors affect the objectives and mechanisms of solar envelopes, and accordingly, solar envelopes should be able to minimize the sunlight coming into the house due to high temperatures. For example, building constructions in Indonesia prefer shaded conditions to lower the hot temperatures inside the building. Accordingly, the concepts and existing parameters of solar envelopes require further adjustments for tropical contexts.

Geometric configuration

During the schematic design phase, it is often important to analyze the solar access of new buildings when selecting the optimal layouts for massing that fulfils the volumetric shape of solar envelopes. Accordingly, the solar collection envelope (SCE) [32] and solar collection multi-isosurface [49] have been developed. However, solar collection surfaces can only be used for single buildings with rectangular or convex footprint layouts. Thus, further research is needed to identify optimal massing and layouts for articulated buildings and clusters in urban environments. Hence, the concept of multi objective optimization is useful for exploring geometric design configurations of solar envelopes to identify the optimal solution.

2.6 **Conclusion**

This study presents a conceptual review of solar envelopes by investigating the qualities of design methods and computational performance aspects in relation to parameters, digital simulation tools, and implementation of the case studies. In particular, 58 selected references of solar envelopes are extensively examined as the basis idea to perform comparative analysis between each categorized method and predefined criteria. This study ultimately allows architects not only to identify different characteristics and levels of complexity for each design method but more importantly also to address the concept of solar envelopes in design practices such as the projects of P15 Ravel Plot and Grotius Towers II by the Dutch architectural and urban design firm MVRDV. As a research framework, the present study may also benefit further for urban planner and related municipality to update the current parameters of local regulation especially related to solar energy building performances and environmental design assessments between proposed building and existing environment. Specific remarks on each section in this review are presented as follows:

- By categorizing the contextual setting of solar envelopes into the inclusion and exclusion of surrounding properties (e.g., vegetation, adjacent buildings, open spaces, and other relevant elements) enables architects to identify the types of methods that predominantly focus on new or existing contexts. Given that urban densities may have scarcity of wide areas, DG plays an essential part to deal with the future scenarios as it considers more site properties than other methods.
- Categorization of design parameters into geographic and climatic properties allows us to identify specific parameters that affect volumetric size of solar envelopes for each design method.

- The comparative analysis among methods and parameters indicates that DG is the most frequently-used method of the three. This is because DG has the greatest number of registered references and thus, it contains more basic parameters (latitude, orientation, cut-off times, and solar altitudes) as compared to other methods. In addition, DG has the greatest flexibility to switch parameters during the establishment of solar envelopes because of its wide range of complementary parameters.
- This study categorizes SOA and CSG method as a group with the low category parameters and thus, it refers to local parameters because their parameters can only apply to particular cases when establishing solar envelopes.
- This study investigates the geometric performance of each solar envelope method with respect to the predefined criteria of the digital tools. For example, SOA is identified as the method with the greatest use of self-developed tools since it has the greatest number of local parameters. In contrast, DG is the most flexible for constructing solar envelopes due to its great accessibility, its ability to use the existing digital tools, and its wide range of dynamic parameter inputs.
- This study identifies that CSG is predominantly implemented in a single building rather than on an urban scale due to the high cost of computational modeling and the mesh generation procedures. Moreover, this study reveals that housing remains a predominant case study of solar envelopes, even though offices and commercial sectors consume a greater portion of urban functions, especially in dense areas.

Furthermore, although the conceptual framework of computational solar envelopes is extensively addressed in this review, there is still a need for an objective evaluation approach to provide a quantitative analysis of different methods. By using a similar set of predefined parameters, digital tools, and case study, volumetric shape and performance criteria of geometric solar envelopes on different methods can be further measured more precisely.
Acknowledgments

The authors would like to thank TU Delft Library for supporting logistics of this publication.

References

- H. Ritchie and M. Roser, "Urbanization," 2020. [Online]. Available: https://ourworldindata.org/ urbanization. [Accessed 30 1 2020].
- [2] United Nations, "World Urbanization Prospects: The 2018 Revision (ST/ESA/SER.A/420)," Department of Economic and Social Affairs, Population Division, New York, 2019.
- [3] IEO, "International Energy Outlook 2019 with projections to 2050," U.S. Energy Information Administration Office of Energy Analysis, U.S. Department of Energy, Washington, 2019.
- [4] G. A. f. B. a. Construction, I. E. Agency and t. U. N. E. Programme, "2019 Global status report for building and construction: Towards a zero-emission, efficient and resilient buildings and construction sector," United Nations Environment Programme, Madrid, 2019.
- [5] A. P. Chan, A. Darko and E. E. Ameyaw, "Strategies for promoting green building technologies adoption in the construction industry - An international study," *Sustainability*, vol. 9, no. 6, p. 969, 2017.
- [6] L. Martin and F. Perry, "Sustainable construction technology adoption," in Sustainable construction technologies. Life-cycle assessment, Oxford, Butterworth-Heinemann, 2019, pp. 299-316.
- [7] P. Anand, C. Sekhar, D. Cheong, M. Santamouris and S. Kondepudi, "Occupancy-based zone-level VAV system control implications on thermal comfort, ventilation, indoor air quality and building energy efficiency," *Energy and Buildings*, vol. 2014, no. 1 Desember, p. 109473, 2019.
- [8] M. Shin, J.-C. Baltazar, J. S. Haberl, E. Frazier and B. Lynn, "Evaluation of the energy performance of a net zero energy building in a hot and humid climate," *Energy and Buildings*, vol. 204, no. 1 Desember, p. 10953, 2019.
- M. W. Ahmad, M. Mourshed, B. Yuce and Y. Rezgui, "Computational intelligence techniques for HVAC systems: A review," *Building simulation*, vol. 9, no. August 2016, pp. 359-398, 2016.
- [10] J. Reijula, R. Holopainen, E. Kähkönen, K. Reijula and I. D. Tommelein, "Intelligent HVAC systems in hospitals," Intelligent building international, vol. 5, no. 2, pp. 101-119, 2013.
- [11] M. Winy, "What's Next?: how do we make vertical urban design?," in Council on tall buildings and urban habitat (CTBUH), Shenzen, 2016.
- MVRDV, "Grotius Towers," MVRDV, - 2019. [Online]. Available: https://www.mvrdv.nl/projects/392/grotiustowers. [Accessed 27 05 2020].
- [13] B. Topaloglu, Solar envelope and form generation in architecture, Graduate School of natural and applied science of the middle east of technical university, 2003.
- [14] R. L. Knowles, Sun, rhytm and form, Massachusetts: The MIT Press, 1981.
- [15] M. White, "Preserving open space amenity using subtractive volumetric modelling," in Aachener Geographische Arbeiten, Aachen, 2014.
- [16] A. Martín-Martín, E. Orduna-Malea, M. Thelwall and E. D. López-Cózara, "Google Scholar, Web of Science, and Scopus: A systematic comparison of citations in 252 subject categories," *Journal of Informetrics*, vol. 12, no. 4, pp. 1160-1177, 2018.
- [17] K. Yang and L. Meho, "Citation Analysis: A Comparison of Google Scholar, Scopus, and Web of Science," Proceedings of the American Society for Information Science and Technology, vol. 43, no. 1, pp. 1-15, 2007.
- [18] R. L. Knowles, Energy and form: An ecological approach to urban growth, Cambridge: The MIT Press, 1974.
- [19] L. Giacomo, The architecture of A. Palladio in Four Books, London: John Watts, 1715.
- [20] D. S. Galton, Healthy hospitals : observations on some points connected with hospital construction, 1 ed., London: Oxford : At the Clarendon Press, 1893.
- [21] W. Atkinson, The orientation of buildings : or, Planning for sunlight, 1 ed., New York: J. Wiley & sons, 1912.
- [22] M. Leidi and A. Schlüter, "Exploring urban space Volumetric site analysis for conceptual design in the urban context," *International journal of architectural computing*, pp. 157-182, 2013.

- [23] J. Da Veiga and P. La Roche, "A Computer Solar Analysis Tool for the Design and Manufacturing of Complex Architectural Envelopes: EvSurf," in *Proceedings of the 6th Iberoamerican Congress of Digital Graphics* [SIGraDi 2002], Caracas, 2012.
- [24] P. Littlefair, "Passive solar urban design: ensuring the penetration of solar energy into the city," *Renewable and Sustainable Energy Reviews*, vol. 2, no. 3, pp. 303-326, 1998.
- [25] B. S. Obaidat, "A Simulation Model for Defining Three Different Solar Accesses in Site Planning," Simulation, vol. 65, no. 6, pp. 357-371, 1995.
- [26] A. Jain, K. Kensek and D. Noble, "Interactive Web-based teaching tool for simplified 3D analysis of solar rhythms," *Automation in Construction*, vol. 8, no. 2, pp. 181-194, 1998.
- [27] R. V. Ralegaonkar and R. Gupta, "Review of intelligent building construction: A passive solar architecture approach," *Renewable and Sustainable Energy Reviews*, vol. 14, no. -, pp. 2238-2242, 2010.
- [28] S. Freita, C. M. Catita, P. Redweik and M. Brito, "Modelling solar potential in the urban environment: State-ofthe-art-review," *Renewable and sustainable energy reviews*, vol. 41, no. -, pp. 915-931, 2015.
- [29] G. Lobaccaro, F. Frontini, G. Masera and T. Poli, "SolarPW: A new solar design tool to exploit solar potential in existing urban areas," *Energy Procedia*, vol. 30, pp. 1173-1183, 2012.
- [30] T. N. Stasinopoulos, "A survey of solar envelope properties using solid modelling," *Journal of Green Building*, vol. 13, no. 1, pp. 3-30, 2018.
- [31] N. N. Staneva, "Approaches for generating 3D solid models in AutoCAD and solid works," *Journal of Engineering*, vol. VI, no. 3, pp. 28-31, 2008.
- [32] G. Capeluto and E. Shaviv, "Modeling the design of urban fabric with soalr rights considerations," Taejon, 1997.
- [33] R. S. Brandao and M. P. Alucci, "Solar access in tropical cities: towards a multicriteria solar envelope," in The 22nd Conference on passive and low energy architecture, Beirut, 2005.
- [34] T. N. Stasinopoulos, "Solar envelope a construction method using AutoCAD 2000," 9 July 2001. [Online].
 Available: http://www.oikotekton.eu/solenvelope. [Accessed 25 October 2016].
- [35] K. Raboudi and A. B. Saci, "A morphological generator of urban rules of solar control," in *The 29th conference Sustainable architecture for a renewable future*, Munich, 2013.
- [36] K. Kensek and A. Henkhaus, "Solar access zoning + building information modeling," in Solar 2013, Baltimore, 2013.
- [37] J. F. Cotton, "Solid modeling as a tool for constructing solar envelopes," Automation and Construction, vol. 5, no. 3, pp. 185-192, 1996.
- [38] J. Niemasz, J. Sargent and C. F. Reinhart, "Solar zoning and energy in detached dwellings," in SimAUD, Boston, 2011.
- [39] A. Vartholomaios, "The residential solar block envelope: a method for enabling the development of compact urban blocks with high passive solar potential," *Energy and Buildings*, vol. 99, pp. 303-312, 2015.
- [40] I. G. Capeluto and E. Shaviv, "Modelling the design of urban fabric with solar rights considerations," in International Conference of IBPSA, Kyoto, 1997.
- [41] I. Capeluto, A. Yezioro, T. Bleiberg and E. Shaviv, "From computer models to simple design tools: solar rights in the design of urban streets," in *Ninth international IBPSA conference*, Montreal, 2005.
- [42] I. G. Capeluto, A. Yezioro and E. Shaviv, "Climactic aspects in urban design a case study," *Building and environment*, vol. 38, no. 6, pp. 827-835, 2003.
- [43] G. I. Capeluto and B. Plotnikov, "A method for the generation of climate-based, context-dependent parametric solar envelopes," *Architectural science review*, vol. 60, no. 5, pp. 395-407, 2017.
- [44] K. Machacova, J. Keppl and L. Krajcovics, "The solar envelope method in education at the faculty of architecture STU Bratislava," in *Central Europe towards sustainable building*, Prague, 2013.
- [45] C. L. Martin and G. Keeffe, "The Biomimetic solar city: solar derived urban form using a forest-growth inspired methodology," in *The 24th Conference on passive and low energy architecture*, Singapore, 2007.
- [46] C. L. Martin, M. Pilling, C. Stott and V. Walsh, "The nectar project: solar development of post-industrial urban communities," in *The 27th conference on passive and low energy architecture*, Louvain-la-Neuve, 2011.
- [47] M. Dekay, "The implications of community gardening for land use and density," *Journal of architectural and planning research*, vol. 14, no. 2, pp. 126-149, 1997.
- [48] F. De Luca, "Solar envelope optimization method for complex urban environments," in CAADence in Architecture, Budapest, 2016.
- [49] F. De Luca and H. Voll, "Computational method for variable objectives and context-aware solar envelopes generation," in SimAUD, Toronto, 2017.

- [50] T. Leduc and K. Hartwell, "Limiting the buildings' envelopes in order to prevent the surrounding mask effect: towards an efficient implementation in the context of SketchUp," in *PLEA, Design to Thrive*, Edinburgh, 2017.
- [51] F. Canan and M. B. Tosunlar, "The implementation of sustainable approach in the architectural design studio developing architectural designs using solar envelope methods," *ICONARP, International Journal of Architecture and Planning*, vol. 4, no. 1, pp. 14–33, 2016.
- [52] R. Franco and B. Beckers, "A study of solar access in Bogotá: the Las Nieves neighborhood," in First International Conference on Urban Physics, Quito – Galápagos, 2016.
- [53] G. Bruce, "High density, Low energy: achieving useful solar access for Dublin's multi-storey apartment developments," in *PLEA*, Dublin, 2008.
- [54] M. M. Saleh and K. Al-Hagla, "Parametric Urban Comfort Envelope: An Approach towards a Responsive Sustainable Urban Morphology," in *Proceeding of ICSAUD 2012 : International Conference on Sustainable Architecture and Urban Design*, Venice, 2012.
- [55] M. M. Saleh and K. S. Al-hagla, "Parametric urban comfort envelope: an approach toward a responsive sustainable urban morphology," *International journal of civil, environmental, structural, construction and architectural engineering*, vol. 6, no. 11, pp. 930-937, 2012.
- [56] D. Noble and K. Kensek, "Computer generated solar envelopes in architecture," *The Journal of architecture*, vol. 3, no. 2, pp. 117-127, 1998.
- [57] P. Camporeale, "Genetic algorithms applied to urban growth optimizing solar radiation," in Proceeding of PLEA 2013 29th Conference Sustainable Architecture for a Renewable Future, Munich, 2013.
- [58] P. Camporeale, "Genetic algorithms applied to urban growth optimization," in *Proceeding of eCAADe Computation and Performance 2013*, Delft, 2013.
- [59] T. Sorayaei and Z. Sorayaei, "An integrated approach to climate conscious urban design using solar envelope concept," *Palma journal*, vol. 16, no. 2, pp. 322-330, 2017.
- [60] A. A. M. Jaff, "Solar envelope method and consideration of the effectiveness of construction density and settlement in Konya," *Journal of solar energy research*, vol. 2, no. 3, pp. 27-31, 2017.
- [61] Y. Mert and N. Saygin, "Energetic and Exergetic Design Evaluations of a Building Block Based on a Hybrid Solar Envelope Method," in *Exergy for A Better Environment and Improved Sustainability 2*, Cham, Springer International Publishing, 2018, pp. 515-531.
- [62] H. Nazer and L. Rodrigues, "Solar access in high density urban developments: A representative case in Matlock," in *Proceeding of PLEA 2015*, Bologna, 2015.
- [63] I. G. Capeluto, A. Yezioro, T. Bleiberg and E. Shaviv, "solar rights in the design of urban spaces," in *The 23rd conference on passive and low energy architecture*, Geneva, 2006.
- [64] F. O. R. Pereira and C. A. Nome Silva, "A proposal for the implementation of the solar envelope in urban planning as a concept for regulating the occupation of urban area," in *Proceedings PLEA 98 – The 15th International Conference on Passive and Low Energy Architecture*, Lisbon, 1998.
- [65] F. O. R. Pereira and C. A. Nome Silva, "Sunlighting in the urban design: A computer-based method for solar and sky vault obstruction control," in *Proceedings PLEA 96 - The 13th International Conference on Passive* and Low Energy Architecture, Louvain-la-Neuve, 1996.
- [66] F. O. R. Pereira, C. A. N. Silva and B. Turkienikz, "A methodology for sunlight urban planning: a computerbased solar and sky vault obstruction analysis," *Solar energy*, vol. 70, no. 3, pp. 217-226, 2001.
- [67] B. Turkienicz, B. B. Goncalves and P. Grazziotin, "CityZoom:A Visualization Tool for the assessment of planning regulations," *International journal of architectural computing*, vol. 6, no. 1, pp. 79-95, 2008.
- [68] P. C. Grazziotin, F. O. R. Pereira, C. M. D. S. Freitas and B. Turkienicz, "Integration of Sunlight Access Control to Building Potential Simulator," in *The Ibero-American Symposium on Computer Graphics*, Guimaraes, 2002.
- [69] M. D. G. V. D. Amaral, "The application of solar architecture in the planning of the campus," in The 2005 World sustainable building conference, Tokyo, 2005.
- [70] B. Paramita and M. Koerniawan, "Solar envelope assessment in tropical region building case study: vertical settlement in Bandung, Indonesia," in *The 3rd International conference on sustainable future for human security SUSTAIN 2012*, Kyoto, 2013.
- [71] R. Emmanuel, "A hypothetical 'shadow umbrella' for thermal comfort enhancement in the equatorial urban outdoors," *Architectural science review*, vol. 36, no. 4, pp. 173-184, 1993.
- [72] R. Emmanuel, H. Rosenlund and E. Johansson, "Urban shading a design option for the tropics? A study in Colombo, Sri Lanka," *Inernational Journal of Climatology*, vol. 27, no. 14, pp. 1995-2004, 2007.
- [73] M. Dekay, "Daylighting and urban form: an urban fabric of light," *Journal of architectural and planning research*, vol. 27, no. 1, pp. 35-56, 2010.

- [74] M. Dekay, "Climatic urban design: configuring the urban fabric to support daylighting, passive cooling, and solar heating," *The sustainable city VII*, vol. 155, pp. 619-630, 2012.
- [75] A. Okeil, "A holistic approach to energy efficient building forms," *Energy and buildings*, vol. 42, no. 9, pp. 1437-1444, 2010.
- [76] A. Okeil, "In Search for Energy Efficient Urban Forms: The Residential Solar Block," in BUilding for the future: The 16th CIB world building congress 2004, Rotterdam, 2004.
- [77] K. Raboudi and A. B. Saci, "Conditions of satisfaction of the solar control box constraints," *Journal of Civil Engineering and Architecture*, vol. 12, no. October, pp. 685-693, 2018.
- [78] K. Raboudi, A. Belkaid and A. B. Saci, "Satisfaction of the solar bounding box constraints," in *Proceeding of 28th International PLEA Conference*, Lima, 2012.
- [79] M. Juyal, K. Kensek and R. Knowles, "SolCAD : 3 D Spatial Design Tool Tool to Generate Solar Envelope," in Proceeding of the 2003 Annual Conference of the Association for Computer Aided Design in Architecture, Indiana, 2003.
- [80] J. Niemasz, J. Sargent and C. F. Reinhart, "Solar Zoning and Energy in Detached Dwellings," *Environment and Planning B: Urban Analytics and CIty Science*, vol. 40, no. 1 January, pp. 801-813, 2013.
- [81] F. De Luca, "Solar form finding," in *ACADIA*, Cambridge, 2017.
- [82] F. De Luca and T. Dogan, "A novel solar envelopes method based on solar ordinances for urban planning," Building Simulation, vol. 12, no. 5, pp. 817-834, 2019.
- [83] I. Darmon, "Voxel computational morphogenesis in urban context : proposition and analysis of rules-based generative algorithms considering solar access," in *Proceeding of Conference on Advanced Building Skins*, Bern, 2018.
- [84] Ž. Kristl and A. Krainer, "Site layout as a function of shading in Karst region," in *International Conference* "Passive and Low Energy Cooling for the Built Environment", Santorini, 2005.
- [85] Ž. Kristl and A. Krainer, "PIRAMIDA, The solar envelope tool," in *TIA Teaching in Architecture Conference*, Krems, 2007.
- [86] M. White, Doctoral dissertation: Informing an Integrated and Sustainable Urbanism through Rapid, Defragmented Analysis and Design, Melbourne: School of Architecture and Design, RMIT University, 2008.
- [87] D. Siret and S. Houpert, "A geometrical framework for solving sunlighting problems within CAD systems," Energy and Buildings, vol. 36, no. 4, pp. 343-351, 2004.
- [88] G. Betti and S. Arrighi, "A differential growth approach to solar envelope generation in complex urban environments," in *PLEA 2017 33rd PLEA International Conference - Design to Thrive*, Edinburgh, 2017.
- [89] K. Anderson, Design energy simulation for architects, New York: Routledge, 2014.
- [90] A. S. Luque and F. De Luca, "Solar Toolbox," food4Rhino, 26 10 2019. [Online]. Available: https://www. food4rhino.com/app/solar-toolbox. [Accessed 10 2 2020].
- [91] M. F. Alkadri, M. Turrin and S. Sariyildiz, "The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts," *Advances in computational design*, vol. 3, no. 4, pp. 319-338, 2018.
- [92] R. J. Koester, "The fundamentals of integrating "the commons": application as community tissue or urban implant," *Renewable energy*, vol. 5, no. 5-8, pp. 1015-1020, 1994.
- [93] M. Maïzia, C. Sèze, S. Berge, J. Teller, S. Reiter and R. Ménard, "Energy requirements of characteristics urban blocks," in *International scientific conference- Cleantech for sustainable buildings*, Lausanne, 2009.
- [94] C. Ratti and E. Morello, "Sunscapes: extending the 'solar envelopes' concept through 'iso-solar surfaces'," in The 22nd conference on passive and low energy architecture, Beirut, 2005.
- [95] C. Ratti and P. Richens, "Raster analysis of urban form," *Environment and Planning B: Planning and Design*, vol. 31, no. -, pp. 297-309, 2004.

Appendix 2A

TABLE A.1 Parameters selection for reference databases				
Topics				
		Conceptual themes	Design workflow	Contextual settings
eration		Solar architecture	Computational design	Urban planning
	Source	Solar envelopes	Solar design	Urban design
do		Solar access	Solar Simulation	Architectural design
OR	WoS	TOPIC: ("solar architecture" OR "solar envelopes" OR "solar access") Refined by: WEB OF SCIENCE CATEGORIES: (CONSTRUCTION BUILDING TECHNOLOGY OR ARCHITECTURE OR GREEN SUSTAINABLE SCIENCE TECHNOLOGY OR ENGINEERING CIVIL OR URBAN STUDIES OR COMPUTER SCIENCE INTERDISCIPLINARY APPLICATIONS OR ENGINEERING MULTIDISCIPLINARY) AND DOCUMENT TYPES: (ARTICLE OR BOOK CHAPTER OR PROCEEDINGS PAPER) AND RESEARCH AREAS: (CONSTRUCTION BUILDING TECHNOLOGY OR ENGINEERING OR ARCHITECTURE OR URBAN STUDIES) Timespan: 1960-2019. Indexes: SCI-EXPANDED, SSCI, A&HCI, CPCI-S, CPCI-SSH, BKCI-S, BKCI-	TOPIC: ("computational design" OR "solar design" OR "solar simulation") Refined by: WEB OF SCIENCE CATEGORIES: (COMPUTER SCIENCE INTERDISCIPLINARY APPLICATIONS OR ARCHITECTURE OR ENGINEERING MULTIDISCIPLINARY OR CONSTRUCTION BUILDING TECHNOLOGY OR ENGINEERING CIVIL OR GREEN SUSTAINABLE SCIENCE TECHNOLOGY) AND DOCUMENT TYPES: (ARTICLE OR BOOK CHAPTER OR PROCEEDINGS PAPER) AND RESEARCH AREAS: (COMPUTER SCIENCE OR ENGINEERING OR ARCHITECTURE OR CONSTRUCTION BUILDING TECHNOLOGY OR URBAN STUDIES) Timespan: 1960-2019. Indexes: SCI-EXPANDED, SSCI, A&HCI, CPCI-S, CPCI-SSH, BKCI-S, BKCI-	TOPIC: ("urban planning" OR "urban design" OR "architectural design") Refined by: WEB OF SCIENCE CATEGORIES: (URBAN STUDIES OR ARCHITECTURE OR REGIONAL URBAN PLANNING OR ENGINEERING CIVIL OR CONSTRUCTION BUILDING TECHNOLOGY OR GREEN SUSTAINABLE SCIENCE TECHNOLOGY) AND DOCUMENT TYPES: (ARTICLE OR PROCEEDINGS PAPER OR BOOK OR BOOK CHAPTER) AND RESEARCH AREAS: (URBAN STUDIES OR ARCHITECTURE OR ENGINEERING OR CONSTRUCTION BUILDING TECHNOLOGY) Timespan: 1960-2019. Indexes: SCI-EXPANDED, SSCI, A&HCI, CPCI-S, CPCI-SSH, BKCI-S, BKCI- SSH, ESCI, CCR-EXPANDED, IC
		SSH, ESCI, CCR-EXPANDED, IC	SSH, ESCI, CCR-EXPANDED, IC	
Total		139	846	10.196

>>>

TABLE A.	ABLE A.1 Parameters selection for reference databases					
	Sources	Topics				
Operation		Conceptual themes	Design workflow	Contextual settings		
		Solar architecture	Computational design	Urban planning		
		Solar envelopes	Solar design	Urban design		
		Solar access	Solar Simulation	Architectural design		
	Scopus	TITLE-ABS-KEY ("solar architecture" OR "solar envelopes" OR "solar access") AND PUBYEAR > 1959 AND PUBYEAR < 2020 AND (LIMIT-TO (DOCTYPE , "ar") OR LIMIT-TO (DOCTYPE , "cp") OR LIMIT-TO (DOCTYPE , "ch") OR LIMIT- TO (DOCTYPE , "bk")) AND (LIMIT-TO (SUBJAREA , "ENGI") OR LIMIT-TO (SUBJAREA , "ENER") OR LIMIT-TO (SUBJAREA , "COMP") OR LIMIT- TO (SUBJAREA , "ARTS")) AND (LIMIT-TO (LANGUAGE , "English"))	TITLE-ABS-KEY ("computational design" OR "solar design" OR "solar simulation") AND PUBYEAR > 1959 AND PUBYEAR < 2020 AND (LIMIT-TO (DOCTYPE , "ar") OR LIMIT-TO (DOCTYPE , "cp") OR LIMIT-TO (DOCTYPE , "ch") OR LIMIT-TO (SUBJAREA , "ENGI") OR LIMIT-TO (SUBJAREA , "COMP") OR LIMIT-TO (SUBJAREA , "ENER")) AND (LIMIT-TO (LANGUAGE , "English"))	TITLE-ABS-KEY ("urban planning" OR "urban design" OR "architectural design") AND PUBYEAR > 1959 AND PUBYEAR < 2020 AND (LIMIT-TO (DOCTYPE, "ar") OR LIMIT-TO (DOCTYPE, "cp") OR LIMIT-TO (DOCTYPE, "ch") OR LIMIT-TO (DOCTYPE, "ch") OR LIMIT-TO (DOCTYPE, "bk")) AND (LIMIT- TO (SUBJAREA, "ENGI") OR LIMIT-TO (SUBJAREA, "COMP") OR LIMIT-TO (SUBJAREA, "ARTS") OR LIMIT-TO (SUBJAREA, "ENER")) AND (LIMIT-TO (LANGUAGE, "English"))		
Total		388	2.548	61.900		
	GS	Sort by date: "solar architecture" OR "solar envelopes" OR "solar access"	Sort by date: "computational design" OR "solar design" OR "solar simulation"	Sort by date: "urban planning" OR "urban design" OR "architectural design"		
Total		43	674	8.530		
AND	WoS	TOPIC: ("solar architecture" OR "solar envelopes" OR "solar access" AND "computational design" OR "solar design" OR "solar simulation" AND "urban planning" OR "urban design" OR "architectural design") Refined by: WEB OF SCIENCE CATEGORIES: (ARCHITECTURE OR URBAN STUDIES OR CONSTRUCTION BUILDING TECHNOLOGY OR ENGINEERING CIVIL OR REGIONAL URBAN PLANNING OR GREEN SUSTAINABLE SCIENCE TECHNOLOGY) AND DOCUMENT TYPES: (ARTICLE OR PROCEEDINGS PAPER OR BOOK CHAPTER OR BOOK) AND RESEARCH AREAS: (ARCHITECTURE OR URBAN STUDIES OR ENGINEERING OR CONSTRUCTION BUILDING TECHNOLOGY) Timespan: 1960-2019. Indexes: SCI-EXPANDED, SSCI, A&HCI, CPCI-S, CPCI-SSH, BKCI-S, BKCI-SSH, ESCI, CCR-EXPANDED, IC				
Total	Total 5.592					
	Scopus	TITLE-ABS-KEY ("solar architecture" OR "solar envelopes" OR "solar access" AND "computational design" OR "solar design" OR "solar simulation" AND "urban planning" OR "urban design" OR "architectural design") AND PUBYEAR > 1959 AND PUBYEAR < 2020				
Total		13				
	GS	Sort by date: "solar architecture" OR "solar envelopes" OR "solar access" AND "computational design" OR "solar design" OR "solar simulation" AND "urban planning" OR "urban design" OR "architectural design"				
Total		1050				



³ Preliminary Simulation Of Solar Envelopes Based On Geometric Properties Of Point Cloud

This chapter has been published as: Alkadri, M. F., Turrin, M. and Sariyildiz, S., 2018. The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts. *Advances in computational design*, 3(4), pp. 319–338. The layout has been adjusted to fit the template of this thesis.

The previous chapter introduced a conceptual review of solar envelopes based on various aspects such as computational methods, parameters, tools, and design applications. It also presented several knowledge gaps and future directions of the study such as 3D contextual models (e.g., limited understanding of site characteristic information), climatic properties (e.g., limited use of the four-season countries and cut-off-times parameters), and geometric configurations (e.g., limited performance criteria). In order to overcome these challenges, this study first focuses on the importance of the surrounding site elements in the existing context. Thus, it proposes the use of point clouds, especially with regard to the geometric properties, not only to capture relevant information for 3D contextual modeling but also to support a comprehensive site analysis before the simulation takes place. As a reality-based dataset, point cloud data enables one to include vegetation and other geometric aspects of the existing context that have not been addressed yet by

current methods of solar envelopes. This chapter specifically investigates this aspect using ALS (Airborne Laser Scanning) datasets gathered from the AHN (Actueel Hoogtebestand Nederland) for selected sites in Rotterdam, the Netherlands. As a preliminary study, the simulation includes a performance evaluation of the design context by comparatively analyzing a series of indicators such as sunlight hours, radiation values, and residual areas between 3D manual model and TIN (Triangulated Irregular Network) model generated from point cloud data. The findings from this chapter describe significant contributions made by geometric properties of point cloud data, not only to filter relevant information from the design context but also to improve the current performance analyses features for solar envelope simulations. Ultimately, these results provide a stepping stone to further explore the potential application of attributes information stored in a point cloud data especially related to surface characteristics and material properties of existing contexts.

The Use And Potential Applications Of Point Clouds In Simulation Of Solar Radiation For Solar Access In Urban Contexts

High-performance buildings should be designed by considering the ABSTRACT interdependencies between the new building and its local context. This is important to avoid any unforeseen failures after the building has been built; particularly ones related to the microclimate impacts that affect human comfort. In this regard, the concept of solar envelopes helps designers to construct a developable mass of the building design, considering solar access and site obstructions. However, the current method of using solar envelopes lacks in integrating detailed information from the existing context during the simulation process due to the limitations of contextual modelling features in preserving the complex architectural geometry and surface characteristics information. To date, emerging point cloud applications offer a great possibility for overcoming these limitations through potential applications of attribute information, such as position (XYZ) and color information (RGB). This study presents a comparative analysis between manually-built 3D geometric models and digital geometric models generated from point cloud data. This geometric modelling comparison focuses on the relevant factors of solar radiation and a set of environmental performance simulations based on selected parts of the generated models. The experimental results emphasize an introduction of the design approach and the visibility of the dataset from existing 3D environments. This paper ultimately aims to improve the support of current architectural design decisions by increasing the correspondence between digital models for performance analysis and the real environment as a design context during the conceptual design phase.

KEYWORDS solar envelope; point cloud data; microclimate impacts; environmental simulation; modeling environments

3.1 Introduction

3.1.1 General background

According to [1] in the Alterra report, the percentage of people living in urban areas will continually increase by more than 60 % of the total world population in 2030. This trend indicates that the impact of increased urban density will result in a 40% increase of energy consumption in the building sectors [2]. Having said that, the implementation of a passive design strategy is an important solution, not only to improve the quality of the built environment but also to achieve optimal building performance during the conceptual design process. The to-be-built environmental analysis, furthermore, avoids unexpected failures once the building has been built.

The concept of vernacular architecture was commonly designed and constructed by considering the interrelation between the new building and the local context. In the case of contemporary architecture, unfortunately, there remain some concerns in the design implementation. For example, the Walkie Schorcie building in the UK presents some practical issues with the lack of integration into the local context. Evidently, its building form causes numerous microclimatic effects. The glazed and curved south facade reflects solar radiation of more than 70 Celsius degree to the surrounding environment. This situation has had a tremendously negative impact on the local context, where the heat produced from the reflected solar radiation melted the body of a car parked on the street below and the expensive adaptation of the building facade [3]. It goes without saying that the building was awarded as "the worst building of the year," by RIBA and received the Carbuncle Cup in 2015. This case clearly points out the unexpected microclimatic impacts caused by a new building on the local environment. In another instance, the UCL Student Housing located on Caledonian Road, North London, exemplifies a situation in which a new building does not fit into the local context. In this case, around half of the 350 student rooms do not gain direct sun access because part of the building is being obstructed by the original 19th century façade.

By focusing on solar radiation performance, the examples described above highlight critical perspectives on how architecture should approach the local context, especially focusing on the early phase of the design process. The new building should develop a reciprocal relationship with the existing environment since the natural resources like sunlight, daylight, and wind are shared. This paper specifically proposes the integration of the existing method of solar envelopes by making use of detailed information extracted from point cloud data. Further overview of solar envelopes and point clouds is addressed below.

3.1.2 Solar envelopes

A solar envelope is an imaginary building mass determined by considering the desired amount of sun access without obstructing the surrounding site during critical times [4]. Figure 3.1 illustrates the basic mechanism for the solar envelope to produce daily and annual time limits, considering sunlight patterns in the Northern hemisphere. In a full-day setting, the morning sun establishes the western boundary envelope while the afternoon sun generates the eastern boundary envelope [5]. During the critical time between 9 am and 2 pm, the envelope limits will be steeper in the east than in the west because the sun's altitude at 2 pm is higher than at 9 am. In this case, the sun's rays hit the plot at a higher position between the selected hours. The same process applies to annual time patterns. The solar envelope is steeper in summer than in winter because the sun angle in winter is lower than in summer.



FIG. 3.1 Design mechanism of solar envelopes, generated from: A. daily time limits and B. annual time limits [6]

Following the basic set of solar envelope rules, its concept was then evolved and elaborated into several identified generation techniques:

- Solar obstruction angle; consists of the three main parameters such as solar planes, perimeter of the site plan, and site orientation [7] [8] [9] [10] [11] [12] [13] [14].
- Descriptive method; is based on trigonometric calculations between sun altitude, sun azimuth and cut-off times [15] [16] [17] [18] [19].
- Digital elevation modeling (DEM); relies on image processing techniques and shadow volume calculations for urban scale [20].
- Constructive solid geometry; is mainly based on site extrusion and solar vectors [21]
 [22] [23] [24] [25].

These existing studies propose different ways of dealing with solar envelopes aligned with the technological development of design tools. However, their solar envelope simulations were predominantly performed without considering the actual environmental context. Consequently, the spatial relationship between the proposed building and the local context poses to some extent contextual issues associated with the microclimate during the design process. Some literature has discussed the role of the surrounding context when generating solar envelopes based on the descriptive method [26] [27] [28] [29] [30] [31] [32] [33]. However, the geometric model of existing environment (design context) in these works is generated on the basis of major simplifications. Currently, the geometric 3D representation of existing contexts is mostly constructed using the basic architectural shapes, especially when it comes to isolated sites with complex building forms. This can significantly affect the interpretation of the simulation results. For example, surface characteristics of existing buildings such as texture and material play an important role when simulations apply the same treatment to all building surfaces, which in fact differ from the actual situation. In addition, the surrounding vegetation [12] should also be further addressed as a relevant parameter for contextual setting because it can influence the design configuration of the solar envelopes.

In contrast to existing research, this study points to the importance of considering the aforementioned aspects to generate a solar envelope, particularly related to the properties of the surrounding environment. The potential of point clouds as 3D data scanning allows mapping of spatial information from the existing context. The inclusion of a large accurate dataset and attribute information contained in the point cloud can be used to generate contextual models of real environments. The further development of accurate modelling not only makes it possible to perform simulations and design analysis afterward but also to minimize any possible discrepancies between the simulated environmental design and the real context.

3.1.3 **Point cloud data**

The rapid development of 3D laser data scanning technology has been widely applied in various disciplines, especially in data visualization and data analysis such as civil engineering, geosciences, photogrammetry, and heritage. Conceptually speaking, Weinmann [34] pointed out that the term "point" in geometry represents the specific unit location in a particular space and the term "cloud" describes the unorganized nature of its spatial arrangement and coherence within blurred boundaries. Thus, the term point cloud is a set of points, Pi = 1, . . .n, which is embedded in the threedimensional Cartesian space [35]. As a 3D data scanning entity, point clouds are discrete three-dimensional locations (points) that can have additional information associated with each record [36]. It is characterized by spatial XYZ coordinates as position information and is optionally assigned by auxiliary information (see Figure 3.2) such as color information (RGB) [37] and intensity information (I) [38] [39] [34].



FIG. 3.2 Attribute information of point cloud data [40] quality.

Furthermore, Randall [40] described that point cloud data can support the following information:

- The position of the object with 3D coordinates in global coordinate system.
- Geometric features such as length between points and orientation of objects.
- Visual information such as 2D and 3D digital images and 3D virtual models.
- Physical features such as textures, cracks and different types of objects with different densities [41].

Utilization of point clouds in several engineering domains shows high-precision results for simulation performance [42]. It can be used to perform numerous tasks in the early stage of architectural design where various design scenarios can be analyzed to support the decision-making process [43]. For example, point cloud data can be employed to generate 3D models of existing complex urban forms [44] [45] [46]. Besides, it also allows one to visualize the entire construction process according to time-based transformations [47] and to capture the deformation of existing structures in the renovation process [48]. Furthermore, the surface attribute information contained in the point cloud is useful for detecting the solar potential on any existing building surfaces [49] [50] [51]. However, studies that specifically investigate the solar envelope performance based on point cloud data remain a blank spot in solar radiation analysis.

This study ultimately aims to explore the potential application of point cloud data in generating the solar envelope. In this case, a comparative analysis between manually-built 3D geometric models and digital geometric models generated from point cloud data is performed; the following section illustrates this process further.

3.2 Proposed procedure: From point clouds to the solar envelopes and design analysis

Given the unstructured raw dataset of the point cloud, the proposed workflow allows simulating of the solar envelope according to selected information from the existing context. Its procedure consists of three phases: (a) data collection, (b) generation of the solar envelope and (c) design analysis. The overview of the process is illustrated in Figure 3.3.



FUTURE RECOMMENDATION

FIG. 3.3 Overview of the proposed procedure

3.2.1 Data collection

The data collection phase is useful not only for gathering the raw point cloud dataset but also for identifying relevant information from the local context as input in the design simulation. During this phase, the source of the dataset plays an important part because it considers the type of attribute information attached to the point cloud data. For example, the Terrestrial Laser Scanning (TLS) technique caters to different levels of robustness and attributes of point cloud dataset compared to Airborne Laser Scanning (ALS). Specifically, the following are steps needed to distribute the selected point cloud dataset:

- Select the plot and the relevant site properties such as vegetation and surrounding buildings.
- Remove outliers to reduce unnecessary information during the meshing process.
- Export the dataset into the designated format.

The output of this phase is a selected point cloud that contains color (RGB) and position (XYZ) information. This attribute information enables one to identify the elevation of each point in the dataset according to its color scale.

3.2.2 Generation of the solar envelopes

This phase focuses on developing a new computational workflow of the solar envelope. In the proposed workflow, point cloud data processing aims not only to classify site properties such as vegetation and buildings but also to perform the mesh generation for the selected building's facade. The mesh model is used to support the environmental performance simulation. As an extension of the solar envelope method from Capeluto and Shaviv [26], the proposed workflow consists of two design principles "solar right envelope" and "solar collection envelope." Generally, this workflow endeavors to tackle two main challenges. The first is to select the relevant parameters of the solar envelope according to the extracted information contained in the point cloud data. The second is to handle the point density as it significantly affects the computation time during the simulation.

3.2.3 **Design analysis**

The third phase is a design analysis that can be performed using the solar envelope model generated from point cloud data. As this phase includes exploratory design studies, a wide variety of possible design scenarios can be defined depending on the intended design case.

Furthermore, a comparative analysis is addressed using two different modelling environments, namely the Triangulation Irregular Network (TIN) model generated from the point cloud and manually-built 3D geometric models based on CAD drawings. This aims not only to investigate further the design configuration produced by the similar urban context but also to identify the discrepancies of simulation results based on three indicators, namely the volumetric shape of solar envelopes, the total radiation value, and the total sunlight hours.

3.3 Case study: The Kruisplein area in Rotterdam, The Netherlands

The selected location is Kruisplein in Rotterdam, The Netherlands (latitude 51.9244° N, and longitude coordinates 4.4777° E). As this area is a densely populated district near Rotterdam Central Station, many people pass by frequently to eat, shop, and rest. This area also varies in functions and scales such as high-rise and wide-span buildings, large open spaces, and vegetation between the buildings. By considering the urban quality of this area, the existence of new buildings should preserve the reciprocal relationship with the surrounding environment. The case study focuses on guaranteeing appropriate solar access between the new design and the surrounding environment.



FIG. 3.4 Geometric representation of the selected site in different 3D models

Figure 3.4 illustrates the selected buildings from the case study consisting of a hotel (high-rise building) and a commercial area (wide-span building). Understanding these specific functions enable us to set limits on sun access for the surrounding environment. For example, commercial functions consist of shops, restaurants, and cafes along the ground floor, and offices on the upper floors. In this regard, commercial areas on the ground floor preferably sit within a shaded area rather than direct sunlight. This is because the durability issue of the food will not last long when it gets direct sunlight at certain times, especially those placed at the storefronts. Meanwhile, open spaces are often designed as a mixed-use function for temporal events such as live music, market, and other festivals.

The plot selected in this case study is set as a multi-function building to support activities in the open space. It consists of 16.5×42 m with commercial areas on the ground floor while offices are on the upper floor. The proposed building is expected to adapt to the environmental performances of the surrounding site without violating sun access to certain building functions.

3.4 Data collection

In terms of dataset availability, this study uses an open data platform for gathering the 3D point cloud. Architects can access and extract the selected context from the *Actueel Hoogtebestand Nederland*, AHN (retrieved from http://www. ahn.nl). This open dataset is a digital altitude map of point cloud according to laser altimetry measurements. During the data collection phase, three steps are performed as illustrated in Table 3.1, namely site selection, data conversion, and computing environment.



3.4.1 Data conversion

Data conversion aims to prepare appropriate datasets to be ready for use in the simulation. To do so, Cloud Compare (CC) is mainly used as supporting tools. The following steps highlight tasks for this phase:

- Activate color fields of the dataset. As the original dataset comes from the AHN map, the initial color remains monochrome. Thus, color fields are essential in order to filter selected areas based on scalar field values. The scalar field value refers to the real number for each coordinate in space such as 3D space $\Phi = \Phi(x, y, z)$ [52], or a structure depicting real (scalar) values [53]. In CC, several scalar field functions are available such as density, curvature, roughness, intensity, and so forth. As illustrated in Table 1 step 2, for example, the scalar field of coordinate Z is activated from the original dataset (monochrome). Embedded with the scalar field, the gradient color in the Z coordinate represents the elevation of the area. In this regard, the red color indicates the highest position while the blue color shows the lowest position. From this point on, the color scale value can be used to filter and identify specific areas according to its elevation although the filtering features in CC are limited to per area, not per object.
- Export the selected dataset. In order to be legible and interoperable in parametric design environment, the dataset needs to meet the appropriate format. In this regard, we export the original dataset (.laz file) into the .e57 file as it keeps the color field of datasets visible in the 3D modelling environment.

3.4.2 **Computing environment**

This study considers parametric designs to enable architects to integrate point cloud data into environmental design platforms. Due to unclassified points from imported datasets in CC, site properties such as buildings, trees, and other objects contain the same point entity. Therefore, the following actions are required to run before performing the simulation:

- Classification of existing objects. This task is important to only select relevant existing objects for the simulation. This study, for example, only selects two relevant building facades to be included in the sunlight hour simulations.
- Generation of reference site plans. It is used to set the boundary of selected building areas, especially for the sun right access threshold.
- Generation of the TIN model from the point cloud data.
- To carry out the above mentioned tasks, this study specifically uses Rhinoceros as a 3D modelling platform in parallel with Grasshopper (GH) for a visual programming environment. Besides, several environmental simulation components are used such as Ladybug [54] for constructing the solar right and the solar collection envelope, Volvox [55] for the point cloud processing, and some custom components [56] for generating the TIN model.

3.5 Development of the design framework for solar envelopes

This paper formulates new computational workflow for establishing solar envelopes by considering two design principles, namely the solar right envelope and the solar collection envelope. The integration of these envelopes generates a maximum developable volume of building design according to solar access criteria.

The solar right envelope stands for a maximum height of the envelope surface based on predefined criteria. To achieve the solar right envelope, the sun access level needs to be regulated considering the adjacent buildings during a desirable period. The main rule is to maintain sun access to surrounding buildings without obstruction from the mass of new building design. To do so, the height level of sun right access from the surrounding buildings should be marked first. This height level refers to the bottom border (minimum level of gaining sun access) of surrounding buildings. For example, by setting the height level on the 1st floor, we compensate the sun right access of neighboring buildings above the 1st floor. This creates a shaded area for surrounding buildings located below a predetermined level.

Meanwhile, the solar collection envelope refers to the lowest possible envelopes in obtaining the sun access (for the proposed design) without being violated by the adjacent buildings during the desirable period. The limit of the solar collection height is determined considering the upper border of surrounding buildings. As a result, areas in the proposed design that are located below the surface of solar collection envelope will not get direct sun access. The mechanism of solar collector in the proposed design. In this case, the solar collector refers to the solar panel instruments to absorb sunlight as a source of building energy.

Furthermore, this study identifies many variations of solar envelope parameters as described in the previous literature. The inclusion of point cloud data allows for the formulation of additional parameters in the existing workflow of solar envelopes. This study particularly considers two main components for the generation of solar envelopes. First, the climate aspect consists of the weather data, longitude and latitude coordinates, azimuth and altitude angle. Climatic properties also include several important information such as the sun path (useful for running the analysis period and radiation analysis by inputting specific location contained in the weather file), sky components (generating a matrix of sky patches based on the Perez's weather model, sky density, and connecting the analysis period for the radiation analysis), and insolation periods. In this case, we set a scenario for cut-off times by testing a half year of the insolation period from June 1st to December 31st in 2017 between 9 am to 10 pm. Second, site rules and environmental aspects. These aspects consider the plot area, set back, height restriction, vegetation and the TIN model generated from the point cloud data. Last, the simulation results consist of a comparative analysis between performance indicators using different 3D modelling platforms. An overview of the selected parameters is illustrated in Figure 3.5



FIG. 3.5 Selected parameters for solar envelope



FIG. 3.6 Computational workflow of solar envelope frequency

According to computational workflow of the solar envelope (see Figure 3.6), the use of 3D point cloud data shows the following impacts on the geometric envelope:

- The role of vegetation in the proposed workflow affects the shape and volume of the solar envelope. The proposed scenario sets the sun access protection to the intended design without being blocked by surrounding vegetation. In this case, vegetation acts as part of the site obstruction. Thus, we set the vegetation elevation level within the limits of the solar collection parameter.
- The ability of 3D point cloud to capture complex building geometry can fill gaps in manually built 3D model regarding building surface properties. For example, point cloud data can be used to construct the 3D model of under construction buildings around the plot or areas located in congested environments such as alleys or isolated places. Therefore, modifications that occur to building facades due to renovations can always be modelled as is and included in the simulation of solar radiation analysis.
- The total radiation value and sunlight hours analysis may create significant simulation results when considering only relevant facades rather than entire building surfaces. Accordingly, the simulation in this study only selects facades facing the plot.
- The inclusion of sunlight hour analysis allows the detection of the most accessible areas that receive solar radiation on the building facades and plot areas during a certain time. As a result, we can identify the potential area of solar collectors in the geometric solar envelopes. The simulation results exemplify the different values of sunlight hours produced by two different 3D models.

3.6 Results and discussion

In order to construct an in-depth analysis of the simulation results, this study investigates two comparative aspects that intersect each other as follows (see Figure 3.7):

 The simulation context is divided into two parts, namely the proposed design and the existing context. In the proposed design, two indicators are calculated: the envelope's volume and the radiation value. On the other hand, we consider three indicators for the existing context: the total sunlight hours, the potential areas that receive more than 8,5 hours of direct sunlight and radiation values. The simulation mainly covers the areas included in the selected wide-span facades, the high-rise building and the plot.

 In the simulation process, the 3D site model compares two types of modelling techniques: manually-built 3D geometric models and digital geometric models generated from point cloud data that refers to the TIN model.



FIG. 3.7 Comparative indicators of the simulation context

3.6.1 Simulation of the existing contexts

As illustrated in Figure 3.7, the simulation assessment based on the existing context consists of three indicators: sunlight hours, radiation values, and residue areas. These indicators are evaluated by comparing two different 3D models, which are manually-built 3D geometric models and the TIN model. In principle, the geometric surface of both 3D models similarly uses the polygon mesh as it contains vertices, edges, and faces [57] and thus, it is easily transformed to any objects. However, the mesh generation techniques are constructed differently for the two models. The manually-built 3D geometric model is generated based on the quadrilateral type of

cell shapes. However, this process is applied on the 2D plan extrusion so it makes the surface morphology of the buildings is mostly flat and plain. On the other hand, the TIN model employs the triangle mesh based on the Delaunay triangulation method. Due to a proximal technique, the Delaunay method often results in depth and irregular structures in the surface morphology of the TIN model. This is because the three nodes of a triangle need to be maintained within the boundaries of the imaginary circle [58]. Nevertheless, the automatic generation process of the TIN model takes less time compared to the manually-built 3D geometric model.

TABLE 3.2 Simulation results based on the existing context			
The manual model	Total sunlight hours (n hours)	Areas > 8.5 hours of direct sunlight (m ²)	Total radiation (kWh)
Façade building A (wide span)	27.476,68	161,888	248.328
Façade building B (high rise)	129.187,77	12,627	207.756
Site area	29.324,46	5,879	724.274,11
The TIN model	Total sunlight hours (n hours)	Areas > 8.5 hours of direct sunlight (m^2)	Total radiation (kWh)
Façade building A (wide span)	18.236,16	0,00785	51.822,194
Façade building B (high rise)	47.040,23	4,45021	114.613,314
Site area	17.387,45	0,9853	45.552,12

Table 3.2 illustrates a comparison of three indicators using two different 3D models. The calculation of total sunlight hours refers to the total hours of direct sunlight received at each test point in which being multiplied by the mesh areas of the test point attached to [59]. Thus, the total hours received by each simulation model is principally the accumulation of direct sunlight retrieved by each mesh face corresponding to each test point. For example, the total test points included in Façade building A are around 124 test points. Then, when the total sunlight hours of Façade building A are divided by 124, the sunlight hours produced for each test point will be approximately 221,58. Furthermore, the simulation process considers only the relevant building surface (selected facades) and do not simulate the entire buildings to avoid unnecessary context information. We also calculate the potential areas that receive more than 8.5 hours of direct sunlight. A scenario has been set to employ 8.5 desirable hours from the total of 13 hours of the simulation period. The total areas for the simulations are then divided into the areas received the sunlight hours for fewer than 8.5 hours. Based on this calculation, we can identify the location of the PV panels on the building surfaces and locate the site plan on the plot areas.

According to the simulation results in Table 3.2, the manual 3D model, on average illustrates larger values for all performance indicators in comparison to the TIN model. In the case of total sunlight hours, although the number of test points for the TIN model is greater than that of the manual 3D model, its test point value is quite small. This is caused by two factors. First, the mesh surface of the resulting TIN model is unequal and unstructured from one side of the triangle to another. Afterward, the shape of these triangle faces is constructed by the interpolation factor between control points (referring to the Delaunay Triangulation method). The robustness of triangle meshes and test points shows a trade-off with the point density from the original dataset as they principally affects the grid distribution on the surface of the final mesh. In contrast, by enlarging the mesh surface of both models (see Figure 3.8), it is clear that the manual 3D model illustrates a flat and continuous surface and thus, it affects the equal grid distribution for the test points.



FIG. 3.8 Comparison of the solar point vectors

Second, the direction of the triangle mesh surface in the TIN model demonstrates an irregular position. This means that the test point vector follows the orientation of the triangle mesh due to its normal perpendicular position. Consequently, these point vectors respond differently to incoming sunlight and yield various values in the simulation results. With respect to surface position, the idea of reflection behavior also significantly impacts on the surface properties of both models. The flat surface of the manual 3D model demonstrates an equal angle between reflection and illumination. According to the law of reflection, the incoming lights is directed to the specular path of the reflection process [60] which refers to the regular reflection. This mechanism indicates that the reflected light is either parallel or converged at a point and the accumulated radiation values converge in the same direction. Meanwhile, the scattered light reflection from the TIN model illustrates "the orangepeel effect" because of the reflection pattern from the wavy surface. As this condition is highly reliant on the surface morphological structure, the distribution of solar reflection is spread unequally which then affects the amount of radiation value accumulated during the simulation.

3.6.2 Simulation of the proposed design

The second simulation discusses the results of the proposed design. In this regard, three different scenarios are applied in the simulation of solar envelopes. Each scenario consists of sun access criteria that corresponds to solar rights and solar collection envelopes. The criteria for sun access refer to the upper and lower limits of the context property. As described in Section 3.5, the upper border indicates the height limit of surrounding environment properties corresponding to the solar collection surface. The lower border marks the minimum level of sun right access of the adjacent building which corresponds to the generation of the solar right envelope. The objective of these scenarios is to investigate the geometric configuration of solar envelopes following the sun access settings of the surrounding environment. Furthermore, geometric solar envelopes are then evaluated using two indicators, namely the volumetric envelope and the radiation value (see Table 3.3). The geometric assessment of solar envelopes aims to evaluate the performance of solar envelopes by considering the use of different 3D models of the surrounding environment.

TABLE 3.3 Simulation results of the proposed design

The manual model	Scenario A (m) Top: 50/20 Bottom: 3/2	Scenario B (m) Top: 60/15 Bottom: 7/5	Scenario C (m) Top: 80/10 Bottom: 1/1
The envelope's volume (m ³)	13485,30	7691,60	7508,46
Radiation value (kWh)	98790,36	75291,73	74324,64
Solar envelopes	A service of service o	A second se	A Construction of the second s

*top = high rise building / wide span building (meter) bottom = high rise building / wide span building (meter)

The TIN model	Scenario A (m) Top: 50/20/14 Bottom: 3/2	Scenario B (m) Top: 60/15/4 Bottom: 7/5	Scenario C (m) Top: 80/10/14 Bottom: 1/1
The envelope's volume (m ³)	11996,63	9920,10	7252,89
Radiation value (kWh)	86736,44	79270,82	71444,75
Solar envelopes	A service of the serv	A Construction of the second sec	And and a second s

*top = high rise building / wide span building/vegetation (meter) bottom = high rise building / wide span building (meter)

According to the simulation results shown in Table 3.3, the following items are discussed:The geometric envelope for all the scenarios illustrate a nearly flat surface that inclines on the top envelope. The spatial context arrangement (see Figure 3.4b) shows that only wide-span building are located adjacent to the edge of the plot area. Due to the long and continuous shape of the wide-span, it covers the dimensions of the plot edges. Thus, the edge of the plot facing the wide-span illustrates a flat curve. However, on the other side, there is a high-rise building, but unfortunately, it is almost out of reach from the edge of the plot. This makes the solar envelopes geometry does not fully consider the site obstruction from the high-rise. The higher curve (facing the high-rise) is ultimately affected by the height limit of the upper border parameter. Furthermore, the inclusion of trees as site obstruction in the TIN model shows non-flat surfaces of the top envelope.

- Scenario A demonstrates the largest volume of the solar envelope in comparison with other scenarios in both 3D geometric models. This result indicates that the maximum volume of the solar envelope can be generated by setting a small difference value between the top and the bottom border. The same rule applies to the distance between the high-rise and the wide-span building. The smaller the difference value assigned to the top borders, the greater the volume of the solar envelope can be produced (scenario A for both models).
- Vegetation has successfully influenced the geometric and volumetric size of solar envelopes for all scenarios. Scenario A and C demonstrate the decreasing volume of solar envelopes and radiation values from the manual 3D model to the TIN model. However, different trends are illustrated in Scenario B, with the volume size of solar envelopes and the radiation value increasingly slightly in the TIN model. This is because we set a small difference between the upper and lower border for the vegetation parameter.
- Scenario C illustrates the lowest value for the volumetric size of solar envelopes as compared to others. This result simultaneously indicates the minimum buildable volumes of solar envelopes that can be considered for architects in design exploration.

3.7 Concluding remarks

The simulation results summarized in Table 3.2 and Table 3.3 illustrate that the use of point clouds not only contributes to the generation of solar envelopes but also offers further possibilities in analyzing the existing environment. The proposed method confirms a positive procedure in achieving the main objectives of this study. However, different workflows may appear with the use of data processing tools and levels of detail information.

Furthermore, the comparative analysis between the manual 3D model and the TIN model demonstrates significant discrepancies with respect to the simulated radiation values and sunlight hours. The use of customized components in Grasshopper also provides great assistance in the development of computational workflows, particularly related to the point cloud processing and 3D mesh modelling.

Although some progress has been made during this study, the incremental approach provides only a partial answer due to acknowledged limitations. For example, the surface accuracy of the TIN model needs to deal with the point density from the original dataset and the unstructured polygon surfaces during the mesh generation process. This study also does not consider further material properties during the simulation process because of the dataset availability from the AHN map. Consequently, the calculation of the radiation value focuses on the surface morphology of the building.

Further study related the aforementioned constraints needs to continue. The concept of "metabolism" by Knowles is of great importance in this study as it can have a broad impact on the quality of the built environment. The following are some potential development areas for future research:

- Inclusion of multiple performance indicators into the simulation framework such as shadow analysis, daylight factors, thermal comfort, and so forth. These indicators allow further environmental assessments of the solar envelope that can be useful during the conceptual design stage.
- Integration of current solar envelopes workflow into multi-objective optimizations.
 This aims to explore geometric solutions of the solar envelope according to specific objectives such as maximizing the energy distribution of the solar envelope and the efficient use of space within the solar envelope.
- Adjustment of different climates into the workflow of solar envelopes. For example, places like South East Asian countries will require a different objective of solar envelopes in comparison with European countries. In this regard, shaded areas are more suitable for tropical countries than European countries.

References

- [1] L. van Hove, G. Steeneveld, C. Jacobs, B. Heusinkveld, J. Elbers, E. Moors and A. Holtslag, "Alterra report: Exploring the urban heat island intensity of Dutch cities - Assessment based on literature review, recent meteorological observations and datasets provided by hobby meteorologist," Wageningen University, Wageningen, 2011.
- V. Brophy and J. O. Lewis, A Green Vitruvius: Principles and practice of sustainable architectural design, London: Earthscan, 2011.
- [3] J. Futcher and G. Mills, "Carbuncle cup," 2 September 2015. [Online]. Available: https://www.researchgate. net/publication/281439064_Have_We_Learnt_Anything_from_the_Walkie_Scorchie_httpowlyRGBmO. [Accessed 2 July 2017].
- [4] R. L. Knowles, Energy and form: An ecological approach to urban growth, Cambridge: The MIT Press, 1974.
- [5] R. L. Knowles and R. D. Berry, Solar envelope concepts; Moderate density Building applications, Springfield: National technical information service U.S. Department of Commerce, 1980.
- [6] R. L. Knowles, Sun, rhytm and form, Massachusetts: The MIT Press, 1981.
- [7] R. S. Brandao and M. P. Alucci, "Solar access in tropical cities: towards a multicriteria solar envelope," in The 22nd Conference on passive and low energy architecture, Beirut, 2005.
- [8] F. O. R. Pereira, C. A. N. Silva and B. Turkienikz, "A methodology for sunlight urban planning: a computerbased solar and sky vault obstruction analysis," *Solar energy*, vol. 70, no. 3, pp. 217-226, 2001.
- [9] P. C. Grazziotin, F. O. R. Pereira, C. M. D. S. Freitas and B. Turkienicz, "Integration of Sunlight Access Control to Building Potential Simulator," in *The Ibero-American Symposium on Computer Graphics*, Guimaraes, 2002.
- [10] M. D. G. V. D. Amaral, "The application of solar architecture in the planning of the campus," in The 2005 World sustainable building conference, Tokyo, 2005.
- [11] B. Paramita and M. Koerniawan, "Solar envelope assessment in tropical region building case study: vertical settlement in Bandung, Indonesia," in *The 3rd International conference on sustainable future for human security SUSTAIN 2012*, Kyoto, 2013.
- [12] R. Emmanuel, "A hypothetical 'shadow umbrella' for thermal comfort enhancement in the equatorial urban outdoors," *Architectural science review*, vol. 36, no. 4, pp. 173-184, 1993.
- [13] M. Dekay, "Climatic urban design: configuring the urban fabric to support daylighting, passive cooling, and solar heating," *The sustainable city VII*, vol. 155, pp. 619-630, 2012.
- [14] A. Okeil, "A holistic approach to energy efficient building forms," *Energy and buildings*, vol. 42, no. 9, pp. 1437-1444, 2010.
- [15] I. Capeluto, A. Yezioro, T. Bleiberg and E. Shaviv, "From computer models to simple design tools: solar rights in the design of urban streets," in *Ninth international IBPSA conference*, Montreal, 2005.
- [16] G. Bruce, "High density, Low energy: achieving useful solar access for Dublin's multi-storey apartment developments," in *PLEA*, Dublin, 2008.
- [17] M. M. Saleh and K. S. Al-hagla, "Parametric urban comfort envelope: an approach toward a responsive sustainable urban morphology," *International journal of civil, environmental, structural, construction and architectural engineering*, vol. 6, no. 11, pp. 930-937, 2012.
- [18] D. Noble and K. Kensek, "Computer generated solar envelopes in architecture," *The Journal of architecture*, vol. 3, no. 2, pp. 117-127, 1998.
- [19] P. Camporeale, "Genetic algorithms applied to urban growth optimization," in *Proceeding of eCAADe Computation and Performance 2013*, Delft, 2013.
- [20] C. Ratti and E. Morello, "Sunscapes: extending the 'solar envelopes' concept through 'iso-solar surfaces'," in The 22nd conference on passive and low energy architecture, Beirut, 2005.
- T. N. Stasinopoulos, "Solar envelope a construction method using AutoCAD 2000," 9 July 2001. [Online].
 Available: http://www.oikotekton.eu/solenvelope. [Accessed 25 October 2016].
- [22] K. Kensek and A. Henkhaus, "Solar access zoning + building information modeling," in *Solar 2013*, Baltimore, 2013.
- [23] A. Vartholomaios, "The residential solar block envelope: a method for enabling the development of compact urban blocks with high passive solar potential," *Energy and Buildings*, vol. 99, pp. 303-312, 2015.
- [24] J. F. Cotton, "Solid modeling as a tool for constructing solar envelopes," Automation and Construction, vol. 5, no. 3, pp. 185-192, 1996.

- [25] J. Niemasz, J. Sargent and C. F. Reinhart, "Solar zoning and energy in detached dwellings," in SimAUD, Boston, 2011.
- [26] G. Capeluto and E. Shaviv, "Modeling the design of urban fabric with soalr rights considerations," Taejon, 1997.
- [27] K. Machacova, J. Keppl and L. Krajcovics, "The solar envelope method in education at the faculty of architecture STU Bratislava," in *Central Europe towards sustainable building*, Prague, 2013.
- [28] C. L. Martin and G. Keeffe, "The Biomimetic solar city: solar derived urban form using a forest-growth inspired methodology," in *The 24th Conference on passive and low energy architecture*, Singapore, 2007.
- [29] C. L. Martin, M. Pilling, C. Stott and V. Walsh, "The nectar project: solar development of post-industrial urban communities," in *The 27th conference on passive and low energy architecture*, Louvain-la-Neuve, 2011.
- [30] K. Raboudi and A. B. Saci, "A morphological generator of urban rules of solar control," in *The 29th conference* Sustainable architecture for a renewable future, Munich, 2013.
- [31] M. Dekay, "The implications of community gardening for land use and density," *Journal of architectural and planning research*, vol. 14, no. 2, pp. 126-149, 1997.
- [32] F. De Luca, "Solar envelope optimization method for complex urban environments," in *CAADence in Architecture*, Budapest, 2016.
- [33] F. De Luca and H. Voll, "Computational method for varibale objectives and context aware solar envelopes generation," in SimAUD, Toronto, 2017.
- [34] M. Weinmann, Reconstruction and Analysis of 3D Scenes, Karlsruhe: Springer International Publishing, 2016.
- [35] J. Otepka, S. Ghuffar, C. Waldhauser, R. Hochreiter and N. Pfeifer, "Georeferenced Point Clouds: A Survey of Features and Point Cloud Management," *ISPRS International Journal Geo-Information*, vol. 2, no. 4, pp. 2038-1065, 2013.
- [36] D. A. White, "LIDAR, Point Clouds, and Their Archaeological Applications," in *Mapping Archaeological Landscapes from Space*, New York, Springer-Verlag New York, 2013, pp. 175-186.
- [37] Y. Fujita, Y. Hoshino, S. Ogata and I. Kobayashi, "Attribute Assignment to Point Cloud Data and Its Usage," Global Journal of Computer Science and Technology, vol. 15, no. 2, pp. 2-B, 2015.
- [38] M. Wand, A. Berner, M. Bokeloh, P. Jenke, A. Fleck, M. Hoffman, B. Maier, D. Staneker, A. Schilling and H.-P. Seidel, "Processing and interactive editing of huge point clouds from 3D scanners," *Computer&Graphics*, vol. 32, no. 2, pp. 204-220, 2008.
- [39] D. D. Lichti, "Spectral filtering and classification of terrestrial laser scanner point clouds," The Photogrammetric Record, vol. 20, no. 111, pp. 218-240, 2005.
- [40] T. Randall, "Client Guide to 3D Scanning and Data Capture," BIM Task Group, UK, 2013.
- [41] A. Gressin, C. Mallet, J. Demantké and N. David, "Towards 3D lidar point cloud registration improvement using optimal neighborhood knowledge," *ISPRS Journal of Photogrammetry and Remote Sensing*, vol. 79, no. -, pp. 240-251, 2013.
- [42] V. Salehi and S. Wang, "Using point cloud technology for process simulation in the context of digital factory based on a systems engineering integrated approach," in *Proceedings of the 21st International Conference on Engineering Design (ICED 17)*, Vancouver, 2017.
- [43] Y. Fujita, I. Kobayashi, W. Chanseawrassamee and F. Hoshino, "Application of Attributed Road Surface Point Cloud Data in Road Maintenance," *Journal of Japan Society of Civil Engineers, Ser. F3 (Civil Engineering Informatics)*, vol. 70, no. 2, pp. I_185-I_192, 2014.
- [44] P. Babahajiani, L. Fan, J. Kamarainen and M. Gabbouj, "Automated super-voxel based features classification of urban environments by integrating 3D point cloud and image content," in *IEEE International Conference on Acoustics, Speech and Signal Processing*, Kuala Lumpur, 2010.
- [45] R. Richter and J. Döllner, "Concepts and techniques for integration, analysis and visualization of massive 3D point clouds," *Computers, Environment and Urban Systems*, vol. 45, pp. 114-124, 2014.
- [46] T. Hackel, J. D. Wegner and K. Schindler, "Fast semantic segmentation of 3D point clouds with strongly varying density," in XXIII ISPRS Congress, Prague, 2016.
- [47] N.-J. Shih and M.-C. Wu, "A 3D point-cloud-based verification of as-built construction progress," in *The CAAD Futures 2005 Conference*, Vienna, 2005.
- [48] T. Várady, R. R. Martin and J. Cox, "Reverse engineering of geometric models—an introduction," *Computer-Aided Design*, vol. 29, no. 4, pp. 255-268, 1997.
- [49] A. Jochem, B. Höfle, M. Hollaus and M. Rutzinger, "Object detection in airborne LIDAR data for improved solar radiation modeling in urban areas," in *Proceedings of Laser scanning '09 : ISPRS Vol. 38 prt 3/W8*, Paris, 2009.

- [50] T. Santos, N. Gomes, S. Freire, M. Brito, L. F. Santos and J. Tenedórioa, "Applications of solar mapping in the urban environment," *Appplied Geography*, vol. 51, no. 2014, pp. 48-57, 2014.
- [51] A. M. Martín, J. Domínguez and J. Amador, "Applying LIDAR datasets and GIS based model to evaluate solar potential over roofs: a review," *Energy*, vol. 3, no. 3, pp. 326-343, 2015.
- [52] P. J. Hirschfeld, "Vector and scalar field," - 2007. [Online]. Available: http://www.phys.ufl.edu/~pjh/ teaching/phz3113/notes/week5.pdf. [Accessed 11 Novermber 2017].
- [53] D. Girardeau-Montaut, "Cloud-Compare," 2015. [Online]. Available: http://www.cloudcompare.org/doc/ qCC/CloudCompare%20v2.6.1%20-%20User%20manual.pdf. [Accessed 5 2 2018].
- [54] M. S. Roudsari and P. Michelle, "Ladybug: a parametric environmental plugin for grasshopper to help designers create an environmentally-conscious design," Lyon, 2013.
- [55] M. Zwierzycki, H. L. Evers and M. Tamke, "Parametric architectural design with point-clouds Volvox," in Complexity & Simplicity : 34th eCAADe Conference, Oulo, 2016.
- [56] E. Lin and C. Girot, "Point Cloud Components Tools for the Representation of Large Scale Landscape Architectural Projects," in *Digital Landscape Architecture*, Zurich, 2014.
- [57] P. Ken, "3D Object representations," Glasnost, 22 10 2012. [Online]. Available: http://glasnost.itcarlow. ie/~powerk/GeneralGraphicsNotes/meshes/polygon_meshes.html. [Accessed 20 9 2017].
- [58] I. Tchoukanski, "Triangulated irregular Network," ET Solution Center Index, - -. [Online]. Available: https:// www.ian-ko.com/ET_SolutionCenter/sc_faq.htm. [Accessed 15 Sepember 2017].
- [59] D. Chauhan, "Tooltip update in sunlight hours," Ladybug-tools, 24 1 2017. [Online]. Available: https:// github.com/mostaphaRoudsari/ladybug/pull/342/commits/b2f775e6f90ee0e9884cb527bf60b0b867. [Accessed 20 7 2017].
- [60] G. Kigle-Boeckler, "Measurement of Gloss and Reflection Properties of Surfaces," *Metal Finishing*, vol. 93, no. 5, pp. 28–31, 1965.


4 Exploring Potential Applications Of Geometric And Radiometric Information Stored In Point Cloud Data

This chapter has been published as: Alkadri, M.F., Turrin, M. and Sariyildiz, S., 2019. A computational workflow to analyze material properties and solar radiation of existing contexts from attribute information of point cloud data. *Building and Environment*, 155, pp. 268-282. The layout has been adjusted to fit the template of this thesis.

The geometric aspects of point cloud contributed to the design context for the solar envelopes simulation in the previous chapter. It has succeeded in including relevant surrounding site properties (e.g., vegetation) to be considered as geographic inputs from the existing context. However, full features of point clouds have yet to be explored optimally due to the limited capacity and features of the ALS dataset regarding accuracy, visual representation, and isolated area coverage. Thus, this chapter investigates further possibilities for attributes information stored in the point cloud such as geometric and radiometric properties based on TLS (Terrestrial Laser Scanning) datasets. This chapter specifically explores surface materials from the existing context by computing optical (e.g., reflectivity, translucency) and thermal (e.g., albedo, emissivity) properties of contextual datasets. It also simulates solar radiation and develops a material database used to establish an integrated environmental simulation based on the TLS dataset. This study allows architects to further conduct contextual analysis by identifying the potential and impact of the design context during the early stage of the design process. The computational workflow generated in this chapter provides many relevant design features for inclusion in solar envelope simulations, especially those related to material properties and surface characteristics of the surrounding context. Furthermore, these aspects significantly contribute to the development of a methodological framework in Chapter 5, especially for generating the SOLEN model in Chapter 5-Part A and the SHADEN model in chapter 5-Part B.

A computational workflow to analyse material properties and solar radiation of existing contexts from attribute information of point cloud data

- This paper investigates a prospective application of point cloud data in supporting ABSTRACT the contextual analysis of the built environment during the conceptual design process. Often, the complexity of site information causes architects to neglect several relevant properties that may affect environmental performance analysis, especially when dealing with a complex design case. For example, the current approaches of 3D site modelling lack an understanding of the site characteristics of existing environments with respect to either geometrical or material properties. With the advancement of 3D laser scanning technologies, capturing complex information from real contexts offers great possibilities for architects. From geometric and radiometric information stored within point cloud data, this study specifically proposes a novel approach to contextual analysis that considers material aspects and simulates solar radiation in the real environment. In doing so, three computational stages are developed. First, the correction of a raw dataset is designed to not only minimize errors during the scanning process but to also clean the selected dataset. Second, material exploration and the simulation of solar radiation are respectively used to calculate material properties and solar energy in the existing built environment. Third, an integrated environmental simulation aims at identifying materials found in existing areas within a certain level of insolation. As a form of design decision-making support, the present study ultimately generates a computational workflow for analyzing the built environment from which architects may conduct a comprehensive analysis of an existing context before initiating design exploration.
- **KEYWORDS** site analysis; point cloud data; attribute information; material properties; solar radiation

4.1 Introduction

The conceptual design stage is of the utmost importance to produce the most significant decision of architectural design process [1]. Within this stage, contextual analysis plays a prominent role in identifying relevant information for design input. Architects should be able to measure the environmental impacts of the proposed design to maintain the quality of the existing context. However, due to the complexity of information related to a building site, understanding the characteristics of the existing context comprehensively remains a great challenge when dealing with a complex design case. In most cases, existing approaches to 3D site modelling (e.g., solid modelling [2] [3]) pose several barriers that may result in the following features missing from the contextual analysis.

- The complex geometry of the existing context is not considered, especially when dealing with isolated and dense areas [4]. Consequently, it will often take too much time for modelling construction to consider detailed site properties.
- Building-focused contexts tend to be explored, which consequently neglects surrounding properties such as vegetation or other temporal site elements [5] [6]. This matters when conducting environmental simulations of processes such as the UHI (urban heat island) effect or when addressing microclimatic issues.
- Material characteristics of the existing environment [7] [8] have yet to be considered properly in 3D site modelling. This can greatly affect the calculation of environmental impacts for both proposed and existing buildings. In conventional design processes, material studies are frequently paid less attention to during the form generation process, and most of the material attributes geometrically applies when adapting architectural components (e.g., concrete for walls or ceramic tiles for flooring) [9] [10]. Consequently, it will be challenging to identify different material properties within a single component due to the use of fixed materials. As a matter of the fact, such material properties will be extremely likely to present the same values for different architectural components due to being positioned within the same environmental settings.

On the other hand, the advancement of 3D laser scanning technologies has enabled one to easily capture complex information from real contexts. A potential application of point cloud data may include making information relevant to the aforementioned gaps available. In this case, a point cloud consists of geometric and radiometric information [11] [12] or may refer to visible and invisible properties, respectively. Visible properties are stored in a colour point cloud, which geometrically represents a real 3D model of the existing environment. Meanwhile, invisible properties include attribute or metadata information associated with each record of measurement [13]. On the basis of point cloud data, this study proposes an approach to environmental performance analysis that involves an integrated workflow of material studies and the solar simulation of an existing context. The present investigation of material aspects aims at mapping surface properties of the existing environment with the support of radiometric information while geometric information may cater to the simulation task of solar radiation. With this integration, the novel method proposed in this work makes several contributions to the current workflow of the architectural design process:

- The attribute information of point cloud data (position-XYZ, color-RGB, and reflection intensity-I) allows one to easily conduct an environmental analysis during the conceptual design phase, especially in regard to the material performance of existing environments. After investigating optical and thermal properties, a surface distribution catalogue can be generated to identify the material characteristics of existing areas. This can be helpful to architects wishing to perform comprehensive site analysis before making a design decision.
- A point cloud-based solar simulation can effectively mitigate geometric concerns related to the 3D modelling context. Relative to the existing approach, the solar simulation method can be performed without necessarily converting a complex dataset of point clouds into a massive 3D mesh model. This process can be used to identify further opportunities to run more environmental simulations directly based on an unstructured point cloud dataset. The inclusion of more site properties during a simulation may also enrich environmental simulation results.
- Integration between a material database and insolation values allows one to measure the specific performance of any surface within an existing dataset. Accordingly, architects are presented with a fully informed site database that can be used to identify vulnerable areas that may affect the performance of the proposed design. In this case, architects not only need to balance the geometrical relationship between surrounding buildings and the proposed design but must also balance surface materials (performance of the building skin) between them.

This work is divided into several sections. Section 4.1 presents an overview of the study, describing information missing from current architectural design approaches and following with a note on the relevance of point cloud data for addressing the highlighted issues emerging during site analysis. Section 4.2 presents a theoretical background that specifically investigates potential applications of point cloud

data to material properties, and solar radiation analysis. An in-depth discussion of computational design methods is then given in Section 4.3 Our dataset collection approach and a detailed analysis our findings are given in Sections 4.4 and 4.5, respectively. Finally, Section 6 presents a conclusion and recommendations for future study.

4.2 Theoretical background

As a facet of design decision-making support, we principally attempt to perform a contextual analysis taking into account the real environment. In doing so, point cloud data are used to provide relevant information in support of the two primary subjects of this study: material properties and solar radiation analysis. Raw dataset correction is also briefly discussed as a guide to dataset preparation before processing the next stage. The present section will describe specific features of these subjects through the following discussion.

4.2.1 Point cloud data

Geometric information

As a product of laser technology, point clouds have been widely implemented in many fields such as building construction [14] [15] [16], landscape modelling [17] [18], cultural heritage [19] [20] [21] and environmental engineering [22] [23] [24]. Weinmann [25] describes a point cloud as having two parts: the "point" geometrically speaking represents the particular unit of location in a specific space and the "cloud" refers to the unorganized arrangement within a blurred spatial boundary. Otepka et.al. [26] define a point cloud as a set of points that are attached to three-dimensional Cartesian space. As an extension of these definitions, Randall [27] exemplifies several particular functions of the geometric information of point cloud data as follows:

- To draw the position of an object with 3D coordinates in a global coordinate system.
- To illustrate geometrical features such as the length between two points and the orientation of objects.
- To illustrate visual information such as 2D and 3D digital images and 3D virtual models.
- To support physical features such as textures, cracks and different types of objects of varying density [28].

These functions, furthermore, create an opportunity to not only cater to data visualization but to also drive precise simulation performance [29]. For example, with the support of 3D colour point cloud data, some existing studies of archaeology and heritage [30] [31] have successfully investigated the geometric aspects of building materials. However, these material studies are predominantly performed on the basis of the massive 3D mesh model. This approach, unfortunately, presents several important issues, including the means of reconstructing point cloud data to a 3D mesh model, which consumes large volumes of computational storage, and the scale of the 3D mesh model for archaeology, which primarily deals with smaller objects than those of architectural building or urban scales. Thus, it is critical to develop an efficient method that can handle the dataset processing of architectural contexts without compromising the quality of performance analysis.

Radiometric information

The entity of radiometric information refers to attribute properties of each point within a dataset. Richter [32] notes six values of attribute information that can be further investigated to provide an accurate representation of a point cloud: color [33], object class information consisting of vegetation [34] and terrain [35], surface normal [36], horizontality [37], and global and local height [38]. However, only a few attributes are practically relevant during architectural design practice. The issue is not rooted in mere design objectives; rather, it originates from the prerequisites of particular attributes. Prior knowledge and pre-processing steps are still needed to extract the full functions of information properties. It is necessary, therefore, for architects to identify the feasibility of using attribute information that fits into the domain of a design framework.

To examine the context of environmental performance analysis in greater detail, this study focuses on the exploration of typical attributes of point cloud data, including coordinate positions (XYZ), colour information (RGB), and reflection intensity (I) [39] [40] [41]. Each of these attributes supports different tasks. For example, colour information is used to extract a certain area or object according to its colour

values. This possibility can be observed during the identification of road signage by converting RGB colour into HSV values [33]. Similarly, Ochmann et al [42] examine colour and surface normal attributes to identify the inner surfaces of room volumes. Meanwhile, the reflection intensity constitutes the return strength value of the laser pulse of each recorded point, which represents degree of reflectivity measured from scanned objects [43]. In other words, intensity values rely heavily on the surface properties and materials of scanned objects [44]. Practical implementation can involve the detection of damaged concrete in a tunnel, pavement lines or stripping; the mapping of the seafloor, and the mapping of geological layers and damage caused by a natural disaster [12]. Furthermore, position information serves as an index of the coordinate location of each point that automatically aligns to both colour and intensity values of a dataset. For this reason, it is feasible to select certain areas of the dataset according to its attached values.

The attributes mentioned above explicitly facilitate the extension of the particular performance of point cloud data, to not only facilitate impressive 3D visualizations but to also drive environmental analysis during the conceptual design stage. When it comes to the site analysis process, these attributes may be able to enhance the sensitivity of site investigations, especially of those dealing with microclimatic issues affecting a dense area.

4.2.2 Correction of the raw dataset

As end-users of 3D scanning technology, architects are responsible for not only understanding relevant information from a dataset but for also identifying factors that may affect a dataset during scanning such as environmental conditions, meteorological conditions, atmospheric pollution [45] [46], scanner mechanisms, object properties, and scanning geometries [47]. In this case, point cloud data usually present technical issues related to dataset transformation and computational processing. For example, the TLS (terrestrial laser scanner) dataset often consists of a highly dense dataset such that it must be employed using a powerful workstation [48], and the interoperability the dataset results due to the use of different formats, quality levels and units among default scanners [49]. Such aspects must therefore be managed beforehand to allow for dataset processing.

After establishing the selected dataset, the correction of a raw dataset involves intensity correction. This aims at examining nearly "true" values of intensity as a means preventing erroneous measurements from being made during scanning (e.g., sensor noise, hardware sensitivity, laser wavelengths, and the surface geometry of

the target) [50] [51]. In addition, Kashani et al [12] specifically describe several parameters that influence intensity values of a raw dataset such as target surface characteristics, acquisition geometries, and instrumental and environmental effects. These factors, however, cannot be fully compensated for due to local constraints such as atmospheric conditions (humidity and temperature pressure levels) and default features from the manufacturer. Consequently, manual adjustment will still be required even though this is challenging for common users such as architects.

This study specifically applies an intensity correction to a raw dataset by focusing on the acquisition geometry from the angle of incident (α). The angle of incident is the angle between the surface normal vector and incident radiation vector [52] [53]. One can be observed from the oblique surface of a building that produces more backscattering cross-sections than a direct surface when struck with a laser beam [12]. As the TLS dataset largely depends on the tools and manufacturers of laser scanners [52], distance effects are not considered further in this case. The distance between the instrument's position and scanned areas also adheres to a tolerated distance between the brightness-reducer and scanner.

4.2.3 Material properties

In the preliminary design stage, a crucial task facing architects is to identify environmental impacts on the existing context that may affect a new building and vice versa. In relation to this, the potential application of point cloud data progressively enables one to investigate materials in the existing environment. Some publications have attempted to address surface material properties by making use of RGB colour and intensity values taken from TLS datasets. Some of these studies perform non-invasive material analysis through the measurement of chemical properties and from the albedo of scanned objects [54], and investigations of reflecting surface properties based on effects on the maximum range and the range delay error [55], and based on different surface conditions (e.g., wetness and darkness) [56] [57] [58]. Material properties can also be found from LiDAR (light detection and ranging) datasets through the BRDF (bidirectional reflectance distribution function) model and intensity data [59], and through the use of lidar return density values in identifying water bodies [60]. The objectives of these works, however, predominantly focus on the geometric accuracy of surface characteristics (roughness and reflectance) of different material samples determined from indoor laboratory experiments while in this study material properties are investigated by taking into account both optical and thermal properties according to a dataset of a real building context. Moreover, we perform intensity corrections on a raw dataset

to compensate for factors related to data acquisition geometries, which are mostly absent from the existing material studies of TLS datasets.

Furthermore, Kigle-Boeckler [61] describes four key elements in defining surface properties of objects, including materials (e.g., coating, plastic, or metal), surface topographies (e.g., smooth, rough, or structured), the degree of transparency, and the substrate. Based on these items, the present study examines surface materials of a real building dataset by focusing on two aspects: thermal and optical properties.

First, thermal properties demonstrate the quality or attributes of a material, which shape the response process when dealing with the conductivity of heat or with heat fluctuations over time [62]. For the dataset of a point cloud, the following indicators are measured to facilitate the identification of surface materials.

Albedo

Albedo refers to the fraction of sunlight reflected from a material or surface [63] [64] [65]. Albedo values have a significant impact related on the energy balance of an urban environment, as they constitute absorbance percentages of solar energy. As such, albedo can be used to describe the environmental characteristics of certain areas. Albedo values range between 0 and 1 where a lower value corresponds to a blackbody [66].

Emissivity

According to Ashby et al [67], emissivity refers to heat radiation emitted by the surface of a material. It is also defined as the ratio of radiated light reflected from a material to the volume emitted to a blackbody for the same temperature, wavelength, and emission direction [68]. In contrast to emissivity data, intensity information of point cloud data takes a value of 1.0 for a bright (white) surface or perfect reflector while a perfect emitter (black) is assigned a value of 0. Examples include materials with smooth and shiny surfaces such as plastics, ceramics, water and polished metals, which are generally characterized by high-intensity values but low levels of emissivity [69]. While in principle both emissivity and intensity determine their values contrarily, they consider a similar aspect related to the material of the object's surface. In so doing, the determination of emissivity values can be identified from corrected intensity information.

Second, materials with optical properties are principally determined through their interactions with light or electromagnetic radiation. Regarding optical properties this study focuses on investigating reflectivity and translucent values of materials by taking into account the RGB colour of each point within a dataset. Section 3.1 presents a detailed calculation of these values.

Reflectivity

In general, reflectivity describes the amount of light reflected from a material in relation to the total amount of incident light that reaches the surface of a material [67] [68]. Marsh [70] furthermore distinguishes reflectivity from reflectance. As a boundary property, reflectivity refers to a layer between two particular types of objects while reflectance corresponds to a layer between two specific instances of objects. Nevertheless, in the case of opaque materials, both reflectivity and reflectance are of equal value except in the case of transparent objects, as internal reflections within such materials appear to have a considerable effect. In accordance with a given point cloud dataset, the calculation of reflectivity values in this study predominantly covers opaque materials due to laser scanning capacities. With current applications of laser scanning, a fully transparent record of objects results in a coarse cloud of points within a dataset. This simultaneously confirms the scope of this study on reflectivity properties.

Translucency

An investigation of translucent properties of a material inevitably also involves the calculation of its transmittance values. Transmittance is a material property reflecting a material's capacity to transmit light received through the material itself [70]. In turn, opaque materials should present a zero transmittance value as they cannot be penetrated by light. In regard to a point cloud dataset, one can calculate transmittance values of materials from colour information for each point. Following from this assumption, as long as a dataset contains RGB colour, the translucence of materials can be identified in parallel with a comparison of its reflectance values. Such a calculation is also supported through material mapping with ElumTools [71] when transmittance values are added to opaque material properties, allowing the selected objects to be categorized into a translucent material.

4.2.4 Solar radiation analysis

An insolation analysis involves evaluating how building surfaces absorb thermal energy from the sun [72]. During the conceptual design, evaluating the relationship between a building and the sun is critical to ensuring the behavioral performance of buildings based on optimal sunlight conditions. This process ultimately becomes relevant not only for a proposed building but it can also be extended to existing buildings through contextual analysis. For example, the simulation of solar radiation allows us to map the average solar energy of each panel on a building's façade. In turn, the potential amount of solar radiation reaching a building or site surfaces can be identified and further analysed.

However, as the issue that matters most for current simulation approaches to solar radiation, the 3D digital model of the existing context is primarily constructed by means of basic architectural geometric shapes. In regard to surface characteristics of buildings, the exclusion of textures from the context model and of other relevant site properties can considerably affect the interpretation of simulation results [73]. Things become even more complex when dealing with the 3D model for an isolated context or urban scale forms. Horvat and Dubois [74] reveal that less than 10 % of architects are satisfied with the utilization of various solar simulation tools. It is thus, important to at least consider surface properties of the existing environment during the simulation of solar radiation.

In the meantime, some existing works have conducted solar radiation analysis by making use of point cloud data. For example, in using a LiDAR DEM (digital elevation modelling) dataset, Kassner et al. [23] attempted to identify areas of high solar potential on a building's roof and the suitability of a PV (photovoltaic) system for urban areas by means of pyranometer measurements [75]. Carneiro et al. [24] and Jochem et al. [76] develop solar radiation models in urban areas to deliver an automated solar potential assessment. Similarly, an interactive 3D visualization of the solar urban model [77] has been generated to calculate the solar irradiance distribution based on roof and building facades [78]. Later, Pavlovski et al. [79] from a solar resource campus map and Li et al. [80] in using a pixel-based approach quantify solar energy stored within an existing building's roof and infrastructures. As a major concern related to these approaches, their workflows are inclined towards the domain of environmental engineering, which predominantly uses ALS (Airborne LiDAR) datasets. Meanwhile, studies have yet to focus on the use of TLS datasets rather than ALS datasets due to accuracy concerns [81] [82], high-resolution formats [27] [83], and broader coverage of isolated areas. Moreover, integration with the architectural design process has yet to be addressed and particularly in relation to site analysis based on TLS datasets. The present study ultimately attempts to address these issues by proposing a computational design method in the following section.

4.3 Development of the computational design method

As introduced in the previous section, the ultimate goal of this study is to develop an integrated workflow of environmental performance analysis for material studies and solar simulations by taking into account point cloud data for the existing environment. A series of computational procedures is developed to support this goal (see Figure 4.1); this involves the use of existing frameworks such as point cloud dataset processing and the normalization of intensity values in a first stage and then calculating material properties and sun angles in a second stage. In line with this, several specific frameworks are developed by the authors, including material database formation and the calculation of insolation values of point cloud data in a second stage and the material selection of point clouds based on insolation values in a third stage.



FIG. 4.1 Overview of the computational workflow

To illustrate the proposed workflow, a detailed discussion of inputs, procedures, digital tools, and outputs of each stage are addressed below.

4.3.1 Correction of the raw dataset

As noted in Section 4.2.2, in this study we perform raw dataset corrections. This process mainly involves two tasks: pre-processing and intensity correction. Figure 4.2 presents a detailed account of this stage.



FIG. 4.2 Correction of the raw dataset 3D models

Before the dataset is readily used for architectural design analysis, several preliminary steps are completed to avoid irrelevant properties that may compromise workflow performance. These steps involve noise reduction and outlier (unnecessary clouds of points) removal, the selection of attribute order and dataset formats due to interoperability issues emerging during analysis, and the activation of scalar field functions to identify metadata information for a point cloud.

These preliminary tasks are computationally supported by Cloud Compare (CC) [84]. One task involves reducing the density of points to manage computation time and calculation requirements of the dataset.

For intensity corrections (I_c), we use the following equation [12].

$$I_c = I_{raw} \cdot \frac{1}{\cos \alpha} \tag{1}$$

where I_c = corrected intensity I_{raw} = original intensity α = angle of incidence

The angle of incidence is determined with equation [50] below based on the assumption that the initial positioning of the laser scanner is known at (0,0,0).

$$i = \cos^{-1} \left(\frac{\overline{dn} \cdot \overline{dl}}{\left| \overline{dn} \right| \cdot \left| \overline{dl} \right|} \right)$$
(2)

where i = incident angle (a) \overline{dn} = direction of surface normal \overline{dl} = direction of the laser pulse

Before applying these algorithmic functions for intensity corrections, the raw intensity values must first be normalized. This procedure is designed to calibrate the numerical index of the original dataset during arithmetic operation.

Material exploration

This stage involves two main steps (see Figure 4.3): *first*, material properties (thermal and optical properties) derived from attribute information such as XYZ, RGB, and corrected intensity values are calculated, and *second*, a material database derived from the calculation of material properties is developed. Material selection can then be performed with the material database according to the established criteria.



FIG. 4.3 Material exploration of point cloud data

As described in Section 4.2, thermal properties include albedo and emissivity values while optical properties include reflectance and transmittance values. On the basis of colour information of point cloud data, reflectance and transmittance values are calculated by referring to a graphics colour taken from the ElumTools library [71], which is also used as a reference for material and artificial lighting simulations in Revit [85]. Reflectance and transmittance values are respectively determined from the following equation

$$\operatorname{Ref} = \left(0.2125 \cdot \frac{R}{255}\right) + \left(0.7154 \cdot \frac{G}{255}\right) + \left(0.0721 \cdot \frac{B}{255}\right) \tag{3}$$

$$Max.Trans = (0.2125 \cdot R_{avail}) + (0.7154 \cdot G_{avail}) + (0.0721 \cdot B_{avail})$$

$$\tag{4}$$

$$R_{avail} = \frac{R}{255}$$
$$G_{avail} = \frac{G}{255}$$
$$B_{avail} = \frac{B}{255}$$

(5)

where Ref = the reflectance value Max.Trans = maximum transmittance R_{avail} = the maximum amount of light reflected from the red component G_{avail} = the maximum amount of light reflected from the green component B_{avail} = the maximum amount of light reflected from the green component

To obtain maximum transmittance values, the most heavily reflected component should be initially identified based on non-dichroic transmittance. The resulting component is first subtracted by 1 to find the ratio of transmitted value to reflected light. This ratio is useful as a multiplier in calculating the relative amount of light for the other colour components. Accordingly, maximum transmittance values can be calculated by performing the mathematical operation given in equation (4). Emissivity values are determined from the corrected intensity attributes as both emissivity and intensity based on similar material aspects related to surface reflectance. In this case, however, their values principally work in reverse.

Furthermore, albedo values are computed from both colour information and corrected intensity. As is exemplified in equation (6), the calculation aims at not only measuring light reflected from the average of RGB colours but also capturing incident solar energy through the multiplication of corrected intensity values.

$$Alb = \left(\frac{\sqrt{\left(\frac{R^2 + G^2 + B^2}{3}\right)}}{255}\right) \cdot I_c$$

(6)

where Alb = the albedo value R = the red value G = the green value B = the blue value $I_c =$ the corrected intensity value After establishing formulas for all material properties, each point of a dataset can readily be inputted into the material database. This material database will then evaluate each point in a conditional loop according to the threshold value taken from optical and thermal properties. Due to a lack of compatibility between existing material libraries (e.g., Honeybee-Grasshopper includes Radiance, Open Studio 1.12.0, Energy Plus V8-5-0, THERM 7.6 [86] and CES Edu Pack 2017 [67]) and the proposed method, we propose a sample of relevant materials that matches the conditions of the selected dataset (see Table 4.1).

TABLE 4.1 Material database for the outdoor building context (conected from various sources)								
Index	Material types	Material properties						
		Emissivity		Reflectance	Transparency			
0	Brick	0.93	0.4	0.2	Opaque			
1	Asphalt	0.9	0.12	0.07	Opaque			
2	Stone (granite)	0.44	0.30	0.25	Opaque			
3	Stone (white marble)	0.75	0.6	0.45	Opaque			
4	Wood	0.8	0.35	0.5	Opaque			
5	Plaster	0.9	0.4	0.65	Opaque			
6	Cement	0.54	0.8	0.4	Opaque			
7	Ceramic tile	0.5	0.35	0.72	Opaque			
8	Glass with a zenith-angle of 40-80 degrees	0.95	0.52	Transmittance – 0.75	Translucent			

TABLE 4.1 Material database for the outdoor building context (collected from various sources)

Discrepancies between existing material libraries can be observed from computational workflows and materials formats. For example, in Radiance, material setup involves using a specific type of material at the beginning of the workflow, which is then followed by material properties. By contrast, the method proposed in this work involves identifying materials at the end of the workflow as the output according to the selection process used for the point cloud dataset. Considerable efforts have been made to compensate for this issue by inputting material properties in CES Edu Pack beforehand. This approach generates several types of materials that correspond with input properties. However, a further consideration not only concerns the availability of material properties but also the complex range of material types that are to some extent seemingly irrelevant to the proposed context. For example, material selection in CES Edu Pack concludes with formation of many unexpected variations of specific engineering materials. Such a result would produce biased outputs from this study, which is intended to deal with architectural materials used for outdoor building surfaces.

Simulation of solar radiation

Due to the requirement of similar inputs from point cloud data (see Figure 4.1), the proposed process (stage 2b) is based on a similar level as that of stage 2a focused on material properties. As such, a portion of the employed dataset also takes selected points from the corrected intensity values. The workflow of our simulation of solar radiation is illustrated in Figure 4.4.



FIG. 4.4 Solar radiation simulation workflow

In general, the simulation of solar radiation involves the use of two main inputs: the sun's direction vector from the environmental variable and normal vectors of the site surfaces. The calculation of the sun vector involves the use of climate and geographic inputs such as latitude, longitude, and the specific point of time of given a year. Specifically, for this study this process is performed by using components available through Grasshopper, some of which include Ladybug and DIVA, but of course this can be done with other tools. Meanwhile, normal vectors of site surfaces are computed in CC due to the large number of points to be calculated.



FIG. 4.5 Calculation of solar irradiance (INC: angle of incidence, G: global irradiance) [87]

With a trigonometric principle (see Figure 4.5) the insolation values of each point of a dataset can be calculated from the normal irradiance (the angle between sun vectors and normal vectors of the site surfaces) multiplied by the cosine of the angle of incidence (INC). In principle, solar energy is fully absorbed by the surface when the sun vector and the normal vectors are aligned. To reflect this, the cosine value of the angle will be 1 when the angle formed is zero such that as the angle increases, the cosine value will decline.

Furthermore, the following process requires the use of a synchronization value between the simulation result of the insolation analysis and the index of the coordinate position from the initial dataset. This allows for the identification of a surface's degree of direct sun exposure for the point cloud dataset. With a colour 3D point cloud, the lowest and the highest degrees of direct sun exposure can then be visualized to each point of the dataset. For outputs the proposed workflow produces a 3D point cloud model aligned with an ASCII file that presents a simulation analysis of the dataset.

Integrated environmental simulation

This stage involves investigating an integrated workflow between a material database and a simulation of solar radiation. It aims to not only identify materials in areas exposed to a certain level of direct sunlight but to also generate a highly informed site database through a fully integrated simulation analysis conducted in the preliminary design stage. The following figure describes the process of the integrated workflow (see Figure 4.6).



FIG. 4.6 Integrated simulation workflow

The main procedure of the workflow focuses on material selection according to material properties of the point cloud dataset. In this workflow, insolation values are integrated with material types such that surfaces subjected to high or low levels of solar exposure can be identified and extracted along with their materials. In this case, a numerical index of each point within the dataset plays a crucial role in synchronizing metadata between material properties and solar simulation results.

Ultimately, a list of insolation values corresponding with material types is the final output of the proposed integrated workflow. An evaluation of the whole dataset will generate the number of points that meet and do not meet the criteria of the material database. Accordingly, a comprehensive analysis of the environmental performance of the existing context can be conducted. To implement the proposed computational workflow, a sample 3D colour point cloud dataset is used in the following section.

4.4 Dataset collection

The proposed approach is implemented on a selected small sample of 3D point cloud data (see Figure 4.7). Several criteria are used during dataset collection, some of which include the following: *First,* the dataset consists of a 3D colour-point cloud. Colors can help architects not only visualize the representation of real objects in the digital environment but also analyse the appearance of materials and textures during simulation. *Second*, the dataset includes information on at least three typical attributes as mentioned above: XYZ, RGB, and reflection intensity (I). *Third*, the dataset includes multiple site properties for the sake of material property investigation. After setting these criteria, dataset collection can be applied to any architectural context.



FIG. 4.7 The selected 3D colour point cloud dataset

The collected dataset is for a Middelstum Church located in Groningen, the Netherlands. Its geometric properties include a building facade, trees and a surrounding landscape with a total number of points spread across 31,5 million points. Via cloud sub-sampling in CC, the minimum distance between two points can be adjusted to limit the density of points. In this case, the dataset is ultimately set at 5 cm, producing roughly 449.267 points from the total collection of points. Dataset collection is supported with a Faro Focus 3D laser scanner set to a wavelength of 950 nm. We also use a Nikon D5300 DSLR camera for the processing of the 3D colour-point cloud. Having established the selected dataset, this section discusses the implementation of the proposed workflow. The analysis results are presented under four categories: intensity correction, the surface distribution of material properties of the dataset, simulation results of the insolation analysis, and material selection for each selected point cloud dataset. These are presented in the chronological order of the proposed workflow described above.

4.5.1 Intensity correction

During dataset preparation it is important to first calculate the incident angle before determining the corrected intensity (I_c) . For an unstructured cloud of points, the surface normal of the points is calculated by using a Hough Normals plugin [88] in CC. This condition requires that such a task identifies optimal normal values in the dataset. Several tolerance angles are then simulated into raw intensity values of 10° to 90° (see Figure 4.8).

Figure 4.8 shows the dispersion of points among cosine values from the range of angles. It shows that points can be scattered at a certain angle. However, the majority of dense points seemingly make up a small portion of each plot of cosine values. It can be said that some points on a certain surface might correspond very well to a laser beam projected at a certain angle during scanning. Roughness and material characteristics of the surface also play an important role in this condition. Furthermore, an evaluation of the scattered points is run by reducing them while maintaining the densest one. In doing so, a standard deviation of each cosine value is plotted to identify the density distribution of points within the dataset (see Figure 4.9).



FIG. 4.8 Distribution of points according to the angle of incidents



FIG. 4.9 Distribution of points according to the standard deviations of different cos values

According to the distribution of points shown in Figure 4.9, the proximate truncation of the dataset is set to range from 0 - 0.015. What results are roughly 266.864 points or 59.3 % of all points. As such, intensity corrections can be calculated using equation (1). Figure 4.10 below presents a comparison between the raw dataset and the corrected intensity values



FIG. 4.10 Comparison of the (A) Original dataset and (B) Corrected intensity

The following aspects describe major changes in the dataset's performance resulting from the initial intensity level:

- As is illustrated by certain RGB colour features, several geometric properties of the original dataset (specifically trees and edges along the building) disappeared. In this case, intensity correction mostly affected not only the scattered points but also the uneven surfaces. This result simultaneously provides us with enough insight to temporarily exclude vegetation properties during the intensity correction phase, as vegetation will considerably affect all dataset values during the calculation of material properties and during the simulation of solar radiation.
- The colour value of intensity also undergoes significant transformation. Four colour steps are set to facilitate the legibility of the intensity range. As a bright colour, the color yellow represents a high-intensity value that concurrently refers to the most reflective surface while the color blue denotes the opposite. In Fig. 10-A (see part of the scalar field) the most reflective surfaces are predominantly distributed across the wall of the building. This intensity distribution reveals a major discrepancy from Fig. 10-B based on tree trunks located on reflective surfaces. In general, this serves as a crucial aspect of surface material properties to consider when determining intensity values particularly dealing with real building contexts.
- According to the histogram results, the distribution of intensity values observed within the original dataset concentrates at a green color representing approximately 89 % of the total density of the points. This means that green values are scattered across the entire surface of the dataset. On the other hand, from the calibration of the incident angle of the scanner and the normal vector of the datasets, the corrected intensity values ultimately present an obvious boundary between areas of high (yellow), moderate (green) and low (blue) intensity values.

4.5.2 Surface distribution of material properties

This step involves mapping material properties of the surface of the dataset in parallel with the calculation of point densities of each identified surface. With information on surface materials architects can measure the specific performance of certain areas within an existing dataset. Figure 4.11 presents the surface distribution of the selected material properties within a range of 0.0 - 1.0.

In general, the point density of surface materials follows a similar pattern as property ranges of the largest and smallest values. The three material properties of reflectance, emissivity, and albedo are measured within an equivalent range for the largest values, at 0.0 - 0.3. These constitute 21.4 %, 28.6 %, and 21.2 % of the total density of points, respectively. The only exceptions found for this largest range are transmittance values, with 19.097 % ranging from 0.4 - 0.5. Meanwhile, for the lowest percentages of values, reflectance, emissivity, and transmissivity occupy a similar range of 0.9 – 1.0. Only the albedo values are valued within a range of 0.0 - 0.1. Moreover, translucent properties are evaluated by comparing values of average reflectance and transmittance for each point. This evaluation is performed with a Boolean operation. When the reflectance is larger than transmittance values, the translucent material will be *False* and vice versa. The results show that translucent materials constitute 48.7 % of the points while the remaining 51.3 % have opaque properties. A comparison of these values statistically reveals that surface materials covered within the dataset primarily constitute non-reflective surfaces. This simultaneously shows that other surface characteristics of the existing context mostly include rough textures and high surface temperatures observed during the daytime [89].

A surface material catalogue allows architects to select particular areas to analyse further under an environmental design framework while providing supplementary information on which design treatments to apply to specific areas based on material parameters.

Threshold values	Average Reflectance	Emissivity	Albedo	Transmissivity
0.0 - 0.1	DoP = 1.882 (0.7 %)	DoP = 76.290 (28.6 %)	DoP = 1.600 (0.6 %)	DoP = 27.657 (10.4 %)
0.1 - 0.2	DoP = 44.491 (17 %)	DoP = 26 755 (10 %)	DoP = 50 846 (19 %)	Dop = 37,030 (13,86 %)
0.2 - 0.3	DoP = 57.105 (21.4%)	DoP = 16.549 (6.18%)	DoP = 56.534 (21.2%)	DoP = 35.411 (13.3 %)
0.3 - 0.4	DoP = 26.470 (10 %)	DoP = 7.885 (3 %)	DoP = 26.475 (10 %)	DoP = 44.358 (16.6 %)
0.4 - 0.5	DsP = 25.025 (0.4.%)	DoP = 12 \$69.(4.7.%)		Dop = 50.974 (10.097.94)
0.5 - 0.6	DoP = 26 579 (10 %)	DoP = 63 872 (24 %)	DoP = 20.554 (11.2 %)	
0.6 - 0.7	DoP = 28.182 (10.4%)	DoP = 58.582 (22 %)	DoP = 26.811 (10.1 %)	DoP = 21.311 (8 %)
0.7 - 0.8	DoP = 23.890 (9 %)	DoP = 4.036 (1.5 %)	DoP = 20.729 (7.8 %)	DoP = 1.919 (0.72 %)
0.8 - 0.9	DoP = 17.284 (6.1%)	DoP = 45 (0.02 %)	DoP = 12.985 (4.9 %)	DoP = 62 (0.02 %)
0.9 - 1.0	DoP = 15.966 (6 %)	DoP = 0	Dop = 12.983 (4.9 %)	DoP = 8 (0.003 %)

FIG. 4.11 Surface distribution catalogue of material properties

4.5.3 Simulation of solar radiation

As noted in Section 4.2.3, this step involves quantifying the solar energy of the existing context according to solar radiation simulation results for a point cloud dataset. In this case, the simulation focuses on the period of June – September 2017 (9 am to 8 pm Central European Time) with a coordinate position of 52.1326° N, 5.2913° E. From this setting, 36 values of sun vectors are produced from the sun angle calculation. These vectors are then computed to each surface normal of the point cloud dataset. Accordingly, each point of the dataset is evaluated from 36 sun vectors. The evaluation yields approximately 9.6 million radiation values. The aggregation of the time range will result in robust output values with the effect of consuming time during the simulation process.





Figure 4.12 presents simulation results for incoming solar radiation for the point cloud dataset. The simulation results are visualized based on five colour properties ranging from red to green, which respectively denote high to low radiation values. As is illustrated in Figure 4.12-B, less than 10 % of all radiation values meet the criteria required to absorb solar energy. The domination of green colours across the surface of the dataset indicates a lack of solar energy absorption. Meanwhile, yellow and red points reflect the dataset remaining from green subtraction (see Figure 4.12-B2) and originate from values of 0.8 - 1.0 according to their histogram bar values

Figure 4.12-C furthermore presents average insolation values received by each point of the dataset. During the simulation, only maximum insolation values from each point are counted. The histogram bar (Figure 4.12-C2) shows that yellow colours constitute the largest proportion of insolation values at 0.4 - 0.5 (58 % of all points). On the other hand, red colours as the highest values can only be found in certain areas with low proportion. For example, red points are unevenly distributed along the edges of the building's walls and along tree branches, which are unfortunately less conducive to solar collection technology application.

4.5.4 Material selection with integrated insolation values

This section discusses the workflow of integration for the material database and for insolation values derived from the simulation of solar radiation. With a developed material database, the proposed algorithm evaluates the material properties of the dataset according to a threshold value of optical or thermal properties. This approach allows for the identification of material types in parallel with the total number of points exhibiting optical or thermal properties. The database is furthermore employed to identify materials presenting certain ranges of insolation values.



FIG. 4.13 Material selection according to optical and thermal properties

According to the material selection results (see Figure 4.13), the dataset most likely fulfils the criteria for material properties for the database (see Table 4.1). Unknown materials are found those points that do not fulfil the criteria. Eight of the 9 materials successfully meet threshold values for optical properties while for thermal properties only 6 materials meet the selection criteria. Regarding the density of points, the resulting optical properties of the materials are predominantly registered throughout the whole dataset, which includes over 44.9 % optical properties. Thermal properties represent only 6.7 % of the points.

Further results on this material selection are as follows:

- We find that transmittance criteria heavily affect dataset selection regardless of the number of points of transmittance values. This can be observed, for example, from optical properties, as wood materials quantitatively represent the largest proportion of registered materials, while stone (granite) makes up the lowest proportion. These results can be examined from the material database (see Table 4.1) and by then referring back to a comparison of surface distribution catalogues (see Figure 4.11). In this case, wood and stone (granite) consist of opaque materials with reflectance values of 0.5 and 0.25, respectively. For the range of 0.5, surface distribution values (see Table 4.1) show that a threshold of 0.5 (0.4 to 0.6) accounts for over 19 % of the density in points in terms of average reflectance. Meanwhile, stone materials (granite) (thresholds of 0.2 to 0,3) account for more points at approximately 21.4 %.
- The resulting materials successfully identify a number of points from opaque and translucent materials in terms of both optical and thermal properties.
- The point properties of the dataset confirm similarities among the material types. Accordingly, various objects with different surface characteristics (e.g., roughness and glossiness) can actually be composed of the same materials. This case can be observed from a group of points that fall within several material thresholds. As an example, a point with index 42633 in terms of optical property criteria refers to three types of materials: stone, wood, and cement. Fleming [90] implicitly confirmed this point in that material estimations computationally support a higher fidelity of representation than mere material categorizations.

The final step of the proposed workflow produces a list of materials found within the dataset according to insolation values generated from the simulation (see Figure 4.14). As an input, insolation values of a different colour range (as illustrated in Section 4.5.3) cover various point densities. From our evaluation of the material database the results show that each colour range on both optical and thermal properties covers a large variety of material types. Eight out of 9 materials were successfully identified for both properties. The properties also follow similar trends in point density for each identified material within the insolation colour range. This result indicates that the developed material database positively responds to calculations of material properties acquired from point clouds of the existing context, highlighting further opportunities to modify the environmental performance of certain areas according to a material analysis.



FIG. 4.14 Material selection according to insolation value simulation results

Figure 4.14 shows that the highest material distributions in terms of optical and thermal properties are respectively found for wood (index 4) and glass (index 8), though yellow colours also denote the presence of cement (index 6) in terms of thermal properties. The least common values are found for stone-granite (index 2) in terms of optical properties and for asphalt (index 1) and white stone-marble (index 3) in terms of thermal properties. From the material performance within these properties, environmental characteristics of the context examined in this study can be hypothetically addressed. For example, a massive discrepancy in insolation values between the largest (denoted by light and dark green areas with a point density of over 70 %) and smallest portions (denoted by red and orange areas with a point density of only 21 %) may affect how the environmental context generates ambient temperature. As Ramirez and Munoz [89] point out, areas absorbing less energy denote high albedo values. It can then be useful to moderate the urban heat island effect. This means that materials within 70 % lower insolation values can be substituted with materials with high albedo values. This strategy may be used to measure the thermal impacts of microclimate issues on the built environment.

4.6 Concluding remarks and future perspectives

In this study, we perform an environmental performance analysis of an existing context by making use of potential applications of point cloud data. Through material studies and simulations of solar radiation, the proposed computational workflow allows architects to fortify contextual analysis during the initial design stage. Several concluding remarks regarding this process are made as follows:

 The exploration of material properties (optical and thermal) from attribute information on point clouds extends particular functions of 3D scanning technologies related to performance simulation and environmental analysis during the architectural design process. The proposed workflow reveals an additional opportunity to map impacts of the built environment before applying a new building to the selected context.

- Post-processing datasets necessarily help architects not only select relevant information to be employed during analysis but also minimize the environmental effects of dataset measurement during scanning.
- The integrated workflow between the material database and solar simulation may reveal specific characteristics of the existing environment. For example, the identification of highly reflective surfaces or high insolation values of a certain area can help architects to decide whether to maintain or alter the design process.

While the workflow presented in this work presents partially positive results, it requires further consideration in certain respects. For example, dataset corrections may require one to consider more radiometric parameters, especially related to intensity properties such as surface roughness, range, and target reflectance. These parameters are expected to reduce erroneousness of dataset measurements. Storing more relevant materials in a database also allows for a broad range of selection criteria used for the datasets. The material database employed in this work focuses demonstrates the feasibility of applying the proposed workflow for architectural design practices.

With further developments from this study may extend several potential features, some of which may include the use of temperature parameters in defining thermal properties to enrich albedo information on existing contexts or expanding the scale of case studies so that aspects of urban morphology and building functions can be addressed. Finally, using a reflectometer in combination with a thermal camera during scanning might prove useful for verifying and calibrating dataset reflectance values.
References

- [1] S. Sariyildiz, Conceptual Design by Means of Islamic- Geometric-Patterns within a CAADEnvironment., Delft: PhD-thesis, Faculty of Architecture, TU Delft, 1991.
- [2] N. N. Staneva, "Approaches for generating 3D solid models in AutoCAD and solid works," Journal of Engineering, vol. VI, no. 3, pp. 28-31, 2008.
- [3] V. Shapiro, "Solid modeling," Spatial Automation Laboratoty, University of Wisconsin, Madison, 2001.
- [4] M. F. Alkadri, M. Turrin and S. Sariyildiz, "Identifying the surface materials of the existing environment through point cloud data attributes," in *SimAUD (Simulation for Architecture and Urban Design)*, Delft, 2018.
- [5] Y. Arayici, A. Hamilton and P. Gamito, "Modelling 3D Scanned Data to Visualize and Analyse the Built Environment for Regeneration," *Surveying and Built Environment*, vol. 17, no. 2, pp. 7-28, 2006.
- [6] Y. Arayici, "Towards Building Information Modelling for Existing Structures," *Structural Survey*, vol. 26, no. 3, pp. 210-222, 2008.
- [7] I. Yüksek and T. T. Karadayi, "Energy-Efficient Building Design in the Context of Building Life Cycle," in *Sustainable Development*, London, IntechOpen, 2017, pp. 93-123.
- [8] M. F. Alkadri, M. Turrin and S. Sariyildiz, "The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts," *Advances in Computational Design*, vol. 3, no. 4, pp. 319-338, 2018.
- [9] L. Bragança, S. M. Vieira and J. B. Andrade, "Early Stage Design Decisions: The Way to Achieve Sustainable Buildings at Lower Costs," *The Scientific World Journal*, vol. 2014, no. -, pp. 1-8, 2014.
- [10] T. Kotnik and M. Weinstock, "Material, Form and Force," Architectural Design, vol. 82, no. 2, pp. 104-111, 2012.
- [11] S. Kaasalainen, A. K. A. Krooks and H. Kaartinen, "Radiometric Calibration of Terrestrial Laser Scanners with External Reference Targets," *Remote Sensing*, vol. 1, no. 3, pp. 144-158, 2009.
- [12] A. G. Kashani, M. J. Olsen, C. Parrish and N. Wilson, "Review of LIDAR Radiometric Processing: From Ad Hoc Intensity Correction to Rigorous Radiometric Calibration," Sensors, vol. 15, no. 11, pp. 28099-28128, 2015.
- [13] D. A. White, "LIDAR, Point Clouds, and Their Archaeological Applications," in *Mapping Archaeological Landscapes from Space*, New York, Springer-Verlag New York, 2013, pp. 175-186.
- [14] N.-J. Shih, "The Application of A 3d Scanner in the Representation of Building Construction Site," in International Association for Automation and Robotics in Construction (IAARC), Madrid, 2002.
- [15] T. Várady, R. R. Martin and J. Cox, "Reverse engineering of geometric models—an introduction," *Computer-Aided Design*, vol. 29, no. 4, pp. 255-268, 1997.
- [16] Y. Fujita, I. Kobayashi, W. Chanseawrassamee and F. Hoshino, "Application of Attributed Road Surface Point Cloud Data in Road Maintenance," *Journal of Japan Society of Civil Engineers, Ser. F3 (Civil Engineering Informatics)*, vol. 70, no. 2, pp. I_185-I_192, 2014.
- [17] E. Lin and C. Girot, "Point Cloud Components Tools for the Representation of Large Scale Landscape Architectural Projects," in *Digital Landscape Architecture*, Zurich, 2014.
- [18] E. Lin, K. Shaad and C. Girot, "Developing river rehabilitation scenarios by integrating landscape and hydrodynamic modeling for the Ciliwung River in Jakarta, Indonesia," *Sustainable cities and society*, vol. 20, pp. 180-198, 2016.
- [19] T. P. Kersten, F. Keller, J. Saenger and J. Schiewe, "Automated Generation of an Historic 4D City Model of Hamburg and its Visualization with the GE Engine," in *Progress in Cultural Heritage*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [20] M. Balzani, C. Bughi, F. Ferrari, L. Rossato and A. Tursi, "Alberti's Box: The Cultural Multimedia Project on the Architectures of Leon Battista Alberti," in *Progress in Cultural Heritage Preservation*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [21] W. Moussa, M. Abdel-Wahab and D. Frisch, "Automatic Fusion of Digital Images and Laser Scanner Data for Heritage Preservation," in *Progress in Cultural Heritage Preservation*, 4th International Conference, *EuroMed 2012*, Limassol, 2012.
- [22] L. Bornaz, A. Lingua and F. Rinaudo, "Engineering and Enviroonmental Applications of Laser Scanner Techniques," in ISPRS Commission III Symposium "Photogrammetric computer vision", Graz, 2002.
- [23] R. Kassner, W. Koppe, T. Schüttenberg and G. Bareth, "Analysis of the Solar Potential of Roofs By Using Official Lidar Data," in *ISPRS Congress Beijing*, Beijing, 2008.

- [24] C. Carneiro, E. Morello and G. Desthieux, "Assessment of Solar Irradiance on the Urban Fabric for the Production of Renewable Energy using LIDAR Data and Image Processing Techniques," in Advances in GIScience, Hannover, 2009.
- [25] M. Weinmann, Reconstruction and Analysis of 3D Scenes, Karlsruhe: Springer International Publishing, 2016.
- [26] J. Otepka, S. Ghuffar, C. Waldhauser, R. Hochreiter and N. Pfeifer, "Georeferenced Point Clouds: A Survey of Features and Point Cloud Management," *ISPRS International Journal Geo-Information*, vol. 2, no. 4, pp. 2038-1065, 2013.
- [27] T. Randall, "Client Guide to 3D Scanning and Data Capture," BIM Task Group, UK, 2013.
- [28] A. Gressin, C. Mallet, J. Demantké and N. David, "Towards 3D lidar point cloud registration improvement using optimal neighborhood knowledge," *ISPRS Journal of Photogrammetry and Remote Sensing*, vol. 79, no. -, pp. 240-251, 2013.
- [29] V. Salehi and S. Wang, "Using point cloud technology for process simulation in the context of digital factory based on a system engineering integrated approach," in *The 21st International Conference on Engineering Design (ICED 17)*, Vancouver, 2017.
- [30] D. Sidiropoulou-Velidou, A. Georgopoulos and J. L. Lerma, "Exploitation of Thermal Imagery for the Detection of Pathologies in Monuments," in *Progress in Cultural Heritage Preservation*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [31] E. Gigliarelli, D. Carlea, A. Corcella and H. Porfyriou, "Historical Social Housing: Smart Analysis and Design for Conservation and Energy Conservation," in *Progress in Cultural Heritage Preservation*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [32] R. Richter and J. Döllner, "Concepts and techniques for integration, analysis and visualization of massive 3D point clouds," *Computers, Environment and Urban Systems*, vol. 45, pp. 114-124, 2014.
- [33] I. Kobayashi, Y. Fujita, H. Sugihara and K. Yamamoto, "Attribute Analysis of Point Cloud Data with Color Information," *Journal of Japan Society of Civil Enginers*, vol. 67, no. 2, pp. 95-102, 2011.
- [34] B. Höfle, M. Hollaus and J. Hagenauera, "Urban vegetation detection using radiometrically calibrated smallfootprint full-waveform airborne LiDAR data," *ISPRS Journal of Photogrammetry Remote Sensing*, vol. 67, pp. 134–147, 2012.
- [35] A. Brzank, C. Heipke, J. Goepfert and U. Soergel, "Aspects of generating precise digital terrain models in the Wadden Sea from lidar–water classification and structure line extraction.," *ISPRS Journal of Photogrammetry* and Remote Sensing, vol. 63, no. 5, pp. 510-528, 2008.
- [36] N. J. Mitra, A. Nguyen and L. Guibas, "Estimating Surface Normals in Noisy Point Cloud Data," International Journal of Computational Geometry and Applications, vol. 14, pp. 261–276, 2004.
- [37] Q.-Y. Zhou and U. Neumann, "Fast and Extensible Building Modeling from Airborne LiDAR Data," in Proceedings of the 16th ACM SIGSPATIAL international conference on Advances in geographic information systems, Irvine, 2008.
- [38] K.-H. Bae and D. D. Lichti, "A method for automated registration of unorganised point clouds," ISPRS Journal of Photogrammetry and Remote Sensing, vol. 63, no. 1, pp. 36-54, 2008.
- [39] Y. Fujita, Y. Hoshino, S. Ogata and I. Kobayashi, "Attribute Assignment to Point Cloud Data and Its Usage," Global Journal of Computer Science and Technology, vol. 15, no. 2, pp. 2-B, 2015.
- [40] Y. Fujita, I. Kobayashi, Y. Hoshino and W. Chanseawrassamee, "Development of Attribute-Assign-Editor for Road Surface Point Cloud Data," *IACSIT International Journal of Engineering and Technology*, vol. 8, no. 3, pp. 170-176, 2016.
- [41] E. Lin, Y. Ninsalam and M. F. Prescott, "Augmenting XYZRGB: Design and the Reality Captured Landscape," Kerb Journal of Landscape Architecture, vol. 23, no. -, pp. 112-113, 2015.
- [42] S. Ochmann, R. Vock, R. Wessel and R. Klein, "Automatic Reconstruction of Parametric Building Models from indoor point clouds," *Computers & Graphics*, vol. 54, pp. 94-103, 2016.
- [43] ArcMap, "What is lidar intensity data?," 2016. [Online]. Available: http://desktop.arcgis.com/en/ arcmap/10.3/manage-data/las-dataset/what-is-intensity-data-.htm. [Accessed 7 12 2018].
- [44] T. Voegtle, I. R. Schwab and T. Landes, "Influences of different materials on the measurements of a terrestrial laser scanner (tls)," *The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, vol. 37, no. 5, p. 1061–1066, 2008.
- [45] C. Suchocki and W. Błaszczak-B, ak, "Down-Sampling of Point Clouds for the Technical Diagnostics of Buildings and Structures," *Geosciences*, vol. 9, no. 2, pp. 1-14, 2019.

- [46] J. Armesto-González, B. Riveiro-Rodríguez, D. González-Aguilera and M. T. Rivas-Brea, "Terrestrial laser scanning intensity data applied to damage detection for historical buildings," *Journal of archaeological science*, vol. 37, no. 12, pp. 3037-3047, 2010.
- [47] S. S. Soudarissanane, The Geometry of Terrestrial Laser Scanning, Delft: PhD-thesis, Faculty of Civil Engineering and Geoscience, TU Delft, 2016.
- [48] A.-B. Mostafa and M. Ebrahim, "3D Laser Scanners' Techniques Overview," International Journal of Science and Research (IJSR), vol. 4, no. 10, pp. 323-331, 2015.
- [49] S. Crutchley, "Using LiDAR in Archaeological Contexts: The English Heritage Experience and Lessons Learned," in *Laser Scanning for the environmental sciences*, West Sussex, Wiley-Blackwell, 2009, pp. 180-200.
- [50] S. Sasidharan, "A Normalization scheme for Terrestrial LiDAR Intensity Data by Range and Incidence Angle," International Journal of Emerging Technology and Advanced Engineering, vol. 6, no. 5, pp. 322-328, 2016.
- [51] R. Blaskow and D. Schneider, "Analysis and Correction of the Dependency between Laser Scanner Intensity Values and Range," in *The International Archives of the Photogrammetry, ISPRS Technical Commision V Symposium*, Riva del Garda, 2014.
- [52] K. Tan and X. Cheng, "Correction of Incidence Angle and Distance Effects on TLS Intensity Data Based on Reference Targets," *Remote Sensing*, vol. 8, no. 3, p. 251, 2016.
- [53] S. Kaasalainen, A. Jaakkola, M. Kaasalainen, A. Krooks and A. Kukko, "Analysis of Incidence Angle and Distance Effects on Terrestrial Laser Scanner Intensity: Search for Correction Methods," *Remote Sensing*, vol. 3, no. 10, pp. 2207-2221, 2011.
- [54] D. Constantino and M. G. Angelini, "Qualitative and Quantitative Evaluation of the Luminance of Laser Scanner Radiation for the Classification of Materials," in *Int. Arch. Photogramm. Remote Sens. Spatial Inf. Sci.*, XL-5/ W2, Strasbourg, 2013.
- [55] D. Lichti and B. Harvey, "The Effects of Reflecting Surface Material Properties on TIme-of-Flight Laser Scanner Measurements," in *Symposium on Geospatial Theory, Processing and Applications*, Ottawa, 2002.
- [56] Y. Reshetyuk, "Investigation of the Influence of Surface Reflectance on the Measurements with the Terrestrial Laser Scanner Leica HDS 3000," *Zeitschrift fur Geodasie, Geoinformation und Landmanagement*, vol. 131, no. 2, pp. 96–103, 2006.
- [57] M. U. Hassan, A. Akcamete-Gungor and C. Meral, "Investigation of Terrestrial Laser Scanning Reflectance Intensity and RGB Distributions to Assist Construction Material Identification," in *the Joint Conference on Computing in Construction (JC3)*, Heraklion, 2017.
- [58] T. Voegtle, I. R. Schwab and T. Landes, "Influences of Different Materials on the Measurement of a Terrestrial Laser Scanner (TLS)," in *Proceedings of the XXI ISPRS Congress, The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, Beijing, 2008.
- [59] X. Li and Y. Liang, "Surface characteristics modeling and performance evaluation of urban building materials using LIDAR data," *Applied Optics*, vol. 54, no. 15, pp. 4750-4759, 2015.
- [60] B. B. Worstell, S. K. Poppenga, G. A. Evans and S. A. Prince, "Lidar Point Density Analysis-Implications for Identifying Water Bodies," Rolla Publishing Service Center, Sioux Falls, 2014.
- [61] G. Kigle-Boeckler, "Measurement of Gloss and Reflection Properties of Surfaces," *Metal Finishing*, vol. 93, no. 5, pp. 28-31, 1965.
- [62] N. Alam, "SlideShare," 18 6 2014. [Online]. Available: https://www.slideshare.net/NajmaAlam/thermalproperties-of-materials. [Accessed 15 11 2017].
- [63] E. Dobos, "Albedo," in Encyclopedia of Soild Science, Boca Raton, CRC Press, 2006, pp. 64-66.
- [64] J. Coackley, "Reflectance and Albedo, surface," in *Encyclopedia of the Atmosphere*, Corvallis, Oregon State University, 2003, pp. 1914-1923.
- [65] M. Councils, "Whiter than White," Creative Materials Consultant Limited, London, 2012.
- [66] H. Taha, D. Sailor and H. Akbari, "High-Albedo Materials for Reducing Building Cooling Energy Use," California Institute for Energy Efficiency (CIEE), Berkeley, 1992.
- [67] M. Ashby, J. Fernandez and A. Gray, "Granta Education Hub," 22 8 2008. [Online]. Available: https:// teachingresources.grantadesign.com/Edition/Architecture/PAPARCEN06. [Accessed 14 5 2018].
- [68] NTC-TTP, "New Technologies Research Centre," 2000. [Online]. Available: https://ttp.zcu.cz/en/ laboratories/optical-properties/optical-properties/emissivity. [Accessed 1 5 2018].
- [69] Optotherm, "Optotherm Thermal Imaging," 2018. [Online]. Available: https://www.optotherm.com/emissexamples.htm. [Accessed 5 5 2018].

- [70] A. Marsh, "Performative Design," 2010. [Online]. Available: http://performativedesign.com/definition/ light-reflectivity/. [Accessed 5 5 2018].
- [71] ElumTools, "Elumtools," 2012. [Online]. Available: http://www.elumtools.com/docs/2012/Content/ Using%20ElumTools/Color%20Reflectance%20or%20Transmittance.htm. [Accessed 6 5 2018].
- [72] D. Nagy, "Generative Design," 27 2 2017. [Online]. Available: https://medium.com/generative-design/solaranalysis-in-grasshopper-5dae76c9b6cb. [Accessed 7 9 2018].
- [73] M. F. Alkadri, M. Turrin and S. Sariyildiz, "The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts," *Advances in Computational Design*, vol. 3, no. 4, pp. 319-338, 2018.
- [74] M. Horvat and M.-C. Dubois, "Tools and Methods for Solar Design–An Overview of IEA SHC Task 41, Subtask B," in 1st International Conference on Solar Heating and Coolingfor Buildings and Industry (SHC 2012), San Fransisco, 2012.
- [75] N. Lukač, D. Žlaus, S. Seme, B. Žalik and G. Štumberger, "Rating of roofs' surfaces regarding their solar potential and suitability for PV systems, based on LiDAR data," *Applied Energy*, vol. 102, no. -, pp. 803-812, 2012.
- [76] A. Jochem, V. Wichmann and B. Höfle, "Large Areas Point Cloud Based Solar Radiation Modeling," Hamburger Beiträge zur Physischen Geographie und Landschaftsökologie, pp. 1–9, 2010.
- [77] J. Liang, J. Gong, W. Li and A. N. Ibrahim, "A visualization-oriented 3D method for efficient computation of urban solar radiation based on 3D-2D surface mapping," *International Journal of Geographical Information Science*, vol. 28, no. 4, pp. 780-798, 2014.
- [78] A. Jochem, B. Höfle and M. Rutzinger, "Extraction of Vertical Walls from Mobile Laser Scanning Data for Solar Potential Assessment," *Remote Sensing*, vol. 3, no. 4, pp. 650-667, 2011.
- [79] A. Pavlovski, J. Fletcher, V. Kostylev and J. Crace, "Solar Architecture and Energy Engineering," *Journal of Green Building*, vol. 5, no. 2, pp. 32-40, 2010.
- [80] Y. Li, D. Ding, C. Liu and C. Wang, "A pixel-based approach to estimation of solar energy potential on building roofs," *Energy and Buildings*, vol. 129, no. -, pp. 563-567, 2016.
- [81] M. E. Charlton, S. J. Coveney and T. McCarthy, "Issues in Laser Scanning," in Laser Scanning for the Environmental Sciences, West Sussex, Wiley-Blackwell, 2009, pp. 35-48.
- [82] S. D. Bartolo and R. Salvini, "Multitemporal Terrestrial Laser Scanning for Marble Extraction Assessment in an Underground Quarry of the Apuan Alps (Italy)," *Sensors*, vol. 19, no. 3, pp. 1-10, 2019.
- [83] M. N. Favorskaya and L. C. Jain, Handbook on Advances in Remote Sensing and Geographic Information Systems, Cham: Springer International Publishing AG, 2017.
- [84] D. Girardeau-Montaut, "Cloud-Compare," 2015. [Online]. Available: http://www.cloudcompare.org/doc/ qCC/CloudCompare%20v2.6.1%20-%20User%20manual.pdf. [Accessed 5 2 2018].
- [85] A. Goyal, "Lmn architects," 29 10 2014. [Online]. Available: https://lmnarchitects.com/tech-studio/bim/ revit-daylighting-tools-2-elumtools/. [Accessed 1 4 2018].
- [86] M. S. Roudsari and P. Michelle, "Ladybug: a parametric environmental plugin for grasshopper to help designers create an environmentally-conscious design," Lyon, 2013.
- [87] S. V. Szokolay, Introduction to architectural science: The Basis of Sustainable Design, Oxford: Elsevier/ Architectural Press, 2008.
- [88] A. Boulch and R. Marlet, "Deep Learning for Robust Normal Estimation in Unstructured Point Clouds," in *Eurographics Symposium on Geometry Processing*, Berlin, 2016.
- [89] A. Z. Ramírez and C. B. Muñoz, "Albedo Effect and Energy Efficiency of Cities," in Sustainable Development -Energy, Engineering and Technologies - Manufacturing and Environment, RIjeka, Intech, 2012, pp. 1-18.
- [90] R. W. Fleming, "Visual perception of materials and their properties," Vision Research, vol. 94, no. -, pp. 62-75, 2013.



5 Computational Models For Constructing Solar Geometry Based On Attribute Point Cloud Information

This first part of this chapter (Part A) has been published as: Alkadri, M. F., De Luca, F., Turrin, M. and Sariyildiz, S., 2020. An integrated approach to subtractive solar envelopes based on attribute information from point cloud data. *Renewable and Sustainable Energy Reviews*, 123, pp. 109742. Meanwhile, the second part of this chapter (Part B) has been published as: Alkadri, M. F., De Luca, F., Turrin, M. and Sariyildiz, S., 2020. A computational workflow for generating a voxel-based design approach based on subtractive shading envelopes and attribute information of point cloud data. *Remote sensing*, 12(16), pp. 2561. The layout has been adjusted to fit the template of this thesis.

As part of methodological framework of the thesis, this chapter presents the development of the computational models (i.e., SOLEN, SHADEN) for solar geometry based on the subtractive design principle of solar envelopes and attribute point cloud information. In chapter 4, attributes information stored in point cloud data (i.e., XYZ, RGB, I) was confirmed to be feasible to analyze the performance of the existing context through several functions such as computing material properties, developing material database, and conducting solar simulations. These functions simultaneously make a relevant contribution in dealing with missing features (i.e., materials, surface morphology) in the preliminary simulation of solar envelopes in chapter 3. Furthermore, this chapter utilizes environmental simulation features generated from point cloud data to develop subtractive solar envelopes and subtractive shading envelopes referring to SOLEN model in chapter 5-Part A and SHADEN model in chapter 5-Part B, respectively.

Each model contains a series of computational workflows that are developed based on their design objectives, contextual settings, and climatic parameters. In this regard, the SOLEN model aims to generate solar envelopes based on a specific amount of direct sunlight obtained from ray tracing analysis between the new building's 3D polyhedra and the existing site's point cloud dataset. Radiometric point cloud information is used to conduct insolation analysis and to detect material behavior of the existing environment as part of the environmental performance assessment of the solar envelope's final geometry. The SOLEN model is particularly applied to temperate climates with a selected case study in Delft, the Netherlands. Meanwhile, the SHADEN model is developed to generate an appropriate building mass based on the shading performance criteria. In this part, radiometric point cloud information is used to investigate material properties and sun visibility of the given context as part of form generation process to establish the final geometry of shading envelopes. As a solar radiation-reduction model, the concept of self-shading envelopes is also integrated into the workflow of SHADEN model based on the solar protection plane. It aims to create a self-building protection from overheating during critical periods. Thus, the new building can receive solar protection both from the surrounding context and the building itself. This model is specifically applied to tropical climates with a case study in Bandung, Indonesia. In design practice, these models may help architects in making comprehensive design decisions during the early stage of the design process based on actual contextual datasets.



PART A An integrated approach to subtractive solar envelopes based on attribute information from point cloud data

- As a passive design strategy, solar envelopes play a significant role in determining ABSTRACT building mass based on desirable sun access during a predefined period. Nowadays, advancements in the area of computational tools permit designers to develop new methods for establishing solar envelopes. However, current approaches lack an understanding of the existing environment's site characteristics, especially when dealing with geometrical information about the surrounding context. Consequently, this aspect affects the contextual analysis process during the generation of solar envelopes because of insufficient information for the relevant input of simulation modelling. With the support of geometric and radiometric properties stored in point cloud data, such as position (XYZ), color (RGB), and reflection intensity (I), this study has proposed novel subtractive solar envelopes that specifically consider the surface properties of the existing environment. Through a subtractive mechanism, the proposed method caters to several computational frameworks such as dataset pre-processing that aims to correct erroneous measurement during scanning. In alignment with that, the proposed building's visible sun vectors, optimal normal values, and 3D polyhedra are generated for the hit-or-miss analysis of subtractive solar envelopes. Furthermore, environmental assessments consisting of insolation and glare analysis are performed on the solar envelopes' final geometry. These performance assessments aim to investigate the potential and impact of the generated solar envelopes as it pertains to the existing buildings. Ultimately, this study supports architects not only in producing a new generation of subtractive solar envelopes based on real contextual settings but also in comprehensively understanding the microclimate condition of design context.
- **KEYWORDS** solar envelopes; point cloud data; solar access: passive design strategy; material properties; computational design method

5.1 Introduction

Interrelations between new buildings and existing context play a significant role in achieving appropriate solar access in the built environment [1]. It is necessary to understand these relations during the early design phase in order to maximize proper daylight within the design process and address microclimatic conditions for a sustainable approach. The significance of this approach has been mentioned explicitly in The United Nations Framework Convention on Climate Change, specifically in Article 4, which addresses the improvement of environmental quality as a response to climate change [2]. In the built environment, it is crucial to avert any potential unforeseeable failures once a new building has been built. These failures may very well consist of unexpected microclimatic impacts created by a building and imposed on the local context (and vice versa). For example, the building at 20 Fenchurch Street in London, EC3M 8AO, United Kingdom (2014) has reflected an extreme amount of sun glare into its surroundings because of its glazed, curved south facade. Multiple reports of severe effects have confirmed this issue; such reports include incidents such as a melted car, a heatwave around the building that exceeded 70 degrees Celsius, and the high cost of finally fixing the design flaw of the façade [3]. On the other hand, a new UCL student housing building, which is located at 465 Caledonian Road, North London, N7 9GU, United Kingdom, perfectly exemplifies a poor relationship between a new building and the existing context. As a result of the student housing building's poor suitability to its environment, over 175 rooms receive an insufficient amount of direct sun access because the building's windows are obstructed by a historic 19th century facade. These cases clearly show a lack of design integration between new buildings and existing contexts. Given that architects have a responsibility to ensure a reciprocal relationship, it is necessary to start at an appropriate point during the preliminary design process, especially when dealing with solar access performance.

The concept of solar envelopes presents a contextual design approach that has great relevance for addressing the aforementioned issues. Designers can use solar envelopes to define solar access between a proposed building and its surrounding context. Initially introduced by Knowles [4] [5], this concept consists of an imaginary mass that allows adjacent buildings the required amount of direct sun access. Since then, several variations of solar envelopes have been proposed [6] [7]. However, the existing method has several gaps in contextual analysis: for example, cut-off times that merely focus on a fixed period, resulting in surrounding facades hardly receiving the same quality of direct sun access due to different site orientations and obstructions. In addition, solar envelopes' geometric properties rely heavily on the

Z-axis of neighboring facades' shadow fences, resulting in the limiting of the design configuration of solar envelopes during the simulation process. Subtractive solar envelopes were developed in response to these issues [8]; these are constructed on the basis of specific quantities of direct sunlight hours as obtained from the surrounding windows of existing building's facades. The subtractive approach also allows for the generation of a larger volumetric size of solar envelopes, which is useful for high-rise buildings in fragmented urban environments [9].

Nevertheless, the integration of subtractive mechanisms into design practices needs to comply with several critical aspects such as sun visibility and ray tracing analysis. *First*, the selection of visible sun data aims to identify the number of sunlight hours that are being blocked (or not) by adjacent buildings. The issue that matters most is that the resulting areas that are identified as having blocked sun are only dependent on the building-oriented context. Consequently, these areas are assumed to be located on a clear plot, which is due to an absence of characteristics and relevant site properties (e.g. vegetation) [10] [11]. *Second*, the use of centroid points on surrounding windows during ray tracing analysis primarily counts on a sample of surrounding windows, which is limited to representing the area of neighboring facades. Indeed, these issues may affect environmental performance calculations during solar envelope simulation, especially in relation to solar radiation analysis between planned and existing buildings.

In order to compensate for missing features within the current workflow of subtractive solar envelopes, this study has employed the technological advancement of 3D laser scanning that includes point cloud data. Its potential application may support several contributions, which are as follows:

- The geometric information offered by point clouds may fill the gaps in sun visibility analysis since it can significantly cover buildings and their surrounding objects. This section specifically provides information about the surfaces that are the most and the least exposed to the sun, and the existing facades. Accordingly, the resulting areas with blocked sun (indicated as the surfaces that are the least exposed to the sun) can be eliminated because those areas are shaded by the surroundings (e.g. buildings and vegetation).
- Ray tracing simulation can be performed on each point that is found in the point cloud dataset, which can be used as an alternative to a limited sample of the surrounding windows. This allows for the identification of any obstructed areas within the 3D polyhedra (a proposed building). Voxels that consist of 0 obstructions can be transformed into solar envelopes while voxels with values above 0 need to be disregarded due to an obstruction of surrounding buildings' direct sun access.

Radiometric point cloud information (RGB color and reflection intensity) can be utilized to detect material behaviors in existing environments. This is specifically done by investigating both optical (reflectivity and translucency) and thermal properties (albedo and emissivity) calculated from a point cloud of the contextual dataset. Thus, material performance analysis between a planned and an existing building can be performed. This is achieved by projecting the ray tracing of the surrounding buildings to the final geometry of solar envelopes.

This study has primarily focused on the development of the computational workflow of subtractive solar envelopes based on attribute information from point cloud data. Two aspects of the workflow have been presented in previous papers. First, a previous paper presented the basic mechanism of subtractive solar envelopes [9]. This work focused on generating and subtracting a proposed buildings' 3D matrix based on a sunlight hours analysis that came from surrounding buildings' windows. This idea has stimulated the further exploration of methods for establishing computational solar envelopes based on point cloud data. Second, another previous paper has presented on the correction of the preliminary dataset and the calculation of material properties based on attribute information stored in a point cloud [12]. The dataset correction contributes to the extraction of the dataset's optimal normal values while material properties become an input parameter for performing glare analysis on the final geometry of solar envelopes. This paper will present a comprehensive approach for establishing new solar envelopes by making use of geometric and radiometric information contained in point cloud data with an integrated environmental performance analysis based on glare and solar radiation.

The present study has a six-part structure. Section 5.1 will give a general overview of the study. Section 5.2 will present the theoretical background, which consists of the basic principle of solar envelopes, the potential application of point clouds, and an existing method for creating subtractive solar envelopes. Section 5.3 will describe the computational workflow of new subtractive solar envelopes. This will be followed by the description of a case study's implementation in Section 5.4; this description will range from dataset collection to the details of the simulation process. Section 5.5 will present the simulation results and an environmental analysis that investigated the performance of glare and solar radiation. Finally, Section 5.6 will offer concluding remarks and suggestions for future research. This study has proposed a new generation of solar envelopes by making use of the subtractive mechanism with specific consideration to the existing environment's surface properties. In order to elucidate this, three major subjects will be comprehensively discussed: the basic concept behind state-of-the-art solar envelope methods, the potential implementation of point cloud data with geometric and radiometric properties, and the design procedures of the existing subtractive mechanism.

5.2.1 Basic principle of solar envelopes

In the distant past, there was some form of awareness of the performance of solar access as part of urban fabrics development. This approach can be recognized as part of the ancient world's concept, and it was represented within several cultures such as the Hanging Gardens of Babylon, which were constructed during the Babylonian period (605-562 BCE) [13] [14], the El-Lahun village of Egypt (1857-1700 BC) with its checkerboard urban layout that rigidly implemented narrow north-south streets [15]; and finally, the classical Greek cities of the 4th century BC, which developed solar-oriented homes [16]. Examples such as these not only laid the foundation for the principle of solar access but also contributed to the generation of the solar envelopes that were introduced by Knowles [4]. In a further development, Knowles [5] translated the concept of solar envelopes with considerations of space-time constraints before putting any design parameters in place. According to this principle, solar envelope's parameters can be categorized as both geographic and climatic [6]. Geographic parameters refer to aspects that relate to the spatial relationship between a land parcel and its surrounding context. These can be physically defined by terms such as longitude, latitude, orientation, courtyard, surrounding facades, sidewalks, building height, floor area ratio (FAR), setback, shadow fences, and streets. In comparison, climatic parameters constitute of non-physical characteristics that are related to the time construction of geometric models; these can be identified by terms such as solar angle, cut-off times, dry bulb temperature, sun path, solar azimuth, solar altitude, and sun access hours. In principle, geographic and climatic parameters correspond inversely to the solar envelopes' volumetric size; that is, the greater the time interval in solar envelope construction, the less solar envelope space is produced.



FIG. 5.1 A design mechanism for establishing solar envelopes, generated from A. Daily time limits and B. Annual time limits. These diagrams are elaborated from a book of Sun, Rhythm, and Form, authored by Knowles [5]. It specifically demonstrates the time setting before 9 am, and after 3 pm at 34° north (Los Angeles, USA).

As illustrated in Figure 5.1, daily and annual time settings demonstrate different solar envelope geometries. In a full day setting (Figure 1-A), the morning sun governs the envelope boundary at the western limits while the afternoon sun establishes the envelope shape at the eastern limit. This setting yields a relatively symmetrical shape that is divided between east and west. This is because of the similarity of their solar altitude angles between 9 a.m. and 3 p.m. On the other hand, the generated envelope for the annual time limit shows moderated steepness toward the north-facing surfaces. This is because the sun's altitude angle is larger in summer when striking the north-facing surfaces.

Furthermore, advancements in the area of computational tools inevitably affect both the workflow of and the method for constructing solar envelopes. Methods of establishing solar envelopes can be sub-divided into four categories that are based on both design parameters and computational platforms [6], and are as follows:

- Descriptive geometry—this method involves an intersection between the geometric lines of solar altitudes and the boundaries of the plot according to a predefined period (cut-off-times). The cross-section between these two lines generates a final envelope that fulfils the determined criteria of solar access [17] [18] [19] [20] [21] [22] [23] [24] [25] [26] [27] [28].
- Solar obstruction angle—the obstruction mechanism is derived from reference lines within a set of profile angles [29]. It transforms the angle of solar altitude to the solar azimuth that is perpendicularly related to the building façade being analyzed during a specific time of the day. As such, the reference lines also include the horizontal property line in the north and the vertical axis of the building within Cartesian space [30] [31] [32] [33] [34] [35] [36] [37] [38] [39] [40] [41] [42].
- Constructive solid geometry—this method primarily consists of site extrusion on the basis of solar vector settings. This setting contains rules that will help generate the initial shapes that were extruded from the plot while also taking into account the solar access constraints of the building and its surroundings. Furthermore, the intersection result of these initial shapes generates the final geometry of solar envelopes [43] [44] [7] [45] [46] [47] [48] [9].
- Digital elevation modelling (DEM) —on the basis of image processing techniques, this method quantifies a solar exposure map in order to generate solar volumes [49] [50]. This approach simultaneously introduces the concepts of iso-solar rights and iso-solar collections, both of which can be used to implement energy-oriented shapes on urban surfaces.

These approaches have successfully showcased a progressive trend of implementing solar envelopes as primary components within the architectural design process. During the generation of solar envelopes, these approaches can be categorized into two parts according to the contextual settings. In this case, 'contextual setting' refers to several parameters such as land parcel and surrounding site properties (e.g. vegetation, surrounding buildings, and open space) that might be relevant for generating solar envelopes. Some references properly consider parameters within the local context so that the interdependent relationship between the planned building and surrounding environments can be negotiated [17] [18] [19] [20] [21] [22] [23] [24] [30] [33] [34] [35] [38] [48] [9] [49] [50]. Meanwhile, others only

consider the given plot so that the implementation of solar envelopes primarily focuses on new development areas [25] [26] [27] [28] [31] [32] [36] [37] [39] [40] [41] [43] [43] [44] [7] [45] [46] [47]. However, the contextual setting of each method needs to be taken into account in terms of design concept and the complexity of the projects due to design parameter variation. Interestingly, the concept of solar envelopes has also been used in the recent projects of the Dutch architectural and urban design firm MVRDV: for example, the P15 Ravel Plot, which is located in the Zuidas district, 1082 LC, Amsterdam [51]. This project addresses the idea of solar-oriented design with the integration of a three-dimensional landscape of terraces and greenery. By studying the optimal line of sight and environmental performance, proposed buildings can have open southern exposures with plentiful sunlight.

Furthermore, the rapid development of 3D laser scanning technology has spanned multiple disciplines, especially in the domain of design and engineering. Point cloud data has also become more accessible than ever before, especially for collecting and processing complex information within an existing contextual environment. While this technology continually improves, a feature of reality-based datasets may provide useful support for the proposed subtractive method of solar envelopes.

5.2.2 Point cloud data

As a product of laser scanning technology, a point cloud consists of a threedimensional location with supplementary metadata that corresponds to each measurement record [52]. Similarly, Otepka et al. [53] described each point (P_i) by defining the three coordinates $(x_i, y_i, z_i)^T \pounds R^3$ and complementing these with auxiliary attributes $(a_{j,i})$ where $j = 1, ...m_i$ and i = the number of point attributes. If Pi is a vector $(x_i, y_i, z_i, a_{ui,...,} a_{m,i})^T$, then the first three components constitute a fixed meaning, like point coordinates, and the other components correspond to a different value for different point entities in the cloud. Another definition was offered by Weinmann [54] who illustrated the entity of a point within Euclidean space (R^D) where D = 2 represents two-dimensional properties such as an image location or a 2D key point while D = 3 indicates a three-dimensional element such as a 3D scanned object. As such, the value of a point can principally be assumed to have no real dimension, such as length, area, or volume because its dimension equates to a container of points. According to the above definitions, the data structure characteristics that result in a point cloud can be described further using permanent information. This can be done through the implementation of spatial XYZ coordinates [55] that are followed with supplementary attributes such as colour information (RGB) [56] and reflection intensity (I) ([57] [58]). To identify a specific function for each attribute, its metadata structure needs to be categorised into geometric and radiometric properties [59] [60] in the following ways:

- Geometric properties—this characteristic not only facilitates the configuration of the geometrical [61] and visual features of 2D and 3D models [55] but also allows simulation performance to be executed reliably [62]. An interesting example of this can be observed during the development of 3D reconstruction models within the fields of archaeology and heritage [63] [64]. In cases such as these, existing buildings have successfully been recreated within 3D models with colour details and representative material characteristics. However, this approach is limited to archaeological projects' contextual scale, which predominantly works with smaller objects, especially when compared to architectural scales. In addition, the modelling procedure from point cloud data to a mesh model inevitably poses a high cost in terms of computational issues. This is a critical matter, especially when dealing with the simulation analysis of complex projects in architectural design practice. For example, a previous study implemented an approach that simulated a descriptive solar envelope method [6]. This method was based on a 3D mesh model created with airborne laser scanning (ALS) datasets. Although this approach has successfully captured more detailed information regarding the site properties of an existing context, it has, unfortunately, presented a major concern for contextual modelling due to the solid modelling process [65] [66].
- Radiometric properties—this characteristic refers to the attribute information that is attached to each point within a dataset (e.g. RGB colour and reflection intensity). In addition to typical attributes, this type of point cloud also contains high order attributes or the so-called point feature. In this regard, some authors have defined these features into several categories. For example, Richter [67] developed six identification categories: colour, vegetation, terrain, surface normal, horizontality, and global/local height. In addition to these, Mallet [68] described 27 types of features that are separated into three groups: geometrical features [69], echo-based features, and full-waveform features [70] [71] [72]. Unfortunately, these features are only partially applicable to architectural design practices. This is because the amount of effort that needs to be invested in dataset processing makes the process unfeasible. In most cases, its practical application should deal with prior knowledge of technical dataset processing because it only caters to very particular tasks.

Understanding the main characteristics of a point cloud allows for the exploration of more relevant properties in architectural design. This can be observed through several implementations such as building construction [73] [74] [75], landscape modelling [76] [77], cultural heritage [78] [79] [80] and environmental engineering [81] [82] [83]. For example, colour information (RGB) can specifically be used not only to identify the inner surfaces of room volumes [84] but also to detect road signage by calculating of hue (H) and saturation (S) values within a dataset [56]. On the other hand, observed reflection intensity primarily deals with the surface and material properties of scanned objects [85] since it contains a return signal strength from laser pulse energy [86]. Thus, this method can be used to identify and map challenging characteristics such as damaged concrete inside tunnels, pavement lines, seafloor stripping, reflective detections signs, and geologic layers [60]. Furthermore, position information (XYZ) plays an essential role in synchronizing the coordinate locations of a dataset with various selected values made up of color and intensity information. This is done so that complex indicated areas can be extracted.

Due to the usefulness of attribute information datasets in point clouds, architects are able to generate environmental performance analyses for proposed architectural designs.

5.2.3 Subtractive solar envelopes

As a part of the solar form-finding concept that De Luca [9] proposed, subtractive solar envelopes consist of a volumetric three-dimensional matrix, namely 3D polyhedra; this is subtracted based on building facades' required sunlight from surrounding windows (Figure 5.2). 3D polyhedra are extruded from a selected land parcel based on a proposed building's criteria, which can encompass width, height, functional utilities, and even the number of floors. In principle, this mechanism has been proven as an effective strategy for dealing with daylight optimization, direct sunlight, and solar radiation since it capitalizes on voxel performance's space tessellation and form generation [87]. In alignment with this approach, Leide and Schlüter [88] developed an analytical and visual methodology through volumetric insolation analysis (VIA) and volumetric visibility analysis (VVA). Their method aimed to embrace the diversities that exist in the urban context combined with various performances such as radiation analysis, airflow, and visibility. Such approaches are also addressed in several references [89], [90], [91], [92], [93]. However, these references merely address the generative architectural form while neglecting the concept of solar envelopes.



FIG. 5.2 The design mechanism of existing subtractive solar envelopes. These diagrams are elaborated from the paper of solar form finding, authored by De Luca [9]

The subtractive simulation process primarily consists of three main stages. First, there is window selection per surrounding facades. At this stage, existing facades are selected and subdivided into window configurations according to predefined criteria, such as dimension and surface grid, which correspond to a particular amount of direct sun access. The selected windows are then represented by centroid points for establishing the sun vectors that are generated from the sunlight hours analysis. Second, there are evaluations of the sun vectors that come from the selected windows, affecting the surrounding facades, and the 3D plot, which uses a hit-or-miss procedure. This step involves Boolean intersection with a true or false statement. The statement will be true when sun vectors pass through the polyhedra and false when sun vectors do not hit the polyhedra. This procedure indicates that a true statement will subtract the voxels within the 3D polyhedra while a false statement will contribute to the generation of solar envelopes. Third, there is the obstruction index of the remaining polyhedra. The process of ray tracing sun vectors to polyhedra principally generates an obstruction range that is represented by colour-coded voxels. It indicates surrounding buildings' direct sun access obstruction level.

The study will now present a new method for generating solar envelopes by making use of the subtractive mechanism and point cloud attributes.

5.3 A proposed method for new subtractive solar envelopes

Based on the subtractive method, this study has developed a new generation method for solar envelopes that considers the potential use of point cloud attributes. A series of computational frameworks were formulated to reach this goal (Figure 5.3). In general, the proposed method consists of three phases: input, tasks, and output. First, input constitutes several parameters that construct solar envelopes' geometric shapes and affect performance. It is then divided into climatic and geographic properties. Climatic properties contain a specific location in addition to the indicated periods that are used to calculate solar vectors. Meanwhile, geographic properties include surrounding context parameters that have been obtained from both TLS datasets and the 3D plot of a proposed building (polyhedra generation). These are in addition to the predefined criteria of building height and number of floors. Second, tasks cater to a set of operations that are performed by calculating and analyzing a varied range of inputs from a raw dataset. These operations involve several off-the-shelf frameworks that have been adapted from existing studies and include factors such as dataset correction, calculation of material properties [12], and the computational procedures of a hit-or-miss analysis for 3D polyhedra [9]. *Third*, **output** consists of the geometric shapes of solar envelopes with an integrated performative analysis based on insolation (solar irradiance) and glare analysis. In this regard, glare refers to the discomfort of visual performance caused by the concentrated reflection of the sun from the surrounding building context to the proposed new building. These analyses aim to evaluate and integrate the environmental performance aspect into the final geometry of solar envelopes. Accordingly, a comprehensive contextual analysis can be performed as decision support for architects during the early design phases. In this case, a simulation of solar radiation would be run to reveal solar energy potential on solar envelopes' geometrical surfaces. This would help architects estimate and identify a new building's solar collector. A glare analysis could be conducted between existing buildings and proposed solar envelopes based on material properties with the goal of avoiding uncomfortable glare as a result of reflective materials on building facades [94].



FIG. 5.3 The computational workflow used in the proposed method of generating new subtractive solar envelopes

The proposed workflow employs several supporting tools that cater to the specific tasks in each step. For example, Faro Scene and Cloud Compare (CC) [95] are digital instruments that are used to prepare the raw dataset, which is necessary to create simulation input. Specifically, Faro Scene facilitates scanner-workstation data transfer. This also includes the registration process, color coding, and clipping box procedures. In comparison, CC is primarily used for dataset preparation. It involves outlier (unnecessary points within a cloud) removal, attribute order selection, point subsampling, dataset formatting, and the scalar field feature, which enables the identification and calculation of point cloud metadata information (e.g. the surface normal of unstructured point clouds and the reflection intensity). Matlab is used to support dataset correction, especially when calculating point clouds' optimal normal values. As a 3D modelling tool, Rhino coupled with Grasshopper is used to simulate solar envelopes. Ladybug (an environmental Grasshopper plugin) [96] is specifically used to calculate direct sunlight hours and perform ray tracing analyses. In general, the proposed workflow uses various dataset formats depending on the tasks and steps that need to be performed. For example, the E57 file is exploited to transfer the 3D point cloud from Faro Scene and CC so that the selected datasets can be interoperable in 3D modelling tools. In alignment with that, the ASCII file is also used to load the 3D point cloud with embedded attribute information. Breps (boundary representation) and TXT files are primarily employed in Rhino and Grasshopper to extract or input geometric properties and textual information, respectively.

The proposed workflow was tested by using a hypothetical case study. Each specific task was performed in correspondence with real datasets that were collected using a Faro laser scanner. Section 5.4 will provide procedural details of the computational workflow.

5.4 The implementation of the case study

5.4.1 Dataset collection

The method that this study has proposed was applied to a building context using on-site scanning datasets. At this particular site, the dataset collection needed to comply with several predetermined criteria. First, the acquired 3D point cloud originated from terrestrial laser scanning (TLS) datasets. This method aimed to supply accurate geometric properties [97] [98] in combination with an attractive, representative 3D model. This can be observed through the inclusion of both the texture and material visualization of real building contexts which enables view point switching between an intensity format and RGB color so that attribute value changes can be quickly and easily detected during the dataset preparation in a point cloud environment. In comparison between ALS and TLS datasets, it is crucial to note that TLS can cover a wide range of surrounding elements from isolated areas. These contextual elements include vegetation, temporal site objects, and other properties that are potentially relevant for solar envelope simulation. Second, attribute information should at least include the properties of position information (XYZ), color information (RGB), and reflection intensity (I). These attributes play a great role in dataset correction and calculating material properties for the environmental analysis of solar envelopes. This study specifically collected a sample of 3D point cloud data located in the Faculty of Architecture, TU Delft, Delft 2628 BL, the Netherlands. Geometrically, the selected dataset consisted of building facades, ground areas (pavements), vegetation, and surrounding buildings and objects. This dataset was harvested using a Faro Laser Scanner Focus S350. As a long-range scanning application, this tool not only supports extended distances and precise angular accuracy but also facilitates on-site data quality optimization. The tool's main technical specifications are as follows (see Table 5.1): [51].

TABLE 5.1 Tech	nical specifications of the tools [99]			
No	Parameters	Performance specification unit		
1.	Ranging unit			
	Range	0.6 – 350 m		
	Ranging noise	@ 10 m - 0.3 mm (90 % reflectivity)		
	Range error	± 1 mm		
	Angular accuracy	19 arcsec for vertical/horizontal angles		
	3D position accuracy	@ 10 m – 2 mm		
2.	Colour unit			
	Resolution	165 megapixels		
	High dynamic range (HDR)	Exposure bracketing 2x, 3x, 5x		
3.	Laser			
	Laser class	Laser class 1		
	Wavelength	1550nm		
	Beam divergence	0.3mrad(1/e)		
	Beam diameter at exit	2.12mm (1/e)		
4.	Extended operating temperature	-20° - 55° C		
5.	Humidity	Non-condensing		

The harvested dataset consisted of five single scans in order to capture the facades of the architecture faculty building and its surroundings (Figure 5.4). Each scan was taken within less than 6 minutes. The approximate scan size was 10240 x 4267 Pt with a point distance around 6.1 mm/10 m. This setting yielded a point density of around 22 million points for each scan.



FIG. 5.4 Dataset collection and the different views captured in relation to the scanning position

5.4.2 The computational design process

As illustrated in Figure 3, the proposed method is divided into four sequential design process, each of which consists of different tasks depending on the type and level of information received from the input section. An in-depth discussion of each step is given below.

5.4.2.1 Step 01

This step contains several parts that can be simultaneously performed. This is due in part to different preliminary inputs. First, the calculation of sun vectors requires a specific coordinate location and a selected period. In this case, the position of Delft is located at latitude 52.0116° N and longitude 4.3571° E. The time setting takes a sample of the required period on the 21st of each month, ranges from January to July, and starts between 9 a.m. and 6 p.m. This setting yields a total of 16 sun vectors that comprise the selection of the visible sun. Second, the pre-processing step caters to the two aspects of dataset preparation and scan correction for each scan. Essentially, this step enables architects to not only filter important information from the raw dataset but also identify parameters that considerably influence the values of metadata properties during scanning. These parameters can originate from environmental and meteorological factors, atmospheric pollution [100] [101], and scanning geometries [102]. To minimize the effects of these conditions, tasks are performed to prepare the selected dataset. For example, removing outliers removal and clipping boxes both determine and clean dataset boundaries (Figure 5.5). These outliers usually arise when scanning objects using a 3D laser scanner. This is due to the varying point densities and measurement errors that are experienced during scanning. In addition, sub-sampling procedures are applied to each scan in order to manage dataset point density by setting the distance between points to 5 cm. On average, this step produces 200.000 total points for each data scan.



FIG. 5.5 Dataset preparation A. Outlier removal B. Clipping box

After the selected dataset is prepared, the correction procedure is applied to each data scan. This specifically aims to correct the erroneous measurements within dataset attributes that have been generated due to noise interference in sensors, equipment sensitivity, laser wavelengths, and by targeting the surface geometry [103]. In this specific case, dataset correction involves the calculation of optimal normal values (Nx, Ny, Nz) and intensity correction (I_c). These normal values are calculated to find the average distribution of points that constitute a good projection as created by the laser beam during scanning. This is because the scattering condition of points corresponds differently at certain angles. This requires that the surface normal of datasets be identified first. Accomplishing this means applying the Hough Normal plugin [104] to each scan because of the unstructured cloud of points. In alignment with the previously stated requirements, the distribution of points is plotted by simulating different cosine values that have been extracted from various angles of incidence, which range from 10° to 90° (Figure 5.6-A). This step is performed using the following equation [103] and taking into account the original location of the laser scanner at (0,0,0):

$$i = \cos^{-1} \left(\frac{\overline{dn} \cdot \overline{dl}}{\left| \overline{dn} \right| \cdot \left| \overline{dl} \right|} \right)$$

(1)

where i = incident angle (α) $\overline{dn} =$ direction of surface normal $\overline{dl} =$ direction of the laser pulse

Afterwards, the distribution of point density is evaluated by calculating a standard deviation of cosine values within the dataset (Figure 5.6-B). Thus, the discovered scattered points are removed and the densest is truncated. As a result, each data scan shows a different number of truncated points depending on the density level of datasets within the plot of cosine values. These truncated datasets consist of optimal normal values that can be used for intensity correction and the calculation of sun visibility, which will be explained in the second step.



FIG. 5.6 The distribution of points on each data scan according to A. Angles of incidence and B. The standard deviation of different cos values

In principle, intensity correction is performed to calculate nearly 'true' values of intensity from the raw dataset (Figure 5.7). According to Kashani et al. [60], intensity information is influenced by several parameters such as the behavior of the target surface, acquisition geometries, equipment and environmental factors. However, some local constraints, such as climatic condition (moisture and temperature pressures) and tools' default settings during scanning, make the full correction of these parameters impossible to perform. Accordingly, this study performed intensity correction by only taking the relevant and significant parameter, such as the angle of incidence (α) to the TLS dataset. The following equation [60] was used for intensity correction:

$$I_c = I_{raw} \cdot \frac{1}{\cos \alpha}$$

where I_c = corrected intensity I_{raw} = original intensity α = angle of incidence



FIG. 5.7 Intensity correction on each data scan

(2)

As illustrated in Figure 5.7, the dataset transformation that resulted from intensity correction employed four color steps. The colors ranged from blue, which represents the lowest intensity value, to yellow, which represents the highest intensity value and refers to the most reflective surface. Dark and light green are intensity colors that straddle the threshold. The following results describe the distribution of the intensity datasets:

- In general, a major discrepancy between initial and corrected intensity can be observed primarily in roof areas and in some minor parts in building walls (see part of the scalar field intensity). The predominant intensity colour changes from dark green (in the column of raw intensity) to yellow (in the column of corrected intensity). This transformation is principally affected by spectral reflectance and oblique surfaces. In this case, the material on the building's roof consists of smoother surfaces (copper roof) than the material on the building's wall (old brick). Spectral reflectance on a smooth surface returns the same incident energy to the scanner due to the mechanism of specular reflection [105]. The same rule also applies to the oblique position of the building's roof in relation to the scanner's location. The backscattering intensity of a laser beam is greater on oblique surfaces than on direct surfaces [60].
- According to the intensity distribution values (from the raw intensity to the corrected one), all scans show a similar statistic regarding the increasing values of yellow (high reflective surfaces) while other color intensities are distributed dynamically depending on the data scan. For example, dark green and blue show a similar pattern regarding the increasing (Scan_001 and Scan_004) and decreasing values (Scan_002, Scan_003, and Scan_005). Meanwhile, light green has more scans with decreasing values (Scan_001, Scan_003, and Scan_004) than increasing values (Scan_002 and Scan_005). Identifying color intensity on these scans enables us to not only create a clear boundary between a different range of values but also more deeply understand the characteristics of surface material properties in the existing building.

Third, the polyhedra act as a proposed building's 3D envelope (Figure 5.8). The envelope is built according to the new building's relevant criteria such as width and height, functional utilities, and number of floors. Principally, each polyhedron represents a typical room in the real building. In this scenario, it consists of $3 \times 3 \times 3$ m with the total number of polyhedra being equivalent to 180 rooms. The polyhedra are then placed in the courtyard of the faculty building so that the spatial relationship with the surrounding buildings can be examined. A proposed building's function should support academic environments, thus including, for example, space for meeting rooms, a library, and staff offices.



FIG. 5.8 3D polyhedra of a proposed building

5.4.2.2 Step 02

In principle, the calculation of sun visibility aims to identify visible sun vectors that are not violated by existing buildings and surrounding properties. This requires that the sun vectors from the indicated period be multiplied by optimal normal values that have originated from truncated points in each data scan. In order to identify points that meet sun visibility criteria, the calculated vectors are filtered by considering only those values that are smaller than 90°. This is because sun vectors with angle values that are equal to and larger than 90° are not visible to the scanned point. Ultimately, the procedure results in a group of points within each data scan that meet the criteria of the visible sun vectors.

5.4.2.3 Step 03

This step calls for the performance of a hit-or-miss analysis that requires input from selected points in the second step and the 3D polyhedra in the first step (Figure 5.9). In this case, 16 sun vectors continuously evaluate the 3D polyhedra over 1,008 hours. This procedure is subsequently run on each data scan, resulting in a different number of ray intersections. With a Boolean operation, each polyhedron is automatically transformed into a voxel and assigned the true or false criteria. Voxels with 'false' values are indicated as obstruction indexes that need to be eliminated. These obstruction indexes can be viewed as green dots attached to the polyhedra (see Figure 5.9, specifically part of the obstructive points). Meanwhile, 'true' values are used to generate solar envelopes. This is predominantly illustrated in yellow boxes (see part of selected voxels for solar envelopes) with a different number of voxels for each data scan.



FIG. 5.9 Hit-or-miss analysis for each data scan

Based on Figure 5.9, some analysis elements can be further discussed as follows:

Scan_004 illustrates the highest number of voxels that ultimately meet the criteria of solar envelopes while Scan_001 and Scan_003 show the least. This trend demonstrates that the densest points are involved in ray tracing analysis, and the smallest number of voxels are produced. This can be observed on the intersection lines of Scan_001 and Scan_003; they nearly cover the polyhedra, especially when compared to the other scans. These intersection lines are principally affected by

the density of existing site properties and the number of truncated datasets that generated correction. On the other hand, it can be argued that increasing site properties (e.g. tress, adjacent buildings, and other elements) during simulation can significantly reduce the total voxels for solar envelopes. However, in this case, dataset correction plays an important part in filtering optimal normal values from the dataset, which means that not all points are available for use. Moreover, the step of sun visibility guarantees that only relevant points that meet the criteria of visible sun vectors may proceed to ray tracing analysis.

- The generated voxels for solar envelopes consist of color-coded values. Sequentially, these go from yellow to green, indicating the level of obstruction from low to high. As illustrated in Figure 9, most of the data scans show yellow and light green voxels, meaning that, for the indicated period, only some are dark green in Scan_004. It follows that the remaining voxels can be further synchronized with a proposed building's functional program.
- With regard to the geometrical configuration of generated voxels, each data scan shows a different voxel composition depending on the distance between the polyhedra and the surrounding facades. In this case, obstructive points that were generated from ray tracing affected most polyhedra facades that faced the scanner's position.

5.4.2.4 Step 04

As a part of environmental performance simulation, this step calculate the material properties of existing buildings as extracted from the point cloud data. A glare analysis between the final geometry of the solar envelopes and the existing buildings is also performed to identify voxels that might be affected by certain material properties. Given that radiometric information (RGB color and corrected intensity) is collected from the point cloud data, the material properties are calculated according to the thermal (albedo and emissivity) and optical parameters (reflectivity and translucency). As Figure 5.3 showed, the detailed procedure for the calculation of material properties has been extensively addressed in previous research [12]. Accordingly, the present study mainly focused on integrating materials workflow into the computational framework of solar envelopes.

Furthermore, the calculation of material properties was performed on the sample of data scans. In this case, Scan_003 was selected to illustrate the result of material distributions because its scanner position was located in the center of the selected site. This was done so that dataset coverage could be evenly captured on all sides

of the building (Figure 5.10). Setting the threshold value of material properties in a range of 0.0-1.0 allows for the density of points in each material property to be calculated. In addition, the resulting areas can be visually illustrated within the dataset.

Figure 5.10 shows that all material properties illustrate a similar trend regarding the point density of the smallest and largest values. For example, the smallest values with a point density of less than 1 % can be identified in several ranges within one material property, such as albedo and reflectance, with three property ranges, and emissivity and transmissivity with 4 and 2 property ranges, respectively. Within these ranges, only the threshold values of 0.8–0.9 and 0.9–1.0 illustrate an equivalent range in all material properties. In this case, the range of 0.9 - 1.0 shows that albedo and average reflectance have similar 0 values. This is also in alignment with the trend exhibited by the largest value point density. With a value of more than 20 %, albedo, reflectance, and emissivity constitute the same range at 0.5-0.6. consisting of 20.9 %, 21.83 %, and 36.36 %, respectively. With regard to the surface distribution of these properties, the range of 0.5–0.6 predominantly covers ground areas around the scanner position with additional areas on the building roof for the albedo and reflectance property, and the building wall the emissivity. Meanwhile, for transmissivity, the largest value of the property range is found within the ranges of 0.2–0.3 and 0.3–0.4. Identified areas are primarily scattered around the building roof and the ground. In general, this surface material catalogue enables architects to identify particular areas within the dataset based on certain property thresholds so that the surface characteristics of the existing context, especially the ones that are related to the material performance of specific areas, can be better understood

Threshold values	Albedo	Emissivity	Average Reflectance	Transmissivity
0.0 - 0.1				
point density	934 (0.93 %)	23825 (23.7 %)	938 (0.93 %)	1122 (1.12 %)
0.1 - 0.2				
point density	5777 (5.74%)	720 (0.72 %)	5466 (5.43 %)	10828 (10.76 %)
0.2 - 0.3				
point density	10622 (10.56 %)	10534 (10.5 %)	10401 (10.34%)	25150 (25 %)
0.3 - 0.4				
point density	15779 (15.68 %)	6598 (6.56 %)	15264 (15.17%)	29164 (28.9 %)
0.4 - 0.5				
point density	22269 (22.13 %)	17294 (17.2 %)	20304 (20.18 %)	14545 (14.46 %)
0.5 - 0.6				
point density	21024 (20.9 %)	36579 (36.36%)	21969 (21.83 %)	10358(10.3%)
0.6 - 0.7		A CAL		A REAL
point density	21304 (21.18%)	5003 (4.97%)	21261 (21.13%)	6179 (6.14 %)
0.7 - 0.8				
point density	2766 (2.75%)	44 (0.044 %)	4731 (4.7%)	2771 (2.75%)
0.8 - 0.9				
point density	123 (0.12 %)	2 (0.002 %)	262 (0.26 %)	433 (0.43 %)
0.9 - 1.0				
				and the second se

FIG. 5.10 The surface distribution of material properties in Scan_003
The present work has articulated the research in three main parts. The first part described the geometric configuration of solar envelopes as a final output of the simulation result. This solar envelope was subsequently followed by an environmental performance assessment that the other two parts addressed. These parts aimed to investigate the potential and impact of the geometric solar envelopes as it pertains to both the building and its surroundings. This involved an insolation analysis on the final solar envelopes and a glare simulation based on material properties. A sequential in-depth discussion of these parts is presented below.

5.5.1 The final geometry of solar envelopes

Having created solar envelopes for each data scan, the resulting geometries can ultimately be combined into single envelopes within the completed datasets of the surrounding context (Figure 5.11). In total, it consists of 217 voxels due to overlapping envelopes from all data scans. However, after overlapping envelopes are removed, the number of final envelopes is reduced to 133 voxels. Based on Figure 5.11, some analytical conclusions can be drawn as follows:

- The final geometry of solar envelopes principally acts as a geometrical core that fulfils the criteria of solar envelopes. Accordingly, additional voxels that are generated beyond the volumetric size of initial envelopes can still be a part of solar envelopes as long as the indicated voxels fit the required number and plot of the 3D polyhedra (the proposed design). Therefore, architects can simultaneously consider the functional program and related activities that correspond to the intended voxels.
- In order to identify the distribution of overlapping voxels from all data scans, the combined solar envelopes were plotted with color-coded attributes. The final voxels were divided into three color ranges: dark, medium, and light red. *First*, voxels with a dark red color illustrate the highest level, consisting of at least more than two overlapping envelopes. This indicates that those voxels have successfully fulfilled the solar access criteria of several data scans. Accordingly, the selected voxels can be prioritized in order to fulfil the main activities of the building that require fulltime sun access during the determined period such as those buildings that contain meeting rooms. *Second*, the voxels with a medium red color consist of two overlapping envelopes. These voxels can be fulfilled with any functional programs that do not

require a full day of sun access criteria, such as staff or work rooms. *Third*, light red voxels might be relevant to the building's tertiary or supporting activities, such as open space, corridors, and service areas.

In general, voxel distributions in each color range are not only varied in terms of the quantity, but are also diverse in terms of geometric configuration. In this case, the light red category constitutes the largest volume of envelopes with 66 voxels followed by the medium and dark red categories with 52 and 15 voxels, respectively. This indicates that the geometric characteristics of each data scan may considerably affect the number of identified voxels within the combined solar envelopes. Moreover, an interesting pattern can be observed in the vertical geometry of stacked voxels that can be assumed as equal to building floor levels. The light red voxels nearly fulfil space on the first, second, and third floors while the voxel distribution of the medium one mainly concentrates on the fifth and sixth floors.



FIG. 5.11 The final geometry of solar envelopes

5.5.2 Insolation analysis for solar envelopes

In principle, insolation analysis performs an integrated environmental assessment on the final geometry of solar envelopes that focus on solar radiation. The objective is not only to identify and calculate the potential of solar energy that can be permeated by a geometric surface of solar envelopes but also to illustrate for architects how integrated environmental analysis can be achieved through this approach. Accordingly, architects can draw a sustainable impact for future scenarios regarding buildings and their surroundings.

Furthermore, the simulation of solar radiation employs a trigonometric principle by calculating cosine values between the surface normal of each voxel and solar vectors that are originated from climatic properties (Step 1 in Figure 5.3) [106]. In this case, only positive cosine values are considered because of the inability of surfaces to absorb negative amounts of energy when dealing with incident angles over 90°.

As shown in Figure 5.12, all scans illustrate a similar trend regarding the surfaces that have the most potential for collecting solar energy, which are north-facing facades. This is because the location of existing buildings creates a massive obstruction to direct sun access for the south-facing facades of solar envelopes. In addition, the site is characterized by a low solar elevation angle due to its location's high latitude. This condition makes the surfaces of solar envelopes inaccessible in terms of receiving direct sun access, especially when there is close proximity to surrounding buildings. Thus, north and west-facing façades are identified as potential surfaces for the placement of solar collectors or the implementation of passive heating strategies.

Ultimately, the resulting values of insolation analysis show that the entire surface of solar envelopes reaches around 844 kWh during a predefined period. For comparison, this amount of energy can be used to supply an average Netherlands household with electricity for around 2.89 months [107]. For a standard practice, considering the system efficiency about 15 to 20 %, the useable energy can be sufficient for 3 weeks. Therefore, this result should spur architects to internalize a good recommendation, enhance environmental awareness, and make better design decisions in future planning. Nevertheless, the proposed insolation analysis would benefit from further consideration with regard to the parameters that are involved in the calculation, such as diffuse and reflected radiation [108], air mass, temperature, and other environmental factors.



FIG. 5.12 The insolation analysis of the final solar envelopes

5.5.3 Glare analysis simulation

Before performing a glare analysis between the final geometry of solar envelopes and the identified material properties of the existing context, two selection criteria must be set, namely selected points based on sun visibility and normal vectors based on specular reflection. *First*, this criterion employs a similar mechanism as in the second step in Sub-section 5.4.2 (part of computational design process). In this regard, the threshold values of material properties only consider selected points for extraction from the sun visibility selection. As a result, the number of points registered in all

material properties decreased significantly due to incompatibility with the visible sun criteria (Figure 5.13). The largest discrepancy (before and after the sun visibility selection) illustrated a similar trend among the material properties, especially for albedo, emissivity, and average reflectance. These properties constituted the same value range of 0.5–0.6, similarly consisting of more than 17 thousand points, while transmissivity was identified at value ranges of 0.2–0.3 and 0.3–0.4. Regarding the surface distribution of the dataset, most of the identified values after the sun visibility selection were located on the building's roof with a significant decrease in the ground areas. This suggested that the sun vector of points located on the ground level predominantly consisted of zero and negative values, meaning that those points were automatically filtered by the criteria of visible sun.



FIG. 5.13 A comparison of material properties before and after the sun visibility selection

Second, sun vectors that were acquired from the selected points and threshold values of material properties were simulated by considering specular reflection. This was done by assuming that the angle between the normal vector and the incoming light was equal to the one between the normal vector and the reflected light. This requires that sun vectors around the point normal be rotated to 180°. Furthermore, the selected vectors and points were carried forward to the hit-or-miss analysis step with the final geometry of solar envelopes.

5.5.4 The geometric configuration of solar envelopes based on glare simulation

This section discusses the result of glare simulation between the proposed solar envelopes and the existing buildings on the basis of material properties. Since this simulation adopted a similar mechanism as the sun visibility criteria, its procedure was therefore limited with regard to the geometric intersection between the solar envelopes and the sun vectors reflected from the surroundings. The simulation result primarily identified affected geometries within the solar envelopes without considering the luminance level of reflected lights. Architects can use this approach to avert proposed building's microclimate issues by considering material performance during the conceptual design stage.

As shown in Figure 5.14, the geometric configuration of the identified voxels have a similar pattern for specific ranges of thresholds. For example, a zero voxel is predominantly identified at the range of 0.8–1.0 except for the transmissivity property, which refers to the range of 0.0–0.2. This value means that the reflected light from the surroundings fails to intersect with the geometry of the solar envelopes due to a different incident angle and point density (refers to Figure 5.13) within the performed threshold. In addition, the emissivity refers to the intensity values within an inverse range. Consequently, it affects the calculation of other properties. The indicated patterns are listed as follows:

- Emissivity consists of three zero voxels (in a range of 0.7–1.0) while others only consist of two zero voxels (albedo and average reflectance, similarly in a range of 0.8–1.0, with transmissivity in a range of 0.0–0.2).
- With regard to the distribution of the voxels, all properties contain four threshold ranges for registered voxels, amounting to less than ten, except for emissivity which consists of six ranges.
- Emissivity results show significant contrast in terms of the quantity of identified voxels in comparison with others. This can be observed in ranges of 0.0–0.1, 0.2– 0.3 and 0.6–0.7.

Threshold values	Albedo	Emissivity	Average Reflectance	Transmissivity
0.0 - 0.1				
total voxels	3	51	3	0
0.1 - 0.2		And A		
total voxels	19	1	23	0
0.2 - 0.3			Attendent of the second	A LA
total voxels	30	4	35	33
0.3 - 0.4				And the second s
total voxels	32	29	30	43
0.4 - 0.5		A REAL	A statement	A strength
total voxels	33	49	30	28
0.5 - 0.6		ALL HELLE	Attender.	And the second
total voxels	37	40	31	37
total voxels	37	40	31	37
total voxels 0.6 - 0.7	37 37	40	31	37
total voxels 0.6 - 0.7 total voxels	37	40	31	37
total voxels 0.6 - 0.7 total voxels 0.7 - 0.8	37 37 45 45	40 40 8 8	31 45	37
total voxels 0.6 - 0.7 total voxels 0.7 - 0.8 total voxels	37 37 45 45 3	40 40 8 8 0	31 45 45 2	37 25 28
total voxels 0.6 - 0.7 total voxels 0.7 - 0.8 total voxels 0.8 - 0.9	37 45 3 3 3	40 40 8 8 0 0	31 45 2 2	37 25 28 28
total voxels 0.6 - 0.7 total voxels 0.7 - 0.8 total voxels 0.8 - 0.9 total voxels	37 37 45 3 3 3 3 0	40 40 8 8 0 0 0	31 45 2 2 0	37 25 28 28 4
total voxels 0.6 - 0.7 total voxels 0.7 - 0.8 total voxels 0.8 - 0.9 total voxels 0.9 - 1.0	37 37 45 3 3 3 0 0	40 40 8 8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	31 45 42 2 0 0	37 25 28 28 4

FIG. 5.14 The geometric configuration of solar envelopes according to glare simulation analysis

5.6 Conclusion and Future Recommendations

This study has built on a novel method of generating subtractive solar envelopes that are based on the attribute information found in 3D point cloud data. The present workflow has confirmed the feasibility of integrating both geometric and radiometric properties of a point cloud, especially when supporting the contextual analysis of a proposed building. In particular, point cloud data acts as a basis for making an informed design context, therefore allowing environmental performance to be embedded in the workflow that is associated with the development of new solar envelopes. This approach ultimately allows architects to not only compensate for missing information during site analysis but also identify the context's microclimate conditions during the preliminary design stage. In general, this work warrants several remarks, which are listed as follows:

- With the overarching goal of developing new subtractive solar envelopes, attribute information in a point cloud enables the expansion of the functional properties that are available with 3D scanning technology. This is important in the architectural design stage, especially when dealing with environmental performance analysis, since it provides an alternative to focusing exclusively on 3D reconstruction and data visualization.
- Dataset correction is necessary not only to obtain a point cloud's corrected optimal normal values but also to minimize erroneousness in datasets during scanning. This will also help architects select relevant information to comprise the input that is needed to develop solar envelope simulations.
- Integrating aspects of material properties into the new workflow of subtractive solar envelopes will help contribute to the expansion of performance analysis in the relationship between new buildings and their local contexts. This also supports architects in more appropriately configuring any activities that are suitable, especially with regard to the solar and material performance of certain voxels and activities that need direct solar access within a certain period.

Acknowledging the limitations of this study, there are some elements that would benefit from further consideration. For example, the computational issues that were experienced during simulation were created by highly dense datasets, and this remains a major concern. Some of the study's aspects merely require very particular adjustments, such as the number of sun vectors or the point density of datasets. Moreover, synchronization between more existing urban properties and different functional utilities may affect solar envelopes' final geometry.

In further research, some potential aspects can be explored by investigating the relationship between the floor plan and an optimized layout design for both solar envelopes and energy efficiency. This will allow architects to define the layout configuration for each voxel within solar envelopes given that solar envelopes are based on the direct solar access that is available in a specific timeframe (per hour and per day). Moreover, the proposed method can be explored using different climatic contexts, such as tropical climates, as doing so can potentially expand possibilities for implementing designed solar envelopes within different urban settings.

References

- L. van Hove, G. Steeneveld, C. Jacobs, B. Heusinkveld, J. Elbers, E. Moors and A. Holtslag, "Exploring the Urban Heat Island Intensity of Dutch cities," Wageningen University and Alterra Report 2170, Wageningen, 2011.
- U. Nations, "United Nations Framework Convention on Climate Change," United Nations, Rio de Janeiro, 1992.
- [3] O. Wainwright, "The Walkie-Talkie skyscraper, and the City's burning passion for glass," The Guardian, 3 September 2013. [Online]. Available: https://www.theguardian.com/commentisfree/2013/ sep/03/walkie-talkie-skyscraper. [Accessed 7 July 2019].
- [4] R. L. Knowles, Energy and form: An ecological approach to urban growth, Cambridge: The MIT Press, 1974.
- [5] R. L. Knowles, Sun, rhytm and form, Massachusetts: The MIT Press, 1981.
- [6] M. F. Alkadri, M. Turrin and S. Sariyildiz, "The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts," *Advances in Computational Design*, vol. 3, no. 4, pp. 319-338, 2018.
- [7] T. N. Stasinopoulos, "A survey of solar envelope properties using solid modelling," *Journal of Green Building*, vol. 13, no. 1, pp. 3-30, 2018.
- [8] F. De Luca and T. Dogan, "A novel solar envelope method based on solar ordinances for urban planning," *Building simulation*, vol. 12, no. 5, pp. 817-834, 2019.
- [9] F. De Luca, "Solar form finding," in ACADIA, Cambridge, 2017.
- [10] Y. Arayici, A. Hamilton and P. Gamito, "Modelling 3D Scanned Data to Visualize and Analyse the Built Environment for Regeneration," *Surveying and Built Environment*, vol. 17, no. 2, pp. 7-28, 2006.
- [11] Y. Arayici, "Towards Building Information Modelling for Existing Structures," *Structural Survey*, vol. 26, no. 3, pp. 210-222, 2008.
- [12] M. F. Alkadri, M. Turrin and S. Sariyildiz, "A computational workflow to analyse material properties and solar radiation of existing contexts from attribute information of point cloud data," *Building and Environment*, vol. 155, no. -, pp. 268-282, 2019.
- [13] D. Stevenson, "A Proposal for the Irrigation of the Hanging Gardens of Babylon," IRAQ, vol. 54, no. -, pp. 35-55, 1992.
- [14] D. J. Wiseman, "Mesopotamian Gardens," Anatolian studies, vol. 33, no. -, pp. 137-144, 1983.
- [15] D. Mazzone, "The Dark Side of a Model Community: The 'Ghetto' of el-Lahun," *The Journal of Ancient Egyptian Architecture*, vol. 2, no. -, pp. 19-54, 2017.

- [16] K. Butti and J. Perlin, A Golden Threat 2500 Years of Solar Architecture and Technology, New York: Van Nostrand Reinhold Company, 1980.
- [17] I. G. Capeluto and E. Shaviv, "Modelling the design of urban fabric with solar rights considerations," in International Conference of IBPSA, Kyoto, 1997.
- [18] I. Capeluto, A. Yezioro, T. Bleiberg and E. Shaviv, "From computer odels to simple design tools: solar rights in the design of urban streets," in *Ninth international IBPSA conference*, Montreal, 2005.
- [19] G. I. Capeluto and B. Plotnikov, "A method for the generation of climate-based, context-dependent parametric solar envelopes," *Architectural science review*, vol. 60, no. 5, pp. 395-407, 2017.
- [20] K. Machacova, J. Keppl and L. Krajcovics, "The solar envelope method in education at the faculty of architecture STU Bratislava," in *Central Europe towards sustainable building*, Prague, 2013.
- [21] C. L. Martin and G. Keeffe, "The Biomimetic solar city: solar derived urban form using a forest-growth inspired methodology," in *The 24th Conference on passive and low energy architecture*, Singapore, 2007.
- [22] C. L. Martin, M. Pilling, C. Stott and V. Walsh, "The nectar project: solar development of post-industrial urban communities," in *The 27th conference on passive and low energy architecture*, Louvain-la-Neuve, 2011.
- [23] M. Dekay, "The implications of community gardening for land use and density," *Journal of architectural and planning research*, vol. 14, no. 2, pp. 126-149, 1997.
- [24] F. De Luca, "Solar envelope optimization method for complex urban environments," in CAADence in Architecture, Budapest, 2016.
- [25] G. Bruce, "High density, Low energy: achieving useful solar access for Dublin's multi-storey apartment developments," in *PLEA*, Dublin, 2008.
- [26] M. M. Saleh and K. S. Al-hagla, "Parametric urban comfort envelope: an approach toward a responsive sustainable urban morphology," *International journal of civil, environmental, structural, construction and architectural engineering*, vol. 6, no. 11, pp. 930-937, 2012.
- [27] D. Noble and K. Kensek, "Computer generated solar envelopes in architecture," *The Journal of architecture*, vol. 3, no. 2, pp. 117-127, 1998.
- [28] P. Camporeale, "Genetic algorithms applied to urban growth optimization," in *eCAADe conference*, Delft, 2013.
- [29] P. J. Littlefair, M. Santamouris, S. Alvarez, A. Dupagne and D. Hall, Environmental site layout planning: solar access, microclimate and passive colling in urban areas, London: Construction research communications Ltd, 2000.
- [30] I. G. Capeluto, A. Yezioro, T. Bleiberg and E. Shaviv, "solar rights in the design of urban spaces," in *The 23rd conference on passive and low energy architecture*, Geneva, 2006.
- [31] R. S. Brandao and M. P. Alucci, "Solar access in tropical cities: towards a multicriteria solr envelope," in The 22nd Conference on passive and low energy architecture, Beirut, 2005.
- [32] F. O. R. Pereira, C. A. N. Silva and B. Turkienikz, "A methodology for sunlight urban planning: a computer based solar and sky vault obstruction analysis," *Solar energy*, vol. 70, no. 3, pp. 217–226, 2001.
- [33] P. C. Grazziotin, F. O. R. Pereira, C. M. D. S. Freitas and B. Turkienicz, "Integration of Sunlight Access Control to Building Potential Simulator," in *The Ibero-American Symposium on Computer Graphics*, Guimaraes, 2002.
- [34] B. Turkienicz, B. B. Goncalves and P. Grazziotin, "CityZoom:A Visualization Tool for the assessment of plannning regulations," *International journal of architectural computing*, vol. 6, no. 1, pp. 79–95, 2008.
- [35] M. D. G. V. D. Amaral, "The application of solar architecture in the planning of the campus," in The 2005 World sustainable building cocnfrence, Tokyo, 2005.
- [36] B. Paramita and M. Koerniawan, "Solar envelope assessment in tropical region building case study: vertical settlement in Bandung, Indonesia," in *The 3rd International conference on sustainable future for human security SUSTAIN 2012*, Kyoto, 2013.
- [37] R. Emmanuel, "A hypothetical 'shadow umbrella' for thermal comfort enhancement in the equatorial urban outdoors," *Architectural science review*, vol. 36, no. 4, pp. 173-184, 1993.
- [38] G. Lobaccaro, F. Frontini, G. Masera and T. Poli, "SolarPW: A new solar design tool to exploit solar potential in existing urban areas," *Energy Procedia*, vol. 30, pp. 1173-1183, 2012.
- [39] R. Dekay, "Climatic urban design: configuring the urban fabric to support daylighting, passive cooling, and solar heating," *The sustainable city VII*, vol. 155, pp. 619-630, 2012.
- [40] M. DeKay, "Daylighting and urban form: an urban fabric of light," *Journal of architectural and planning research*, vol. 27, no. 1, pp. 35-56, 2010.
- [41] A. Okeil, "A holistic approach to energy efficient building forms," *Energy and buildings*, vol. 42, no. 9, pp. 1437-1444, 2010.

- [42] A. Okeil, "In Search for Energy Efficient Urban Forms: The Residential Solar Block," in BUilding for the future: The 16th CIB world building congress 2004, Rotterdam, 2004.
- [43] K. Raboudi and A. B. Saci, "A morphological generator of urban rules of solar control," in *The 29th conference* Sustainable architecture for a renewable future, Munich, 2013.
- [44] T. N. Stasinopoulos, "Solar envelope a construction method using AutoCAD 2000," 9 July 2001. [Online].
 Available: http://www.oikotekton.eu/solenvelope. [Accessed 25 October 2016].
- [45] K. Kensek and A. Henkhaus, "Solar access zoning + building information modeling," in Solar 2013, Baltimore, 2013.
- [46] A. Vartholomaios, "The residential solar block envelope: a method for enabling the development of compact urban blocks with high passive solar potential," *Energy and Buildings*, vol. 99, pp. 303-312, 2015.
- [47] J. F. Cotton, "Solid modeling as a tool for constructing solar envelopes," Automation and Construction, vol. 5, no. 3, pp. 185-192, 1996.
- [48] J. Niemasz, J. Sargent and C. F. Reinhart, "Solar zoning and energy in detached dwellings," in SimAUD, Boston, 2011.
- [49] E. Morello and C. Ratti, "Sunscapes: "solar envelopes' and the analysis of urban DEMs," Computers, Envrironment and Urban Systems, vol. 33, no. 1, pp. 26-34, 2009.
- [50] C. Ratti and E. Morello, "Sunscapes: extending the 'solar envelopes' concept through 'iso-solar surfaces'," in The 22nd conference on passive and low energy architecture, Beirut, 2005.
- [51] W. Maas, "What's Next?: How Do We Make Vertical Urban Design?," in Council on Tall Buildings and Urban Habitat (CTBUH), Shenzen, 2016.
- [52] D. A. White, "LIDAR, Point Clouds, and Their Archaeological Applications," in *Mapping Archaeological Landscapes from Space*, New York, Springer-Verlag New York, 2013, pp. 175-186.
- [53] J. Otepka, S. Ghuffar, C. Waldhauser, R. Hochreiter and N. Pfeifer, "Georeferenced Point Clouds: A Survey of Features and Point Cloud Management," *ISPRS International Journal Geo-Information*, vol. 2, no. 4, pp. 2038-1065, 2013.
- [54] M. Weinmann, Reconstruction and Analysis of 3D Scenes, Karlsruhe: Springer International Publishing, 2016.
- [55] T. Randall, "Client Guide to 3D Scanning and Data Capture," BIM Task Group, UK, 2013.
- [56] I. Kobayashi, Y. Fujita, H. Sugihara and K. Yamamoto, "Attribute Analysis of Point Cloud Data with Color Information," *Journal of Japan Society of Civil Enginers*, vol. 67, no. 2, pp. 95-102, 2011.
- [57] Y. Fujita, Y. Hoshino, S. Ogata and I. Kobayashi, "Attribute Assignment to Point Cloud Data and Its Usage," Global Journal of Computer Science and Technology, vol. 15, no. 2, pp. 2-B, 2015.
- [58] Y. Fujita, I. Kobayashi, Y. Hoshino and W. Chanseawrassamee, "Development of Attribute-Assign-Editor for Road Surface Point Cloud Data," *IACSIT International Journal of Engineering and Technology*, vol. 8, no. 3, pp. 170-176, 2016.
- [59] S. Kaasalainen, A. K. A. Krooks and H. Kaartinen, "Radiometric Calibration of Terrestrial Laser Scanners with External Reference Targets," *Remote Sensing*, vol. 1, no. 3, pp. 144-158, 2009.
- [60] A. G. Kashani, M. J. Olsen, C. Parrish and N. Wilson, "Review of LIDAR Radiometric Processing: From Ad Hoc Intensity Correction to Rigorous Radiometric Calibration," *Sensors*, vol. 15, no. 11, pp. 28099-28128, 2015.
- [61] A. Gressin, C. Mallet, J. Demantké and N. David, "Towards 3D lidar point cloud registration improvement using optimal neighborhood knowledge," *ISPRS Journal of Photogrammetry and Remote Sensing*, vol. 79, no. -, pp. 240-251, 2013.
- [62] V. Salehi and S. Wang, "Using point cloud technology for process simulation in the context of digital factory based on a system engineering integrated approach," in *The 21st International Conference on Engineering Design (ICED 17)*, Vancouver, 2017.
- [63] D. Sidiropoulou-Velidou, A. Georgopoulos and J. L. Lerma, "Exploitation of Thermal Imagery for the Detection of Pathologies in Monuments," in *Progress in Cultural Heritage Preservation*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [64] E. Gigliarelli, D. Carlea, A. Corcella and H. Porfyriou, "Historical Social Housing: Smart Analysis and Design for Conservation and Energy Conservation," in *Progress in Cultural Heritage Preservation*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [65] N. N. Staneva, "Approaches for generating 3D solid models in AutoCAD and solid works," Journal of Engineering, vol. VI, no. 3, pp. 28-31, 2008.
- [66] V. Shapiro, "Solid modeling," Spatial Automation Laboratoty, University of Wisconsin, Madison, 2001.
- [67] R. Richter and J. Döllner, "Concepts and techniques for integration, analysis and visualization of massive 3D point clouds," *Computers, Environment and Urban Systems*, vol. 45, pp. 114-124, 2014.

- [68] C. Mallet, Analyse des données lidar aéroportées à Retour d'Onde Complète pour la cartographie des milieux urbains, Paris: De Télécom ParisTech, 2010.
- [69] M. Bueno, J. Martínez-Sánchez, H. Gonzales and H. Lorenzo, "Detection of geometric keypoints and its application to point cloud coarse registration," *The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, vol. XLI, no. B3, pp. 187-194, 2016.
- [70] M. Hollaus, B. Höfle, W. Dorigo, N. Pfeifer, C. Bauerhansl and B. Regner, "Tree species classification based on full-waveform airborne laser scanning data," in *Proceedings of the SILVILASER*, Texas, 2009.
- [71] J. Heinzel and B. Koch, "Exploring full-waveform LiDAR parameters for tree species classification," International Journal of Applied Earth Observation and Geoinformation, vol. 13, no. 1, pp. 152-160, 2011.
- [72] A. L. Neuenschwander, L. Magruder and M. Tyler, "Landcover classification of small-footprint, full- waveform lidar data," *Journal of Applied Remote Sensing*, vol. 3, no. 1, pp. 1-13, 2009.
- [73] N.-J. Shih, "The Application of A 3d Scanner in the Representation of Building Construction Site," in International Association for Automation and Robotics in Construction (IAARC), Madrid, 2002.
- [74] T. Várady, R. R. Martin and J. Cox, "Reverse engineering of geometric models—an introduction," *Computer-Aided Design*, vol. 29, no. 4, pp. 255-268, 1997.
- [75] Y. Fujita, I. Kobayashi, W. Chanseawrassamee and F. Hoshino, "Application of Attributed Road Surface Point Cloud Data in Road Maintenance," *Journal of Japan Society of Civil Engineers, Ser. F3 (Civil Engineering Informatics)*, vol. 70, no. 2, pp. I_185-I_192, 2014.
- [76] E. Lin and C. Girot, "Point Cloud Components Tools for the Representation of Large Scale Landscape Architectural Projects," in *Digital Landscape Architecture*, Zurich, 2014.
- [77] E. Lin, K. Shaad and C. Girot, "Developing river rehabilitation scenarios by integrating landscape and hydrodynamic modeling for the Ciliwung River in Jakarta, Indonesia," *Sustainable cities and society*, vol. 20, pp. 180-198, 2016.
- [78] T. P. Kersten, F. Keller, J. Saenger and J. Schiewe, "Automated Generation of an Historic 4D City Model of Hamburg and its Visualization with the GE Engine," in *Progress in Cultural Heritage*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [79] M. Balzani, C. Bughi, F. Ferrari, L. Rossato and A. Tursi, "Alberti's Box: The Cultural Multimedia Project on the Architectures of Leon Battista Alberti," in *Progress in Cultural Heritage Preservation*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [80] W. Moussa, M. Abdel-Wahab and D. Frisch, "Automatic Fusion of Digital Images and Laser Scanner Data for Heritage Preservation," in *Progress in Cultural Heritage Preservation*, 4th International Conference, EuroMed 2012, Limassol, 2012.
- [81] L. Bornaz, A. Lingua and F. Rinaudo, "Engineering and Enviroonmental Applications of Laser Scanner Techniques," in ISPRS Commission III Symposium " Photogrammetric computer vision", Graz, 2002.
- [82] R. Kassner, W. Koppe, T. Schüttenberg and G. Bareth, "Analysis of the Solar Potential of Roofs By Using Official Lidar Data," in *ISPRS Congress Beijing*, Beijing, 2008.
- [83] C. Carneiro, E. Morello and G. Desthieux, "Assessment of Solar Irradiance on the Urban Fabric for the Production of Renewable Energy using LIDAR Data and Image Processing Techniques," in Advances in GIScience, Hannover, 2009.
- [84] S. Ochmann, R. Vock, R. Wessel and R. Klein, "Automatic Reconstruction of Parametric Building Models from indoor point clouds," *Computers & Graphics*, vol. 54, pp. 94-103, 2016.
- [85] T. Voegtle, I. R. Schwab and T. Landes, "Influences of different materials on the measurements of a terrestrial laser scanner (tls)," *The International Archives of the Photogrammetry, Remote Sensing and Spatial Information Sciences*, vol. 37, no. 5, p. 1061–1066, 2008.
- [86] ArcMap, "What is lidar intensity data?," 2016. [Online]. Available: http://desktop.arcgis.com/en/ arcmap/10.3/manage-data/las-dataset/what-is-intensity-data-.htm. [Accessed 7 12 2018].
- [87] J. Aurgho, "High Rise Morphologies: Architectural Form FInding in a Performance Design Search Space of Dense Urban Contexts," in *The 35th Annual Conference of the Association for Computer Aided Design in Architecture*, Cincinnati, 2015.
- [88] M. Leidi and A. Schlüter, "Volumetric insolation analysis," in CleanTech for Sustainable Buildings From Nano to Urban Scale (CISBAT 2011), Lausanne, 2011.
- [89] F. Ozel, "SolarPierce: A Solar Path-Based Generative System," in Computation and Performance Proceedings of the 31st eCAADe Conference, Delft, 2013.
- [90] I. W. Foged, "Architectural thermal forms," in Computation and Performance Proceedings of the 31st eCAADe Conference, Delft, 2013.

- [91] L. Armstrong, G. Gardner and C. James, "Evolutionary Solar Architecture Generative Solar Design Through Soft Forms and Rigid Logics," in *Parametricism Vs. Materialism: Evolution of Digital Technologies for Development* [8th ASCAAD Conference Proceedings], London, 2016.
- [92] P. Marin, J.-C. Bignon and H. Lequay, "Generative exploration of architectural envelope responding to solar passive qualities," in *Design and Decision Support Systems in Architecture and Urban Planning* [DDSS 2008], Eindhoven, 2008.
- [93] J. Da Veiga and P. La Roche, "A Computer Solar Analysis Tool for the Design and Manufacturing of Complex Architectural Envelopes: EvSurf," in *Proceedings of the 6th Iberoamerican Congress of Digital Graphics* [SIGraDi 2002], Caracas, 2012.
- [94] X. Yang, L. Grobe and W. Stephen, "Simulation of reflected daylight from building envelopes," in Proceeding of BS2013: 13th Conference of International Building Performance Simulation Association, Chambery, 2013.
- [95] D. Girardeau-Montaut, "Cloud-Compare," 2015. [Online]. Available: http://www.cloudcompare.org/doc/ qCC/CloudCompare%20v2.6.1%20-%20User%20manual.pdf. [Accessed 5 2 2018].
- [96] M. S. Roudsari and P. Michelle, "Ladybug: a parametric environmental plugin for grasshopper to help designers create an environmentally-conscious design," Lyon, 2013.
- [97] M. E. Charlton, S. J. Coveney and T. McCarthy, "Issues in Laser Scanning," in Laser Scanning for the Environmental Sciences, West Sussex, Wiley-Blackwell, 2009, pp. 35-48.
- [98] S. D. Bartolo and R. Salvini, "Multitemporal Terrestrial Laser Scanning for Marble Extraction Assessment in an Underground Quarry of the Apuan Alps (Italy)," Sensors, vol. 19, no. 3, pp. 1-10, 2019.
- [99] Faro, "Faro Laser Scanner User Manual," Faro Technologies Inc., Florida, 2018.
- [100] C. Suchocki and W. Błaszczak-B, ak, "Down-Sampling of Point Clouds for the Technical Diagnostics of Buildings and Structures," *Geosciences*, vol. 9, no. 2, pp. 1-14, 2019.
- [101] J. Armesto-González, B. Riveiro-Rodríguez, D. González-Aguilera and M. T. Rivas-Brea, "Terrestrial laser scanning intensity data applied to damage detection for historical buildings," *Journal of archaeological science*, vol. 37, no. 12, pp. 3037-3047, 2010.
- [102] S. S. Soudarissanane, The Geometry of Terrestrial Laser Scanning, Delft: PhD-thesis, Faculty of Civil Engineering and Geoscience, TU Delft, 2016.
- [103] S. Sasidharan, "A Normalization scheme for Terrestrial LiDAR Intensity Data by Range and Incidence Angle," International Journal of Emerging Technology and Advanced Engineering, vol. 6, no. 5, pp. 322-328, 2016.
- [104] A. Boulch and R. Marlet, "Deep Learning for Robust Normal Estimation in Unstructured Point Clouds," in *Eurographics Symposium on Geometry Processing*, Berlin, 2016.
- [105] J. Shin, H. Park and T. Kim, "Characteristics of laser backscattering intensity to detect frozen and wet surfaces on roads," *Journal of Sensors*, vol. 2019, no. 1, pp. 1-9, 2019.
- [106] S. V. Szokolay, Introduction to architectural science: The Basis of Sustainable Design, Oxford: Elsevier/ Architectural Press, 2008.
- [107] D. Majcen, Predicting energy consumption and savings in the housing stock, Delft: PhD dissertation ABE tudelft, 2016.
- [108] D. Chwieduk, Solar energy in Buildings: Thermal Balance for Efficient Heating and Cooling, London: Elsevier Science Publishing Co Inc, 2014.



PART B A computational workflow for generating a voxel-based design approach based on subtractive shading envelopes and attribute information of point cloud data

- This study proposes a voxel-based design approach based on the subtractive ABSTRACT mechanism of shading envelopes and attributes information of point cloud data in tropical climates. In particular, the proposed method evaluates a volumetric sample of new buildings based on predefined shading performance criteria. With the support of geometric and radiometric information stored in point cloud, such as position (XYZ), color (RGB), and reflection intensity (I), an integrated computational workflow between passive design strategy and 3D scanning technology is developed. It aims not only to compensate for some pertinent aspects of the current 3D site modeling, such as vegetation and surrounding buildings, but also to investigate surface characteristics of existing contexts, such as visible sun vectors and material properties. These aspects are relevant for conducting a comprehensively environmental simulation, while averting negative microclimatic impacts when locating the new building into the existing context. Ultimately, this study may support architects for taking decision-making in conceptual design stage based on the real contextual conditions.
- KEYWORDS voxel-design approach; shading envelopes; point cloud data; computational design method; passive design strategy

5.7 Introduction

5.7.1 General background

The rapid development of 3D laser scanning technology has reached across multipledisciplines within design and engineering. However, the practical implementation of this technology is often applied in major fields, such as photogrammetry [1] [2], cultural heritage [3] [4] [5] [6], and environmental engineering [7] [8]. Digital reconstruction as one of the main subjects in these scopes has been used predominantly for building performance assessments [9] [10], where the contextual modeling of existing studies is frequently based on a 3D solid modeling context [11] [12]. As a consequence, high computational costs and time are required to cover the entire set of complex building forms. On the other hand, the use of point clouds during the early stage of architectural design has not yet been fully explored, especially related to the performance simulation task and design decision support.

As an entity of 3D data scanning, Otepka et al. [13] illustrates the point cloud as a universal denominator for laser scanning and photogrammetric data. Its data structure is principally characterized by position information (XYZ) as a permanent element coupled with auxiliary information attached to it, such as color attributes (RGB), reflection intensity (I), and any abstract information [14]. The prospective applications of these attributes not only represent metadata information of the real environment, but also enable designers and researchers to perform numerous tasks, such as data processing, visualization, and analysis. Moreover, this can help architects further to address environmental design issues, such as solar and shading performance.

The technological advancement of point cloud reveals the relevance of integrating it into the passive design strategy, especially when dealing with generative architecture designs that currently lack several relevant aspects. For example, first, understanding the site characteristics of an existing environment. While 3D site modeling primarily deals with a building-oriented context, surrounding properties, such as vegetation and adjacent buildings, are often neglected [15]. This may not only affect the performance simulation of a proposed design, but also potentially create microclimatic issues when it comes to the real context. Second, the absence of surface properties, such as roughness and material characteristics on a manually-built 3D model, may cause a crucial discrepancy when dealing with environmental simulation between planned

and existing buildings [16]. With geometric and radiometric information extracted from point cloud data, this study, therefore, proposes an integrated passive design approach based on shading performances of new and existing contexts.

As a contextual design approach, this study specifically investigates the idea of subtractive shading envelopes that are principally extracted from the concept of solar envelopes initially introduced by Knowles [17]. In this regards, solar envelopes permit architects to design appropriate massing of a new building into the existing environment by guaranteeing desirable sun access for surrounding buildings during the critical period [18], while subtractive shading approach aims to extract potential performances of the existing contexts and integrate it with a 3D volumetric massing of a proposed building based on predefined shading performance criteria.

Since then, various computational methods of solar envelopes, such as descriptive geometry, solar obstruction angle, and constructive solid geometry have been defined [19]. These approaches have successfully demonstrated the concept of solar envelopes into various urban settings (e.g., single building, open space, and urban scale) and multiple functional utilities (e.g., housing, offices, and commercial buildings). It is worth noting that the contextual settings of the existing methods primarily focus on temperate zones of southern and northern hemisphere countries, which have distinct climatic conditions during the four-seasons. This means that design objectives and climatic parameters of most existing methods for solar envelopes become less applicable when it comes to tropical countries, especially for those located on the Equator, such as Indonesia. Since tropical countries present wet and dry seasons all year round, the objective of solar envelopes significantly shifts and aims to minimize the penetration of direct sun access to the buildings, due to high temperatures. For example, housing in Indonesia is typically designed in a way that prohibits direct sunlight from penetrating the dwelling, especially into primary living spaces, so that temperatures are kept low during the day. Consequently, the air conditioner (AC) frequently becomes a short-term solution to mitigate the building's temperature, which unfortunately contributes to the annual increase in energy consumption [20]. Accordingly, shading conditions become considerably relevant for urban forms generation in tropical contexts. This study specifically proposes an environmental design strategy that integrates shading performance aspects and attributes information from point cloud data through a computational workflow of a voxel-based design approach.

Furthermore, the following section will present a theoretical background of the existing studies, starting from solar envelopes, shading envelopes, subtractive solar envelopes, and subtractive solar envelopes based on point cloud data so-called SOLEN approach. This will be followed with a description of a proposed method in

Section 2, while case studies in Section 3. Section 4 will comprehensively discuss the findings of the simulation results. Lastly, Section 5 will describe the conclusions, limitations, and future recommendations of the study.

5.7.2 Related works

5.7.2.1 Solar envelopes

In the remote past, the concept of vernacular architecture has successfully contributed to preserving sustainable building envelopes [21] [22]. This can be observed through the development of the Indus Valley, Mohenjo-Daro in India, 2500 BC [23], El-Lahun village in Egypt (1857–1700 BC) [24], and many classical Greek cities, such as Olynthus in North Hill—a city designed to benefit from passive solar energy for the heating of buildings [25]. This strategy was known as solar-oriented homes or so-called "solar architecture". Since then, solar architecture has become an essential quide for designers to develop sustainable urban planning. For example, Andrea Palladio has discussed the proper norms of city planning by considering wide streets for cold climate countries and narrow streets for tropical countries [26]. Additionally, Ildefons Cerdà integrated green areas into the public and private space of Barcelona in his masterplan of the city so as to enhance the comfort of inhabitants [27]. During the industrial revolution, the idea of urban solar policy or refers to the post-war housing was also implemented in France (in 1912), Germany (in 1920), and New York (in 1916). These examples have shown a positive contribution to architectural buildings, not only to reduce the energy consumption of the built environment [28], but more importantly, to support a healthy living environment [29]. Furthermore, the idea of solar accessibility has been elaborated further through the concept of solar envelopes.

By definition, solar envelopes stand for imaginary boundaries that are constructed based on the sun's movement. It is regulated based on specific space-time constraints [18]. According to this principle, solar envelopes can be transformed into geographic and climatic properties within the size of on-site buildings [19]. Geographic properties deal with a group of parameters that define the spatial relationship between the design plot and existing context related to orientation typology, surrounding facades, sidewalks, building height, longitude, latitude, floor area ratio (FAR), setback, shadow fences, and street sizes. On the other hand, climatic properties consist of parameters that determine the geometric

transformation of the proposed building based on the time construction, such as cut-off-times, solar angle, sun path, dry bulb temperature, sun access hours, solar altitude, and solar azimuth. These parameters are used not only to generate solar envelopes, but also to identify the character and qualities of the built environment. For example, orientation plays a great role in examining the geometrical shape of solar envelopes, especially when dealing with the street layout in relation to various angular values, colonnades with a variety of direct solar radiations, and solar urban layouts [30]. A seasonal leaf cover from the surrounding vegetation can also affect the geometrical configuration of solar envelopes as it may be considered as a part of geographic elements for violating excessive direct sun access during summer [16]. Besides, the solar angle as a climatic property is used to determine geometric solar envelopes based on the construction planes [31]. It is mostly employed for simple shape plots, with borders aligned with the main cardinal directions in east-west (EW) and north-south (NS) and for the main hours and days, such as the noontime during summer and winter solstice, and spring/autumn equinoxes.

5.7.2.2 Shading envelopes

As opposed to solar envelopes, the concept of shading envelopes primarily deals with the solar radiation-reduction to achieve appropriate daylight for urban equatorial climates. This permits architects not only to establish a geometrical configuration of solar shading envelopes, but also to control the direct sun exposure of the building's own façades and surroundings during a critical time. Two different types of shading approaches are identified as follows:

Building forms

In this part, the concept of shading envelopes aims to promote a passive design strategy through the form generation within the conceptual design stage. This means that the volumetric shape of proposed buildings is developed based on the consideration of solar shading criteria. An interesting example can be observed through the concept of "shadow umbrella" introduced by Emmanuel [32]. This concept proposed a design approach of shading strategies incorporated with natural elements, such as vegetation and water bodies, aiming to create shading for adjacent buildings and to mitigate the urban heat island for tropical neighborhood areas. Accordingly, a new configuration of urban block shapes with a thermally comfortable can be generated. DeKay [33] addressed a similar strategy with the concept of climatic envelopes. The climatic envelope primarily aims to generate a building mass

that guarantees access for diffused daylight and solar energy resources for the surrounding buildings. This concept specifically contains a geometrical intersection between daylight and solar envelopes based on the sky exposure plane and solar protection plane, respectively. In this case, sky exposure plane refers to imaginary sloping planes that allow penetration of natural light and air on the building facades in higher density districts [35]; meanwhile, solar protection plane refers to an inclined plane that is generated from the profile angle, the so-called vertical shadow angle (VSA) [34]. In a similar vein, Capeluto [35] proposed the concept of self-shading envelopes by extending the functional properties of the sky exposure plane through the solar collection envelopes (SCE) model. The geometrical configuration of self-shading orientation. Therefore, the envelope's roof areas should be larger than the bottom part of the envelope geometries. This approach aims to avoid overheating and at the same time, to maximize the self-building protection for a certain period during summer.

Building components

In addition to building forms, shading approaches are also applied to specific building components, such as windows, cantilevers, and openings on the building façade, based on the determined building shapes. Although this study limits the scope of investigation on a form-finding design solution, some studies on shading mechanisms drive potential efforts to handle more complex projects. For example, Yezioro and Shaviv [36] proposed SHADING as a design and evaluation tool for analyzing mutual shading between buildings and other surrounding properties, such as vegetation. It specifically calculates the insolated fraction on the building surfaces quantitatively and performs a ray-tracing algorithm to identify the shadows visually at a particular time. Similarly, Marsh [37] also used a ray-tracing analysis to identify the external obstruction of solar intensity on the optimized shape of shading geometries. In this case, optimized shading designs have effectively accommodated passive solar control through the building apertures [38] [39]. Other approaches deal with a graphic solution of shading design tools [42], form-finding of static exterior shading devices called SHADERADE [40], and a cellular method to define optimal shading patterns [41].

Although these approaches may address the aspect of tropical design contexts, they lack some critical aspects during the simulation of shading envelopes. First, the quality of solar radiation (i.e., the quantity of direct sunlight hours) received by surrounding buildings is not taken into account by most methods and tools, due to a fixed period when determining direct sun access. Consequently, all geometrical

shapes of existing buildings are treated similarly when receiving the irradiation qualities without considering the obstruction of properties and building orientation of the plot. Second, shadow fences of surrounding buildings' facades are primarily regulated by a Z-axis. This makes the design configuration of the resulting envelopes rely only on the horizontal shading lines. In fact, shading areas of surrounding façades are more complex, especially when dealing with dense areas and multiple urban forms. In order to compensate for these issues, the existing studies present some relevant aspects, such as subtractive mechanism and point cloud data that may be useful for further development. This will be discussed in the following section.

5.7.2.3 Subtractive solar envelopes

As previously mentioned in the general background, this approach specifically subtracts a volumetric matrix of the 3D plot according to solar accessibility criteria. This is done by projecting solar vectors acquired from the number of direct sunlight hours on surrounding building facades. In principle, this mechanism has been addressed by Leide and Schlüter [42] [43] via volumetric site analysis (VSA). Their approaches aim to explore urban site information by simulating multiple environmental performances, such as solar radiation, airflow, visibility, thermal comfort, and wind velocity through volumetric insolation analysis (VIA), volumetric visibility analysis (VVA), and computational fluid dynamics (CFD). Such this approach and other related developments [44] [45] [46], however, merely focus on the architectural form-finding without any further consideration on the concept and design principles of solar or shading envelopes.

On the other hand, De Luca [47] [48] [49] and Darmon [50] have proposed a similar subtractive mechanism based on the performance criteria of solar envelopes or socalled subtractive solar envelopes. This approach specifically involves sun visibility that aims to evaluate sun vectors from a predefined shadow grid of surrounding building windows without any obstruction from other existing buildings. In parallel, ray-tracing analysis is performed from surrounding windows to the voxels within the 3D plot using a Boolean expression (true or false statement). In this operation, the true statement will be executed when sun vectors hit or intersect the 3D polyhedra, and accordingly, the voxels subtraction procedure to the 3D polyhedra can be performed. Meanwhile, the false statement indicates an unsuccessful intersection. This condition means that voxels that are not intersected may contribute to the generation of geometric solar envelopes. The subtractive solar envelopes ultimately permit architects not only to deal with various geometric configurations based on solar performance analysis, but also to highlight the potential use of voxel-based generative designs for urban environments. However, despite such potential improvements, aspects (such as sun visibility and ray-tracing analysis) pose several critical considerations, especially when addressing the contextual design strategy. For example, the identification of visible sun vectors merely considers the surrounding building contexts while neglecting relevant geometric properties, such as vegetation and other site characteristics (e.g., material properties). This consequently can affect irradiation analysis during the environmental performance simulation between a proposed building and the existing contexts. Besides, the window's grid configuration lacks in representing the insolation values of building facades during the ray-tracing analysis, due to its limited consideration of geometric centroids of each surrounding window.

In order to address these gaps, the existing workflow of subtractive solar envelopes has been improved by incorporating it with the prospective application of 3D laser scanning (point cloud data). By exploiting the practical usability of the point cloud in different fields, the scope of information properties in the real contexts can be improved, and to some extent, it becomes relevant to the specific aforementioned issues. Regarding sun visibility, the 3D point cloud not only captures the most and the least sun-exposed areas through buildings and vegetation, but also investigating material performances of contextual datasets through optical (reflectivity and translucency) and thermal properties (albedo and emissivity). Besides, the ray-tracing analysis between a proposed building and surrounding contexts is performed based on 3D point cloud datasets of the existing context. In other words, it substitutes the surrounding window's grid on the building facades proposed by the existing approach.

5.7.2.4 Subtractive solar envelopes based on point cloud data (SOLEN)

With the support of geometric and radiometric properties [51] [52] stored in a 3D point cloud, the integrated computational workflow between subtractive solar envelopes and attribute information of point cloud data has been established [53] [54]. It specifically integrates functional properties of position information (XYZ), color information (RGB), and reflection intensity (I). Each of these attributes caters to different potential tasks. For example, color information (RGB) can be used not only to extract and segment certain areas within the dataset based on its values [55] [56], but also to translate them into new information properties [57] [58].

This can include converting data attribution of RGB into HSV values to perform the measurement analysis and road maintenance [59] and extracting the semantic information of the indoor environment with automatic room labeling [60] [61]. Meanwhile, reflection intensity (I) predominantly deals with surface and spectral properties of the scanned objects [62] as it constitutes the return strength values of laser pulse or backscattered echo for each recorded point [52]. Accordingly, the intensity values can be used not only to map geological layers and pavement lines [55], but also to detect natural phenomena, such as frozen and wet surfaces on roads [65], and the measurement of seasonal snow cover [67]. On the other hand, position information (XYZ) constitutes of geographic coordinate that marks each recorded point's specific location. This attribute plays a great role in synchronizing index between color and intensity values as it can attach to both attributes. Thus, complex areas of the dataset can be precisely extracted based on selected values. In general, these attributes contribute not only to extend particular performances of 3D point cloud data, such as identification of existing material properties [68]. but also to extend the applicability of environmental analysis during the conceptual design stage.

Before performing the subtractive solar envelopes, the dataset correction is required to minimize erroneous levels during scanning. In this case, some aspects, such as environmental and meteorological conditions, atmospheric pollution [69] [70], unit specification of the scanner, surface properties of the scanned objects, and scanning geometries [71] can principally affect metadata information of the datasets during scanning. While correcting all these variables seems impractical, due to some local constraints (i.e., manufacturers), this study specifically focuses on correcting the acquisition geometry based on the angle of incidence, which is relevant to the proposed subtractive mechanism. Having established the corrected datasets, it can be further used not only to perform the ray-tracing analysis between the 3D polyhedra and selected solar vectors, but also to calculate the material properties of existing contexts that are useful to evaluate the environmental performance of a proposed building. In parallel, insolation analysis is performed to identify the potential solar energy of the resulting solar envelopes.

5.8 Proposed methods for subtractive shading envelopes



FIG. 5.15 An overview of the proposed computational workflow.

This study proposes a computational framework that consists of three phases: input, simulation process, and output (see Figure 5.15). Within a simulation process, five sequential procedures are developed, ranging from A (input parameters) to E (form generation process of self-shading envelopes).

To perform specific tasks in each predefined step, the proposed workflow was supported by several digital tools. For example, Topcon GLS 2000 [71] was employed to collect high-resolution point cloud datasets. It was complemented with Maptek I-Site [72] to perform dataset registrations, coloring, modeling, and most importantly, to facilitate the data transfer from scanner to the workstation in any designated format. Moreover, Cloud Compare (CC) [73] was used, not only for dataset preparation, but also for dataset pre-processing, such as attribute selection, dataset formatting, scalar field features, and the normal surface calculation. In alignment with that, Matlab [74] was specifically used to assist dataset correction (i.e., optimal normal values, intensity correction, and dataset subsampling), while Rhino [75] (coupled with Grasshopper [76] for visual scripting) was employed to develop a 3D geometric model of a proposed building and to perform solar simulation analysis by using a Ladybug [77] component in Grasshopper. Furthermore, a detailed task, dataset formats, and outputs of each step are addressed below.

5.8.1 Stage A – Input

5.8.1.1 Step A – Preparation of Input Parameters

As a starting point for the computational procedures, the input section refers to step A, which contains a series of parameters that are used to construct contextual settings of the subtractive shading envelopes. Specifically, it consists of climatic properties that correspond to coordinate location (i.e., longitude and latitude position based on World Geodetic System 1984, WGS84) and selected periods (i.e., month, year, day, and hours) (see Figure 5.16). The longitude and latitude coordinates influence the envelope's geometrical properties on the basis of sun position and solar angle. For example, high-latitude sites are characterized by a small angle of solar altitude and smaller degrees of solar radiation. Moreover, specific periods are required to obtain the number of critical hours of natural illumination that affects a proposed building and surrounding contexts.



FIG. 5.16 Preparation of input parameters.

On the other hand, geographic properties include surrounding contexts (i.e., existing buildings and vegetation) and a geometric model of the proposed building. In this regard, the surrounding context contains a 3D point cloud data of the existing environment. As a raw dataset, its format properties often rely on the type of 3D scanner, but as long as the required attribute information (i.e., XYZ, RGB, and reflection intensity) is legible, any raw dataset formats are acceptable (i.e., PTX file). Meanwhile, several parameters, such as height, width, floors, setback, and building function, must be established to generate an initial 3D envelope for a proposed building. These inputs are then executed into the following Step B (dataset preparation) based on corresponding parameters and tasks.

5.8.2 Stage B – Simulation Process

This section focuses on translating the raw datasets into a simulation model by examining three main steps (i.e., dataset preparation, selection criteria, and form generation process of shading envelopes). Each step serves specific actions that are performed sequentially based on specific computational tasks. As illustrated in Figure 5.15, this section adopts several workflows that are partially implemented from previous works. For example, pre-processing point cloud datasets and the calculation of material properties (i.e., albedo values) are indicated with a green text

[67]. These works focused on developing a material database of existing contexts and solar radiation analysis based on a small sample of point cloud data. Next, 3D polyhedra and sun visibility analysis are illustrated in the blue text [49,50]. These works primarily investigated a voxel-based generative design of subtractive solar envelopes based on 3D and parametric modeling. Last, the ray-tracing analysis between a proposed building and point cloud data of surrounding contexts is illustrated in the red text [56]. This work refers to a form generation of subtractive solar envelopes that consider surface properties of existing contexts based on geometric and radiometric information of point cloud data. While these previous works address different objectives, some features are still relevant for supporting an integrated design concept and computational workflow for establishing the subtractive shading envelopes.

To illustrate specific tasks in this stage, a detailed discussion of each proposed step is presented below.

5.8.2.1 Step B – Dataset preparation

This step aims to prepare all the necessary datasets to be readily used in the simulation model. After establishing the input from climatic and geographic properties, five tasks are required to perform. Two of these tasks (i.e., sun vectors calculation and the initial envelopes generation) can be run in parallel while the other three tasks (i.e., pre-processing datasets, dataset correction, and subsampling dataset) can be executed sequentially (see Figure 5.17).

Task 1, sun vectors refer to the number of sunlight hours that must be preserved on surrounding facades during the required period. As compared with cut-off times, which refer to a fixed period, sun access duration can be selected from a range of available hours for a specific façade on a specific day or for each day during a specific period. In doing so, sun visibility plays a crucial role when determining the relevant sun vectors.

Task 2, the initial envelope of a proposed building, is generated based on the predefined criteria. In this case, the functional program of a proposed building is projected as a public library, as well as communal space for the local community. To support the main activities, some spaces are established, such as a reading room, meeting room, toilet, and exhibition areas. Accordingly, the building needs to be accessible, and its indoor environment should be thermally comfortable for supporting the daily activities.



FIG. 5.17 Detailed procedures for the dataset preparation.

Task 3, the pre-processing tasks start with the dataset registration. This process aims to locate recorded datasets into a standard coordinate system based on the reflector position and GPS orientation. Afterwards, the outlier (unnecessary cloud of points) removal is performed not only to clean the boundary of selected datasets, but also to filter noises created during scanning. The dataset formatting also plays an important role in compensating for interoperability issues during the simulation process. In this case, the initial format of 3D raw point clouds (i.e., PTX) and its metadata are converted to E57 and ASCII files, respectively, in order to be accessible for various digital processing tools. In addition to this, the scalar field of the dataset is activated to identify the attached values' scale in each attribute. Last, normal surfaces (NxNyNz) of the dataset are calculated to find the appropriate normal values of each projected point during scanning. As normal values of the raw point cloud are excluded in the typical attribute properties, various angles of incidence, ranging from the sample of 10° to 90°, are firstly computed to each data scan. This is done by using the Hough Normal plugin [78] in CC, due to the original form of unstructured raw point cloud data. In this regard, each point within each data scan has different preliminary normal values. Ultimately, task 3 results in several outputs, such as a 3D model of selected datasets, attribute information of point cloud data with raw intensity values, and the normal surface of each data scan at various incidence angles.

Task 4, the dataset correction, is performed to compensate for the scattering condition of unstructured point clouds during scanning, specifically for amending radiometric properties of the dataset. It is worth noting that this procedure can only be applied on a single scan, due to the potentiality of mixing a reference point of the scanner and intensity values on the merge datasets [54]. The correction step starts with finding the average distribution of optimal projection points from the preliminary normal values. This achieves a reliable normal surface on each applied angle in the data scan. To do so, the following equation [79] is applied with an assumption that the original position of a 3D scanner located at (0,0,0).

$$i = \cos^{-1}\left(\frac{\overline{dn} \cdot \overline{dl}}{\left|\overline{dn} \cdot \left|\overline{dl}\right|}\right)$$

(1)

where:

i = incident angle (α) \overline{dn} = direction of surface normal \overline{dl} = direction of the laser pulse

After configuring the point distribution from a various range of cosinus products, an evaluation is conducted to the standard deviation of each registered cosine value within the dataset. It aims to identify the pattern of point density to reduce scattered and coarse point clouds through the dataset truncation. Afterwards, the truncated datasets can be used for the intensity correction. In this regard, the angle of incidence becomes a relevant factor for correcting the dataset's acquisition geometry, given that instrumental effects highly affect the raw intensity value of TLS datasets. This procedure is executed based on the following equation [54].

$$I_c = I_{raw} \cdot \frac{1}{\cos \alpha}$$

where: I_c = corrected intensity I_{raw} = original intensity α = angle of incidence

Task 5, the subsampling dataset, is performed in CC to reduce the density of points during the simulation. However, this procedure creates interoperability issues when the resulting dataset is matched and visualized to the initial 3D model in Rhino. Specifically, the 3D model cannot directly recognize the input of subsampled datasets, due to different units and scales of the attribute information. Therefore, synchronizing the initial index between the two datasets identifies metadata information in the geometric 3D model. This further permits the extraction of certain areas in the 3D dataset based on selected attributes values.

5.8.2.2 Step C – Selection criteria

After preparing the corrected datasets from climatic and geographic properties, this study sets two environmental performance criteria that support the geometric generation: sun visibility and material properties (see Figure 5.18). Sun visibility plays a crucial role in filtering the sun vectors that have direct access to the dataset of surrounding contexts. To do so, sun vectors generated from the indicated period are multiplied with normal vectors extracted from subsampled datasets. The resulting normal irradiance values are then evaluated on the basis of the projected angle. In this case, irradiance values with equal and larger than 90° are eliminated as they consist of zero and negative cosine values. Accordingly, these values are then excluded within the list of visible sun vectors because the surface of the datasets does not properly absorb their solar energy. Afterward, the resulting values can be used to select the corresponding points within the dataset.

On the other hand, material aspects are used to measure the performance behavior of the existing site's surface properties. This allows architects to identify susceptible areas that may affect the geometrical performance of the proposed design. This study specifically computes albedo values to detect the absorbance percentage of solar energy on surrounding contexts by considering the RGB color and corrected intensity (Ic) of contextual datasets. A detailed procedure regarding the calculation of albedo values can be found in our previous research [67]. Furthermore, the resulting albedo values are filtered based on the threshold below 0.3. Although this setting indicates the low albedo, it can be used not only to identify areas that contain a high level of heat absorbance, but also to analyze and mitigate microclimatic impacts of the surrounding areas especially related to thermal issues and urban heat island (UHI) effect. In order to avoid that, these areas need to be blocked from direct sun exposures by excluding the indicated surfaces with high albedo values. Afterward, the resulting indexes of low albedo values (below 0.3) are synchronized not only with selected normal vectors from the sun visibility to register the corresponding normal surface of the dataset, but also with XYZ attributes to select the matching points within the dataset.





5.8.2.3 Step D – Form generation process of shading envelopes

After establishing all the required parameters from the selection criteria, this step focuses on developing the simulation workflow (see Figure 5.19). It starts with the first ray-tracing analysis that requires input from visible sun vectors, selected corresponding points from sun visibility, and 3D polyhedra of a proposed building. This procedure principally applies a Boolean expression to assign true or false conditions on selected voxels within the 3D polyhedra array. The ray-tracing analysis-01 generates intersecting rays that are then evaluated based on the predefined criteria of direct sun access. In this case, voxels that are not blocking sun access to the proposed buildings or are categorized as an unsuccessful intersection with the 3D polyhedra will be considered part of the shading voxels (refer to voxels-02 in Figure 5.19). This workflow can also be called as a reverse solar envelope. Meanwhile, voxels that receive sun access will be forwarded to a later step in the second ray-tracing analysis (refer to voxels-01 in Figure 5.19).

Furthermore, reference points are generated from the voxels-02 to be used in the simulation of ray-tracing analysis-02. As a follow-up to the previous procedure, the ray-tracing analysis-02 aims to maximize the geometric generation of shading voxels. In doing so, by changing the basis projection of initial reference points originated from surrounding buildings to the voxels-02 may compensate for geometric obstruction of polyhedra at a certain projection angle in the ray-tracing analysis-01. Instead of applying this procedure to the original 3D polyhedra, it is used to re-evaluate voxels-01 based on reference points of voxels-02 so as to identify additional voxels (refer to voxels-03) that fulfill the criteria of receiving shading condition. As for the input for material properties, a similar procedure of ray-tracing analysis is also performed by considering the lowest albedo values (ranging from 0 to 0.3) applied to surround contexts. These results in voxels-04 so that in total, three groups of voxels (i.e., voxels-02, voxels-03, and voxels-04) are generated to shade surrounding buildings. These voxels are then combined into one group, voxels-05. To ensure that a shading condition also applies to a proposed building, the self-shading workflow is performed in the following stage.



FIG. 5.19 Detailed procedures for the design simulation.

5.8.3.1 Step E – Form Generation Process of Self-Shading Envelopes

As a last stage of the computational workflow, the output contains final tasks to generate a geometric configuration of self-shading envelopes (see Figure 5.20). The first task begins with selecting the upper part of the stacking voxels. This upper part acts as a roof or shelter to guarantee a shading condition for all properties under its envelope within a predefined period. Afterward, a solar protection plane can be applied on the bottom surfaces of the upper part of the voxels. It aims to establish reference points that are used as the basis for ray projections. Furthermore, the raytracing analysis-04 is performed by considering inputs from reference points of the upper voxels, the remaining stacking voxels (i.e., bottom part), and initial sun vectors calculated for the analysis period under consideration. This simulation will evaluate the remaining voxels by maintaining the one that receives a shading condition while removing the unshaded voxels. The resulting voxels (refer to voxels-06) are then combined with the upper voxels to establish the geometric envelope for each data scan (refer to voxels-07 and voxels-08). After identifying self-shading voxels for each data scan, the final step is to combine all these voxels into a final geometric envelope that represents a final configuration of envelopes (refers to voxels-09).



FIG. 5.20 Detailed procedures for generating the final output of self-shading envelopes.

5.9 Dataset collection

In regards to the dataset collection of on-site scanning, this study needs to fulfill at least two major aspects, as also mentioned in our previous study [56]. First, the collected point clouds should consist of Terrestrial Laser Scanning (TLS) datasets. It aims not only to accommodate more accurate geometric properties and better reality-based representation than Airborne Laser Scanning (ALS), but also to capture specific areas with various contextual elements, such as vegetation, temporal objects, material, and other elements, that may potentially contribute to the simulation model. Second, metadata information stored in a point cloud should at least contain typical attributes, such as XYZ, RGB, and reflection intensity. Geometric and radiometric properties within these attributes are used to identify material properties of the existing context and to conduct environmental performance analysis.

Furthermore, to demonstrate the proposed workflow, this study collaborates with SHAU, an architectural firm located in Bandung, West Java, Indonesia, to collect 3D point cloud data to design a new public library. This project is located in Citarip, West Java, Indonesia, (6°56'18.4"S 107°35'15.1"E, with ellipsoid reference WGS84) and surrounded by some vegetation and massive walls of neighboring buildings. The collected dataset was gathered using Topcon GLS-2000M incorporated with reflectors and GPS devices to achieve an accurate position during the registration process. As a light-weight and high-speed 3D laser scanner, this tool is also featured with a rugged design instrument (for use in the field) and full-dome field of view (FOV) with a selectable laser class, enabling the user to scan in extreme work environments and eye-safety concerns in dense areas. A detailed specification of the tool can be observed below (see Table 5.2):
TABLE 5.2 Detailed specifications of 3D scanner [71]							
Parameters	Performance specification unit						
System performance							
Maximum range (at 90% reflectivity) GLS-2000M	350 m (standard)						
Single point accuracy Distance Angle	3.5 mm (1-150m), 1 sigma 6"						
Tilt sensor Type Range	Liquid 2-axis tilt sensor +/- 6						
Target detection accuracy	3" at 50 m						
Laser scanning system							
Туре	Pulse (time-of-flight) precise scan technology						
Laser class	3R (high speed/standard) 1 M (low power)						
Field of view (per scan)	Horizontal - 360° Vertical - 270°						
Spot size	4 mm at 20 m (FWHM)						
Scan rate	High Speed: Up to 120,000 points/sec						
Physical and environmental							
Operation temp	-5°C to 45°C						
Dust/humidity	IP54						
Scanning control							
Scan time and resolution	Interval 12.5mm: High Speed: 01:46 Standard: 03:31 Low Power: 04:22						

This tool's performance specifications allow the dataset collection with only three single scans to capture a sufficient scene for the selected site (see Figure 5.21). With these three single scans, the dataset's computational performance simulation can also be more manageable and relevant to the currently proposed workflow. Given that this approach is part of the exploratory study, architects need to select not only important information, but also a degree of detailed properties that are relevant to the simulation. In this regard, the setting of each scan approximately consists of four million points per scan within five minutes with a resolution 12.5×12.5 mm @10m. The distance between scanners and the designated objects are principally determined based on the approximate coverage of the scanner capacity, which may cover 360° of the horizontal field of view (FOV) with the distance between the scanner and the object with 2.75 m. This is because data scan coverage areas

need to capture the surrounding wall and corner spots behind the temporary shelters. Meanwhile, the longest distance is obtained by Scan 01 with 34.9 m, due to the diagonal position of the scanner to the corner of the site. Besides, in this dataset collection, it is worth noting that as long as the entire scenes of the selected site and relevant objects are covered, the scanner can be located at any appropriate distances.



FIG. 5.21 Dataset collection with different views captured in relation to the scanner position

After establishing the selected dataset, this section presents the analysis results of the implemented workflow. It follows the five aforementioned steps as follows:

5.10.1 Step A – Input parameters

As previously described in the section about presenting the proposed workflow, input parameters consist of climatic and geographic properties. As for climatic properties, due to a constant temperature over the year in Indonesia [80], this study sets April 21st, 2019, as a sample of the selected period that represents a starting date for a dry season. Although the selected time range is limited, this simulation takes place as an exploratory study that focuses on exploring the feasibility of integrated computational workflow for shading envelopes based on attributes information of point cloud data. This specific duration is furthermore defined for four hours, starting from 11 am to 3 pm, which is averagely representing the highest temperature during the daylight hours. In parallel, a time-step for the simulation is set smaller than one hour to improve the simulation accuracy. This setting ultimately yielded four sun vectors that are simulated on each point of the dataset. Meanwhile, for a proposed building, the 3D polyhedra are extruded from the plot by considering predefined criteria of the buildings and local regulations, such as a 10 m height and 4–6 m setback requirement (see Figure 5.22). Each polyhedron is set to $2 \times 2 \times 2$ m with the total number 690 units, including the generated asymmetrical shape, due to irregular order of the plot's boundary. In principle, this polyhedron's dimension can be vary depending on the architectural concept and functional program of the proposed building.



FIG. 5.22 Selected dataset and 3D polyhedra for a case study

5.10.2 Step B – Dataset preparation

While tasks (such as sun vectors calculation and 3D polyhedra generation) have been addressed previously, this section primarily discusses the result of sequential tasks, which consist of dataset correction and dataset subsampling. The dataset correction can be performed after establishing normal values with various incident angles for each selected data scan. According to Figure 5.23-A, all points are plotted on the basis of the cosine values of each incident angle. The pattern shown by all data scans is that points are distributed and assemble at a certain angle. This means that the laser beam projection may correspond very well at certain areas of the dataset during scanning, depending on the position of the scanner and geometric properties of the building surfaces (i.e., roughness). Afterward, to find the optimal normal values within this pattern, a standard deviation of cosine values is calculated based on a sample of points (see Figure 5.23-B). This becomes a basis for truncating the relevant datasets based on their distribution density.

In order to determine a threshold value of the dataset truncation, a standard deviation of the whole population of the dataset is plotted (see Figure 5.23-C). This facilitates us to generate the cosine values pattern resulting in Figure 5.23-B so that the number of points with a high distribution level of density can be identified reliably. As a result, ground areas of the plot (refer to blue-coded pattern in Figure 5.23-B) that mostly contain grass are automatically removed, due to irrelevant properties with the predefined criteria, while datasets with a red pattern are used for the intensity correction (See Figure 5.24).



FIG. 5.23 Dataset correction (A). Point distributions based on cosine values of each incident angle (B). The standard deviation of cosine values based on a group sample of points (C). The standard deviation of the whole dataset population.

Figure 5.24 specifically demonstrates the result of intensity correction from the raw attributes to the corrected one through the scalar field of intensity and intensity distribution values. The dataset transformations of each scan are plotted based on four-color steps, ranging from blue to yellow, which represent the lowest and the highest intensity values. The threshold between these values is represented by light and dark green. The trend clearly shows that the raw intensity of all data scans contains yellow areas (refers to the "before" part), representing high reflective surfaces. However, these areas predominantly assign incorrect surface materials as it should be. For example, some parts of the buildings, such as wood façade and clay roof tiles, are attributed to yellow intensity. In fact, these materials principally contain a high level of emissivity value or low intensity, which means that the return incident energy from these materials to the scanner is highly decreased, due to its

spectral reflectance mechanism on rough, dark, and dull surfaces [81]. In addition, the intensity correction also compensates for scanned areas around the scanner, due to the impact of the brightness level of intensity. These areas are massively indicated with yellow-coded values, due to the high level of intensity produced by the scanner at a very short distance. In this case, the scanner's brightness reducer is assumed not entirely and comprehensively applied in these areas, especially within the distance range of 10 m from the scanner position. This is because some local constraints, such as atmospheric variables (i.e., humidity and temperature pressures) and surface roughness of the objects, may influence specular and diffuse reflection of the laser pulse during scanning [54]

The corrected datasets are then performed with the subsampling procedure in CC in order to control the unstructured point density when it comes to the simulation. In this case, the distance between points is set to 0.05 or equal to 5 cm, which results, on average, in a decrease point, up to 70% of the truncated datasets. This procedure, however, only works on position information (XYZ) because it can cause interoperability issues when it transforms into the 3D model in Rhino, due to the different nature of the algorithmic operation. To tackle this issue, index information of each point (refers to ID in Figure 5.25) needs to be extracted beforehand during the truncation process and then used to synchronize the original attributes (i.e., XYZ and RGB) with the corrected ones in the 3D model. The workflow of this dataset transformation is illustrated further in Figure 5.25.



FIG. 5.24 Intensity correction on each data scan.



FIG. 5.25 Transformation of the dataset attributes.

In general, Figure 5.25 demonstrates a unit conversion of the dataset attributes that take place according to the step and digital tool used, starting from the raw data scan to the final stage of the workflow. A major transformation principally occurs when converting the original datasets from the 3D scanner to the 3D model in Rhino, which changes the unit of position (XYZ), color (RGB), and the raw intensity values (Ir). Therefore, the interoperability aspect plays a critical role not only to support the simulation analysis on the metadata information of the dataset (i.e., correction, truncation, and subsampling step), but also to help the visualization of geometric configuration based on the selected attributes.

5.10.3 Step C – Selection criteria

Before running the simulation, the subsampled datasets are evaluated based on the criteria of sun visibility and albedo values. These criteria identify relevant points that will be used for the voxels generation of shading envelopes. According to Figure 5.26, the resulting points for both criteria are significantly decreased as compared to the total points from the subsampled datasets. This is because a majority of subsampling points on the ground areas contain zero and negative sun vector values, which do not fulfill the sun visibility criteria. Accordingly, these points are automatically isolated, while successful ones are used as the basis datasets to perform the criteria of albedo values and the ray-tracing analysis. In general, the trend in Figure 5.26 shows that the majority of selected points both for albedo values and the visible sun are similar and overlapping in a certain spot, such as the surrounding areas of the scanner position. On the other hand, each data scan also illustrates different specific areas for point distribution. For example, points for Scan 01 are primarily detected on the edge of the building's roof, while Scan 02 and Scan 03 are partially distributed in the brick wall and clay roofs of the surrounding houses, respectively. Furthermore, the selected datasets for each data scan are then used in the simulation process. This will be presented in the following step.



FIG. 5.26 The resulting points after performing the selection criteria

5.10.4 Step D – Form generation process of shading envelopes

The resulting datasets that successfully fulfill the predefined criteria in the previous step are used in the simulation by following a series of tasks illustrated in Figure 5.19. The outputs of these tasks are demonstrated in Figure 5.27. It consists of five steps to generate volumetric shapes and three steps to simulate the ray-tracing analysis. Beforehand, each data scan is divided into two parts in order to minimize the high computational cost during the simulation. According to Figure 5.27, some analysis can be further discussed as follows:

- The ray-tracing analysis (see part RTA-01) shows the intersection lines that occur between visible sun vectors, selected points of the existing site, and 3D polyhedra. The result of these intersections is then illustrated in voxels-01, which indicate a group of voxels that fulfill the criteria of receiving direct sun access. In this case, the simulation of Scan 03 yields a lower number of voxels-01 as compared with other scans, due to the density and position of each point to the 3D polyhedra. Specifically, the existing datasets of Scan 03 approximately cover all sides of the 3D polyhedra during the simulation, while Scan 01 only intersects a certain part of the polyhedra because the existing points are predominantly distributed on one side. Conversely, the resulting voxels-02 show a contrasting number of voxels, due to the different sun access criteria.
- Voxels-03 illustrate a simulation result of shading criteria. A major trend is shown by Scan 02, which results in a significantly decreasing number of voxels from the voxels-02. This is because sun vectors originated from the reference points of voxels-02 are massively intersected with the geometric shape of voxels-01.
- Voxels-04 represent a group of polyhedra that block the direct sun access to certain areas of surrounding properties. Each data scan consists of a different voxel configuration depending on the selected areas that have been identified below the albedo values 0.3. On the other hand, voxels-05 constitute as concatenating polyhedra gathered from the resulting voxels-02, voxels-03, and voxels-04. Although the volumetric size of voxels-05 shows a significant improvement, especially for Scan 03, unfortunately, the resulting voxels can only compensate for the shading condition of the surrounding properties and still lack in protecting the proposed building from the direct sun access within a certain angle. Accordingly, a self-shading envelope workflow needs to be applied afterwards.

Steps	Scan 01		Scan 02		Scan 03				
	Α	В	A	В	A	В			
RTA - 01									
voxels 01									
total	520	4/3	332	404	209	209			
voxels 02	2 Arriver Arriver	2000							
total	170	217	338	206	421	431			
RTA - 02									
voxels 03					15				
RTA - 03									
voxels 04									
total	22	24	3/	11	59	09			
voxels 05		Mar in							
total	217	287	367	218	487	484			
RTA - 01 RTA - 02 RTA - 03 voxels 01	TA - 01 : ray tracing analysis between visible sun vectors, selected points and 3D polyhedra (TA - 02 : ray tracing analysis between voxels 01, selected vectors and reference points obtained from voxels 02 (TA - 03 : ray tracing analysis between 3D polyhedra, normal vectors and selected points gathered from material properties model : the resulting availance if share Carefords that receive sun access criteria								

voxels 01 the resulting volumetric shapes (voxels) that focus so in eccess or ineral voxels 02 the resulting volumetric shapes (voxels) that block direct sun access or surroundings voxels 03 the resulting volumetric shapes (voxels) that fulfill shading condition criteria voxels 04 the resulting volumetric shapes (voxels) that fulfill shading condition criteria gathered from albedo values

voxels 05 : the resulting volumetric shapes (voxels) after combining voxels 03, 02, 04 and removing the overlapping geometries

FIG. 5.27 The geometric configuration of subtractive shading envelopes based on sequential steps of design simulation.

5.10.5 Step E – Form generation process of self-shading envelopes

In order to perform self-shading envelopes, solar protection planes are applied on the voxels-05 by excluding voxels located on the upper part of the 3D polyhedra. This is because the upper voxels are considered as the roof or shelter for the remaining voxels located on the bottom envelopes. To do so, the upper voxels in each data scan are firstly separated from the set of 3D polyhedra (see Figure 5.28 step 02), while performing the ray-tracing analysis (see step 03) on the remaining voxels. As a result (see steps 04), voxels-06 illustrate the resulting envelopes that fulfill the criteria of a self-shading mechanism. The trend shows that the resulting voxels-06 for Scan 03 reduce, on average, 50% of the volumetric 3D polyhedra (refers to voxels-05 bottom part), while other data scans also decrease nearly a half of the initial 3D polyhedra. In this case, most of the reducing voxels are located on the edge of the 3D polyhedra or voxels that act as an exterior wall of the 3D polyhedra.

Voxels-06 of each data scan is combined with the upper part voxels that are previously separated in the previous step (refers to voxels-05 step 02). This process results in voxels-07 (refers to step 05), which are then used to perform the merging procedure in step 06—voxels-08. This step also includes elimination procedures for the voxels located in the same location. In so doing, the total number of polyhedra represented by voxels-08 shows the core geometry for each data scan. For example, the initial amount of voxels-08 after combining Scan 1A and 1B is 366 polyhedron, but after eliminating the overlapping polyhedron, it yields 259 polyhedra. Thus, 107 voxels are indicated as overlaps during the merging procedure. Afterward, the voxels-08 of each data scan are merged using the same steps in order to generate the final geometry of self-shading envelopes. Ultimately, the total number of voxels that fulfill the shading criteria for surrounding contexts and the 3D polyhedra is 514 of 690 (see Figure 5.28)

Furthermore, the resulting geometries of self-shading envelopes can be identified as a core geometry that fulfills the building's main activities based on the daylight condition. For example, architects may plot voxels located on the ground level as a communal space, workshops, and playing areas for children as it requires more open activities while the upper floor can be fulfilled with a reading room or any other activities that match with the required shading. In addition, the roof of this upper floor can be utilized as a solar collector for PV panels that may produce supplementary solar energy for the buildings. This is relevant because Indonesia's geographical condition is located in the Equator can support the electricity production around 1,534 kWh/year for each installed solar panel based on 4.5 kWh/m² average daily radiation [82].

Steps		Scan 01		Scan 02		Scan O3			
		A	В	A	В	A	В		
01	Voxels 05 bottom part								
	total	165	227	285	166	296	309		
02	Voxels 05 upper part								
ļ	total	52	60	82	52	91	75		
03	RT - 04								
04	Voxels 06								
ļ	total	96	158	195	99	119	145		
05	Voxels 07								
ļ	total	148	218	277	151	210	220		
06	Voxels 08	Jean Car							
ļ	total	259		416	•	333			
07	Voxels 09								
1	total			514					
RTA · voxe voxe voxe voxe voxe	RTA - 04 : ray tracing analysis between selected sun vectors, bottom part of the voxels 05, and reference points obtained from the upper part of voxels 05 voxels 05 - bottom part : the resulting volumetric shapes (voxels) that are used for the simulation of self-shading envelopes voxels 05 - upper part : the resulting volumetric shapes (voxels) that are used to perform solar protection plane voxels 05 : the resulting volumetric shapes (voxels) that recurse self-shading condition voxels 07 : the resulting volumetric shapes (voxels) that recurse self-shading envelopes voxels 07 : the resulting volumetric shapes (voxels) that recurse self-shading envelopes voxels 07 : the resulting volumetric shapes (voxels) after combining dan removing overlapping uvoxel for each data scan voxels 09 : the final polyhedra after combining dan removing overlapping voxels from each data scan voxels 04								



As compared to the existing method of self-shading envelopes proposed by Capeluto [36], this approach not only creates potential shading that merely comes from the roof of the envelopes, but also originates from a dynamic form of the envelope facades that correspond to different solar vectors. For these reasons, geometrical configurations of the final envelopes in this study show more variations in different orientations. In design practices, the proposed approach can be adopted by architects as a further step for designing and analyzing high performing envelopes both for new and existing development areas. The Dutch architecture and urban design firm, MVRDV, has been implemented the basic design principles of this approach into several projects based on solar-oriented designs. As a new development area, for example, the P15 Ravel Plot, which is located in the Zuidas district, 1082 LC, Amsterdam was constructed based on the idea of the optimal line of sight integrated with the three-dimensional landscape and greenery [83]. Meanwhile, in urban scale, MVRDV proposed the idea of solar energy as a part of design intervention for zero energy neighborhoods to the existing historic infrastructures in Bordeaux, France [84]. These examples clearly show a stepping-stone for architectural design practices to explore further the relevance and potential application of a voxel-based design approach in supporting sustainable environmental design. In this regard, point cloud data can be a powerful instrument that fortifies environmental performance simulation of design context during the early design phase.

5.11 Conclusion

This study investigates the potential application of attribute information stored in point cloud data to support a new computational method for a voxel-based design approach based on shading performance criteria. As a part of the contextual design strategy, the proposed workflow specifically presents form generation based on subtractive shading envelopes and material properties of existing contexts that are used to generate a new method for self-shading envelopes in tropical countries. Ultimately, this integrated approach may support architects in taking a comprehensive design decision during the early stage of the design process based on real contextual datasets. As an exploratory study, this work presents several concluding remarks as follows:

- The attribute information of point cloud data (i.e., XYZ, RGB, and reflection intensity) contributes not only to calculate material properties of existing context, but also to be a part of selection criteria for generating voxel-based subtractive shading envelopes.
- The dataset preparation includes pre-processing and correction steps that help architects minimize the environmental effects of the dataset measurement during scanning and to select reliable and relevant information for the contextual analysis and the simulation process.
- The proposed workflow enables architects to produce more variation of geometrical facades, especially related to the final geometry of self-shading envelopes that are not only depending on the roof perimeters, but also considering reference points that attach to the upper part of each generated voxel.
- The ray-tracing analysis between 3D polyhedra and selected points of surrounding contexts permits one to identify specific areas and voxels that fulfill shading performance criteria during a predefined period.
- As a contextual design approach, self-shading envelopes not only receive environmental performance responses from surrounding buildings to the proposed design, but also deliver feedback from the new building to the existing contexts.

Despite the findings mentioned above, there still some limitations. For example, first, the implementation of this proposed workflow in design practice may require a collaboration with the field of remote sensing, especially related to the dataset

collection and preparation. This is because some specific tasks need prior knowledge regarding the dataset pre-processing and particular digital tools. Second, the limitations of simulating highly dense datasets imply the number of solar vectors that need to be reduced, especially when dealing with the ray-tracing analysis procedure. Third, normal surface of the datasets and voxels of the proposed building should be evaluated by each solar vector, but as a consequence, it requires high computational processing that is currently lacking in our 3D modeling tools. Therefore, further research is expected to consider these issues in order to enhance the quality of the simulation results. It is also recommended to implement this study in different urban settings with multiple building functions, such as high-rise or new development areas, so that various urban forms can be further explored.

References

- N.-J. Shih and M.-C. Wu, "A 3D point-cloud-based verification of as-built construction progress," in *The CAAD Futures 2005 Conference*, Vienna, 2005.
- [2] A. Bashar, M. Gerke and G. Vosselman, "Visibility analysis of point cloud in close range photogrammetry," *Remote sensing and spatial information sciences*, vol. 2, no. 5, pp. 9-16, 2014.
- [3] M. Andriasyan, J. N.-J. J. E. Moyano and D. Antón, "From Point Cloud Data to Building Information Modeling: An Automatic Parametric Workflow for Heritage," *Remote sensing*, vol. 12, no. 7, pp. 1-22, 2020.
- [4] M. M. Shanoer and F. M. Abed, "Evaluate 3D laser point clouds registration for cultural heritage documentation," *The Egyptian Journal of Remote Sensing and Space Sciences*, vol. 21, no. 3, pp. 295-304, 2018.
- [5] F. Remondino, "Heritage Recording and 3D Modeling with Photogrammetry and 3D Scanning," *Remote Sensing*, vol. 3, no. 6, pp. 1104-1138, 2011.
- [6] M. A. Núñez-Andrés, F. Buill, A. Costa-Jover and M. Puche, "Structural assessment of the Roman wall and vaults of the cloister of Tarragona Cathedral," *Journal of Building Engineering*, vol. 13, no. 2017, pp. 77-86, 2017.
- [7] L. Bornaz, A. Lingua and F. Rinaudo, "Engineering and Enviroonmental Applications of Laser Scanner Techniques," in ISPRS Commission III Symposium " Photogrammetric computer vision", Graz, 2002.
- [8] R. Kassner, W. Koppe, T. Schüttenberg and G. Bareth, "Analysis of the Solar Potential of Roofs By Using Official Lidar Data," in *ISPRS Congress Beijing*, Beijing, 2008.
- [9] C. Carneiro, E. Morello and G. Desthieux, "Assessment of Solar Irradiance on the Urban Fabric for the Production of Renewable Energy using LIDAR Data and Image Processing Techniques," in Advances in GIScience, Hannover, 2009.
- [10] G. Desthieux, C. Carneiro, R. Camponovo, P. Ineichen, E. Morello, A. Boulmier, N. Abdennadher, S. Dervey and C. Ellert, "Solar Energy Potential Assessment on Rooftops and Facades in Large Built Environments Based on LiDAR Data, Image Processing, and Cloud Computing. Methodological Background, Application, and Validation in Geneva (Solar Cadaster)," *Frontiers in Built Environment*, vol. 4, no. 14, pp. 1-22, 2018.
- [11] N. N. Staneva, "Approaches for generating 3D solid models in AutoCAD and solid works," *Journal of Engineering*, vol. VI, no. 3, pp. 28-31, 2008.
- [12] V. Shapiro, "Solid modeling," Spatial Automation Laboratoty, University of Wisconsin, Madison, 2001.
- [13] J. Otepka, S. Ghuffar, C. Waldhauser and R. P. N. Hochreiter, "Georeferenced Point Clouds: A Survey of Features and Point Cloud Management," *International Journal of Geo-Information*, vol. 2, no. 4, pp. 1038-1065, 2013.
- [14] R. Richter and J. Döllner, "Concepts and techniques for integration, analysis and visualization of massive 3D point clouds," *Computers, Environment and Urban Systems*, vol. 45, pp. 114-124, 2014.

- [15] Y. Arayici, "Towards Building Information Modelling for Existing Structures," *Structural Survey*, vol. 26, no. 3, pp. 210-222, 2008.
- [16] M. F. Alkadri, M. Turrin and S. Sariyildiz, "The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts," *Advances in Computational Design*, vol. 3, no. 4, pp. 319-338, 2018.
- [17] R. L. Knowles, Energy and form: An ecological approach to urban growth, Cambridge: The MIT Press, 1974.
- [18] R. L. Knowles, Sun, rhytm and form, Massachusetts: The MIT Press, 1981.
- [19] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "Understanding Computational Methods for Solar Envelopes Based on Design Parameters, Tools, and Case Studies: A Review," *Energies*, vol. 13, no. 13, pp. 1-24, 2020.
- [20] GlobalABC, IEA and UN, "2019 Global Status Report for Buildings and Construction: Towards a zeroemission, efficient and resilient buildings and construction sector," IEA and United Nations Environment Programme, -, 2019.
- [21] C. S.S, V. Sharma and B. M. Marwah, "Review of energy efficient features in vernacular architecture for improving indoor thermal comfort conditions," *Renewable and Sustainable Energy Reviews*, vol. 65, no. -, pp. 459-477, 2016.
- [22] M. Ehzan, M. A. Farshchi and A. Ford, "Vernacular Architecture And Energy Use In Buildings: A Comparative Study," *International Journal of Advances in Mechanical and Civil Engineering (IJAMCE)*, vol. 2, no. 4, pp. 35-42, 2015.
- [23] M. Lepore, "The right to the sun in the urban design," Vitruvio, International journal of architecture and technology and sustainability, vol. 2, no. 1, pp. 24-43, 2017.
- [24] D. Mazzone, "The Dark Side of a Model Community: The 'Ghetto' of el-Lahun," *The Journal of Ancient Egyptian Architecture*, vol. 2, no. -, pp. 19-54, 2017.
- [25] K. Butti and J. Perlin, A Golden Threat 2500 Years of Solar Architecture and Technology, New York: Van Nostrand Reinhold Company, 1980.
- [26] L. Giacomo, The architecture of A. Palladio in Four Books, London: John Watts, 1715.
- [27] S. Curreli and H. Coch, "Solar access in the compact city: a study case in Barcelona," in *Passive & Low energy cooling for the built environment*, Rhode Island, 2010.
- [28] X. Wang, Z.-S. Wei and X. Pang, "Solar design in the application of the city planning," in Proceeding of International conference on Low-Carbon transportation and logistics, and green buildings, Beijing, 2012.
- [29] A. Sarkar, "Low energy urban block: morphology and urban guidelines," in *The 45th ISOCARP Congress*, Porto, 2009.
- [30] P. J. Littlefair, M. Santamouris, S. Alvarez, A. Dupagne and D. Hall, Environmental site layout planning: solar access, microclimate and passive colling in urban areas, London: Construction research communications Ltd, 2000.
- [31] M. Dekay and G. Z. Brown, Sun, Wind & Light: Architectural design strategies, 3 ed., New Jersey: John Wiley & Sons, Inc, 2014.
- [32] R. Emmanuel, "A hypothetical 'shadow umbrella' for thermal comfort enhancement in the equatorial urban outdoors," Architectural science review, vol. 36, no. 4, pp. 173-184, 1993.
- [33] M. Dekay, "Climatic urban design: configuring the urban fabric to support daylighting, passive cooling, and solar heating," *The sustainable city VII*, vol. 155, pp. 619-630, 2012.
- [34] M. Dinić and P. Mitković, "Planning regulations in the USA and their implication on urban design in the central city zone," FACTA Universitatis, Series: Architecture and Civil Engineering, vol. 9, no. 2, pp. 289-299, 2011.
- [35] S. V. Szokolay, Introduction to Architectural Science: The Basis of Sustainable Design, Oxford: Architectural Press, 2004.
- [36] G. Capeluto, "Energy performance of the self-shading building envelope," *Energy and Buildings*, vol. 35, no. 3, pp. 327-336, 2003.
- [37] A. Yezioro and E. Shaviv, "Shading: A design tool for analyzing mutual shading between buildings," Solar Energy, vol. 52, no. 1, pp. 27-37, 1994.
- [38] A. Marsh, "Computer-optimised shading design," Eindhoven, 2003.
- [39] S. Stevanović, D. Stevanović and M. Dehmer, "On optimal and near-optimal shapes of external shading of windows in apartment buildings," *PLoS ONE*, vol. 14, no. 2, pp. 1-23, 2019.
- [40] C. Kabre, "WINSHADE: A computer design tool for solar control," *Building Environment*, vol. 34, no. 3, pp. 263-274, 1999.

- [41] Y. Etzion, "An improved solar shading design tool," *Building and Environment*, vol. 27, no. 3, pp. 297-303, 1992.
- [42] J. Sargent, J. Niemasz and C. F. Reinhart, "SHADERADE: Combining rhinoceros and energyplus for the design of static exterior shading devices," Sydney, 2011.
- [43] E. Kaftan and A. Marsh, "Integrating the cellular method for shading design with a thermal simulation," Santorini, 2005.
- [44] M. Leidi and A. Schlüter, "Exploring urban space Volumetric site analysis for conceptual design in the urban context," *International journal of architectural computing*, pp. 157-182, 2013.
- [45] M. Leidi and A. Schlüter, "Volumetric insolation analysis," in CleanTech for Sustainable Buildings From Nano to Urban Scale (CISBAT 2011), Lausanne, 2011.
- [46] F. Ozel, "SolarPierce: A Solar Path-Based Generative System," in Computation and Performance Proceedings of the 31st eCAADe Conference, Delft, 2013.
- [47] J. Da Veiga and P. La Roche, "A Computer Solar Analysis Tool for the Design and Manufacturing of Complex Architectural Envelopes: EvSurf," in *Proceedings of the 6th Iberoamerican Congress of Digital Graphics* [SIGraDi 2002], Caracas, 2012.
- [48] P. Littlefair, "Passive solar urban design: ensuring the penetration of solar energy into the city," *Renewable and Sustainable Energy Reviews*, vol. 2, no. 3, pp. 303-326, 1998.
- [49] F. De Luca, "Solar form finding," in Proceeding of the 37th Annual Conference of the Association for Computer Aided Design in Architecture: Disciplines and Disruption, ACADIA, Cambridge, 2017.
- [50] F. De Luca and T. Dogan, "A novel solar envelope method based on solar ordinances for urban planning," Building simulation, vol. 12, no. 5, pp. 817-834, 2019.
- [51] F. De Luca and H. Voll, "Computational method for variable objectives and context-aware solar envelopes generation," in *The 8th annual symposium on simulation for architecture and urban design, SimAUD*, Toronto, 2017.
- [52] I. Darmon, "Voxel computational morphogenesis in urban context : proposition and analysis of rules-based generative algorithms considering solar access," in Advanced Building Skins, Bern, 2019.
- [53] S. Kaasalainen, A. K. A. Krooks and H. Kaartinen, "Radiometric Calibration of Terrestrial Laser Scanners with External Reference Targets," *Remote Sensing*, vol. 1, no. 3, pp. 144-158, 2009.
- [54] A. G. Kashani, M. J. Olsen, C. Parrish and N. Wilson, "Review of LIDAR Radiometric Processing: From Ad Hoc Intensity Correction to Rigorous Radiometric Calibration," *Sensors*, vol. 15, no. 11, pp. 28099-28128, 2015.
- [55] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "Making use of point cloud for generating subtractive solar envelopes," in eCAADe SIGraDi 2019: Architecture in the Age of the 4th Industrial Revolution, Porto, 2019.
- [56] M. F. Alkadri, F. De Luca, M. Turrin and S. Sariyildiz, "An integrated approach to subtractive solar envelopes based on attribute information from point cloud data," *Renewable and Sustainable Energy Reviews*, vol. 123, no. -, p. 109742, 2020.
- [57] Y. Fujita, Y. Hoshino, S. Ogata and I. Kobayashi, "Attribute Assignment to Point Cloud Data and Its Usage," Global Journal of Computer Science and Technology, vol. 15, no. 2, pp. 2-B, 2015.
- [58] Q. Zhan, L. Yu and Y. Liang, "A point cloud segmentation method based on vector estimation and color clustering," in *The 2nd International Conference on Information Science and Engineering (ICISE)*, Hangzhou, 2011.
- [59] I. Kobayashi, Y. Fujita, H. Sugihara and K. Yamamoto, "Attribute Analysis of Point Cloud Data with Color Information," *Journal of Japan Society of Civil Enginers*, vol. 67, no. 2, pp. 95-102, 2011.
- [60] A. K. Aijazi, P. Checchin and L. Trassoudaine, "Segmentation Based Classification of 3D Urban Point Clouds: A Super-Voxel Based Approach with Evaluation," *Remote Sensing*, vol. 5, no. 4, pp. 1624-1650, 2013.
- [61] Y. Fujita, I. Kobayashi, W. Chanseawrassamee and F. Hoshino, "Application of Attributed Road Surface Point Cloud Data in Road Maintenance," *Journal of Japan Society of Civil Engineers, Ser. F3 (Civil Engineering Informatics)*, vol. 70, no. 2, pp. I_185-I_192, 2014.
- [62] M. Tamke, I. Blümel, S. Ochman, R. Vock and R. Wessel, "From point clouds to definitions of architectural space: Potentials of automated extraction of semantic information frompoint clouds for the building profession," in *eCAADe*, Newcastle upon Tyne, 2014.
- [63] E. Turner, P. Cheng and Z. A, "Fast, Automated, Scalable Generation of Textured 3D Models of Indoor Environments," *IEEE Journal of Selected Topics in Signal Processing*, vol. 9, no. 3, pp. 409 - 421, 2014.
- [64] K. Tan and X. Cheng, "Correction of Incidence Angle and Distance Effects on TLS Intensity Data Based on Reference Targets," *Remote Sensing*, vol. 8, no. 3, pp. 1-20, 2016.

- [65] J. Shin and H. K. T. Park, "Characteristics of Laser Backscattering Intensity to Detect Frozen and Wet Surfaces on Roads," Advanced Sensor Technologies in Geospatial Sciences and Engineering in Journal of Sensors, vol. 2019, no. -, pp. 1-9, 2019.
- [66] S. Kaasalainen, H. Kaartinen and A. Kukko, "Snow cover change detection with laser scanning range and brightness measurements," *EARSeL eProceedings*, vol. 7, no. 2, pp. 133-141, 2008.
- [67] M. F. Alkadri, M. Turrin and S. Sariyildiz, "A computational workflow to analyse material properties and solar radiation of existing contexts from attribute information of point cloud data," *Building and Environment*, vol. 155, no. -, pp. 268-282, 2019.
- [68] C. Suchocki and W. Błaszczak-B, ak, "Down-Sampling of Point Clouds for the Technical Diagnostics of Buildings and Structures," *Geosciences*, vol. 9, no. 2, pp. 1-14, 2019.
- [69] J. Armesto-González, B. Riveiro-Rodríguez, D. González-Aguilera and M. T. Rivas-Brea, "Terrestrial laser scanning intensity data applied to damage detection for historical buildings," *Journal of archaeological science*, vol. 37, no. 12, pp. 3037-3047, 2010.
- [70] S. Soudarissanane, Doctoral thesis: The geometry of terrestrial laser scanning; identification of errors, modeling and mitigation of scanning geometry, Delft: Faculty of Civil Engineering and Geosciences, TU Delft, 2016.
- [71] TOPCON, "GLS-2000 Series: Multi Functional 3D Laser Scanner," TOPCON, - . [Online]. Available: https:// www.topconpositioning.com/sites/default/files/product_files/gls-2000series_broch_7010-2152_reve_ team_en_us_lores.pdf. [Accessed 6 1 2020].
- [72] Maptek, "Maptek I-Site Studio," Maptek, Denver, 2015.
- [73] C. Compare, "Cloud Compare: 3D point cloud and mesh processing software, open source project,"
 EDF R&D / TELECOM ParisTech (ENST-TSI), 2009. [Online]. Available: http://www.cloudcompare.org/.
 [Accessed 1 April 2020].
- [74] Matlab, "MathWorks," MathWorks, 2020. [Online]. Available: https://nl.mathworks.com/products/matlab. html?s_tid=hp_ff_p_matlab. [Accessed 2 4 2020].
- [75] Rhinoceros, "Rhinoceros: design, model, present, realize, analyse...," Robert McNeel & Associates, - 2020.
 [Online]. Available: https://www.rhino3d.com/. [Accessed 4 4 2020].
- [76] S. Davidson, "Grasshopper: Algorithmic modelling for rhino," Grasshopper3d, - 2020. [Online]. Available: https://www.grasshopper3d.com/. [Accessed 4 4 2020].
- [77] Ladybug, "Ladybug tools," Ladybug Tools LLC, 2020. [Online]. Available: https://www.ladybug.tools/.[Accessed 5 4 2020]
- [78] A. Boulch and R. Marlet, "Deep Learning for Robust Normal Estimation in Unstructured Point Clouds," in Eurographics Symposium on Geometry Processing, Berlin, 2016.
- [79] S. Sasidharan, "A Normalization scheme for Terrestrial LiDAR Intensity Data by Range and Incidence Angle," International Journal of Emerging Technology and Advanced Engineering, vol. 6, no. 5, pp. 322-328, 2016.
- [80] Ministry of Foreign Affairs, "Climate change profile: Indonesia," Ministry of Foreign Affairs of the Netherlands, The Hague, 2018
- [81] Optotherm, "Optotherm Thermal Imaging," Optotherm Inc. 2018. [Online]. Available: https://www. optotherm.com/emiss-examples.htm. [Accessed 9 1 2020].
- [82] H. Damayanti, F. Tumiwa and M. Citraningrum, "Residential rooftop solar: Technical and market potential in 34 provinces in Indonesia," Institute for Essential Services Reform (IESR) - Accelerating Low-Carbon Energy Transition, Jakarta, 2019.
- [83] W. Mass, "What's Next? how do we make vertical urban design?," in Council on Tall Buildings and Urban Habitat (CTBUH), Shenzen, 2016.
- [84] MVRDV, "Bastide Niel," MVRDV, 2010. [Online]. Available: https://www.mvrdv.nl/projects/46/bastide-niel-. [Accessed 3 8 2020].



6 Conclusions

6.1 Introduction

In this thesis, an integrated computational design method was developed for establishing solar geometry by making use of solar and shading performance aspects between the new building and its local contexts based on attribute information stored in point cloud data. In particular, two computational models consisting of SOLEN (Subtractive Solar Envelopes) and SHADEN (Subtractive Shading Envelopes) were developed, which were applied to temperate and tropical climates, respectively. In design practice, these models help architects to produce informed-design decisions during the conceptual design stage.

This thesis was carried out through a series of studies including a mixed-methods of deductive and inductive reasoning with quantitative and qualitative analysis, such as literature reviews to identify relevant references related to the topic, field studies for surveys and collection of raw datasets, environmental design simulations, project collaboration with an architectural design firm to demonstrate the developed method, and interviews with design practitioners (i.e., architects, urban designer, developer, city planner) to obtain constructive feedback on the proposed computational method.

Before answering each research question, this chapter will first discuss a summary of the constructive feedback obtained from design practitioners regarding the practical application of the proposed method in design practice. Afterward, comprehensive answers to sub-research questions will be presented in each chapter and will, proceed with the main research question to provide an inclusive response to the findings of the thesis. Furthermore, the limitations of the study and future recommendations will be presented to draw both scientifically and socially significant contributions to the body of knowledge in the context of the built environment.

6.2 Reflections from design practitioners

As an exploratory study, this research aimed at demonstrating the proposed computational method into design practices. This is a matter of not only identifying the level of practical application of the workflows proposed in existing design schemes, but also of gaining critical responses from architects, developers, and urban designers for future implementation. In doing so, interviews were conducted with selected participants. These participants were selected according to their scope of work, experience, and current office positions. In total, around ten participants from different specific backgrounds (i.e., developers, design consultants, urban designers, local governments) and countries (i.e., Indonesia, Qatar, The United Kingdom, The Netherlands) participated in this interview.

Furthermore, this interview was started with a brief presentation from the researcher about the research background and was followed by a booklet containing computational procedural steps of the workflow (see Appendix 6A). The booklet also includes a sample of datasets so that participants can explore the simulation steps shown. In order to drive comprehensive feedback, this study establishes nine questions (see Appendix 6B) that help participants raise critical points during the discussion.

As a result, participants generally confirmed positive feedback regarding this study, especially with regard to the potential application of 3D scanning technology that instantly enables one to convert existing contexts into 3D digital models. Despite the current complexity of processing datasets, incorporating point clouds into the architectural design process shows great interest for architects, especially related to radiometric properties stored in point cloud data. Potential features such as material properties, solar radiation simulation, and ray tracing analysis can be used to further explore the geometric envelope. These features significantly extends the main contribution of point clouds into architectural design that are predominantly used only for 3D reconstruction. The results of the interview are summarized into the following points:

Computational environments

Although the proposed model was demonstrated with clear sequential procedures and were largely based on open source tools, the main barrier for participants laid not only in the operation of the tools but also the technical knowledge of dataset processing. Given that participants mostly used different computing platforms during the design process in current working environments, technical issues such as digital interoperability between architectural tools and remote sensing equipment require some adjustments during demonstration.

Project cost

When it comes to design application, the budget aspect raises two main concerns, namely dataset preparation and design features. These aspects particularly apply the terms and conditions agreed between the architect and client. First, dataset preparation requires not only a high-cost investment to use 3D scanners but also a labor management task that usually demands specific skills to handle dataset processing and extraction of relevant information. Second, the integration of the proposed method into existing design workflows may stretch the time and cost slots for the client. In practice, this can be another trade-off between architects and clients when making design decisions for a project.

Local regulations

Although the study presented potential applications for integrating environmental performance analysis into existing design processes, unfortunately, it poses difficulties in dealing with architectural design practices due to lack of local regulations. Consequently, architectural design firms often put aside environmental parameters and focus merely on the geometric properties of design forms. For example, parameters such as shadow fences and sun access quantity were not listed yet in the Indonesian building regulations. Specific consideration regarding direct sun access in the new building in relation to surrounding contexts rests solely with the architect. That is why this aspect often causes contextual conflicts between new buildings and existing sites. To some extent, the concept of sustainable architecture simply becomes marketing jargon when bidding on a project or during the building permit appraisal process.

6.3.1 Sub-research questions

What is a state-of-the-art review of computational design methods for establishing solar envelopes? (chapter 2)

This question aimed to identify background studies, knowledge gaps, and potential developments on the topic of solar envelopes. To answer this question, a systematic review study was conducted. Several criteria were determined to narrow the relevant findings of the research scope such as selecting three topics and sub-topics (i.e., conceptual themes: solar architecture, solar envelopes, and solar access, design workflows: computational design, solar design, and solar simulation, and contextual settings: urban planning, urban design, and architectural design), establishing a search span between 1960 and 2019, and exploring three-wide citation literature databases (i.e., Web of Science, Scopus, and Google Scholar). The search results were then analyzed comparatively based on various design parameters, digital tools, and case studies taken from 58 selected references. Although the growing number of references related to the proposed topics has increased moderately over the course of the study, the research objectives focus on relevant references that specifically employ the solar envelope simulation concept. Accordingly, the studies related to solar-form findings that only investigated generative architectural forms [1] [2] [3] and solar radiation analysis [4] [5] were not further considered in the main discussion of the research.

After formulating the framework of the review, this chapter resulted in three conceptual issues that correspond to knowledge gaps and future directions of the study. First is the 3D contextual model. It is currently lacking in understanding the site characteristics information, including the necessity to present the geometric properties of the contextual model. By exploring the potential applications of 3D scanning technologies (e.g., ALS, TLS datasets) providing a reality-based dataset may bring further possibilities to mitigate missing features in current 3D site models. The second is the climatic properties. The climatic settings from the existing literature were found merely in four-season countries and were predominantly based on the cut-off times parameter. These conditions will be different when it comes to tropical climates due to different design objectives. The third is geometric

configuration. Due to limited performance criteria, the resulting geometric envelopes from the existing solar envelope's methods are also limited. Thus, exploring and integrating new features such as multi-objective optimization and multi-performance criteria (i.e., material properties, floor plan layouts) can bring a variety of geometric configurations to the final solar envelope.

Although the results of this chapter may need a quantitative evaluation analysis for future work, the resulting conceptual reviews were sufficiently robust to identify new parameters and develop new computational methods for establishing the solar envelope.

How do geometric properties of point cloud data contribute to the simulation of solar envelopes? (chapter 3)

According to the preliminary simulations of solar envelope discussed in chapter 3, the geometric properties of the point cloud collected from the ALS dataset present new features that contribute to the generation of solar envelopes, such as surrounding site elements (i.e., vegetation) and surface properties of the surrounding building contexts. These elements were used not only to generate 3D contextual models based on the existing site but also to perform environmental simulations on the final geometry of solar envelopes such as sunlight hours and insolation analysis. A comparative analysis between the manual 3D model (used by the existing solar envelope) and the TIN model (generated from the point cloud data) was examined to identify significant discrepancies between these models. In this regard, although the geometric representation of the TIN model is still far from being an ideal model that presents reality-based datasets, the simulation results of the TIN model at least contribute to capturing relevant information for further consideration in solar envelope simulations, especially with regard to parameters such as vegetation and surface morphology of the existing building.

Given that this chapter focused on exploratory studies of geometric properties stored in ALS datasets, some potential features such as material properties and high accuracy datasets [6] [7], were yet to be explored. In this regard, the potential application of TLS datasets offers a relevant match to these aforementioned features. Through the radiometric information, surface material properties of existing contexts can be further investigated.

What are potential features that can be developed further from attribute information of point cloud data? (chapter 4)

The simulation results in chapter 3 successfully presented the contribution of geometric properties stored in ALS datasets, not only to generate contextual 3D models but also to support the generation of solar envelope geometry. Further consideration of these results was carried out in chapter 4 through an exploration of geometric and radiometric information stored in TLS datasets. More specifically, a series of computational workflows was developed to conduct environmental simulations based on the point cloud of existing contexts. These workflows started with correcting the raw dataset based on the angle of incidence. It was aimed to select relevant information during site analysis and minimize erroneousness of the dataset measurement during scanning. The corrected dataset was then used to calculate material properties and perform insolation analysis. For material exploration, geometric (position-XYZ) and radiometric (color-RGB and reflection intensity-I) information of point clouds were used to calculate thermal properties (i.e., albedo, emissivity) and optical properties (i.e., reflectivity, translucency) of the dataset. These properties serve as input parameters for developing a database of surface materials from the existing site. On the other hand, insolation analysis aimed at calculating the solar energy absorbed by the site surface, especially the facade of surrounding buildings. This enables one to identify potential impacts from surrounding sites regarding solar radiation analysis.

As a final result of this chapter, material properties and insolation analysis were essential features used to develop an integrated computational workflow for contextual analysis. This workflow is essential for architects as it supports not only in revealing the specific characteristics and performances of the existing context but also in integrating the design simulation of solar envelopes based on real contextual datasets. While these features demonstrated positive results in performing environmental simulations, the list of materials stored in the developed database should be updated in order to be relevant to the design context.

How can new computational models for constructing solar geometry in temperate and tropical urban contexts integrate attribute information stored in point cloud (i.e., position-XYZ, color-RGB, and reflection intensity-I)? (chapter 5)

According to the simulation results presented in chapter 5, the attribute point cloud information through geometric and radiometric properties strongly supported the demonstration of the SOLEN and SHADEN model into case studies. These properties particularly contributed not only to dealing with geographic parameters but also to addressing features missing in existing solar envelope methods, such as sun

visibility analysis, ray-traced obstructions, and material properties of the existing context. In this regard, sun visibility was performed to identify sun vectors that are not blocked by existing buildings, ray-traced obstructions were used to capture more elements in surrounding buildings during the simulation process, and the last, material properties were used to examine surface performance behaviors of the existing context through thermal (i.e., albedo, emissivity) and optical (i.e., reflectivity, translucency) properties. In addition to these features, environmental performance aspects such as glare analysis, surface material database, and solar radiation simulation were integrated as performance criteria into both models. However, the two models used these aspects differently in computational workflows. In SOLEN models, these aspects were used to assess the potential and impact of the final geometry of subtractive solar envelopes while in SHADEN model, they were used to generate the final geometry of subtractive shading envelopes.

Regardless of the integrated environmental aspects, the computational workflow of SOLEN model in chapter 5-Part A demonstrated a comprehensive design and analysis for establishing subtractive solar envelopes. It has been successfully applied in temperate climates based on a hypothetical case study in Delft, The Netherlands. However, different climatic settings (e.g., tropical climate) of course require different adjustments, especially with regard to design objectives, methods, and local parameters. For example, typical buildings in a tropical context are predominantly designed to minimize penetration of direct sunlight access, due to high temperatures in almost all year round. Thus, instead of applying solar envelopes, shading envelopes become relevant concepts for maintaining energy consumption and building thermal conditions.

Chapter 5–Part B focuses on developing computational workflows for establishing subtractive shading envelopes suited to the context of tropical climates. The computational workflow of SHADEN model not only considers the shading performance criteria of surrounding contexts but also takes into account the shading condition of the building itself through a solar protection plane. It aims to maximize self-building protections from overheating conditions by maintaining the geometric configuration of envelope's roof areas. The SHADEN model has also been successfully demonstrated in a selected project in Bandung, Indonesia, in collaboration with an architectural firm, SHAU. This case study not only demonstrated the great feasibility of implementing a new method to subtractive shading envelopes in design practice but also presented a good starting point for a collaborative approach between architectural design and the field of remote sensing.

6.3.2 Main research-question

How can we develop a computational design method of solar geometries as a decision-making support for architects based on solar and shading envelopes and attribute point cloud information?

This research aims to address critical aspects of environmentally sustainable design by investigating a reciprocal relationship between the new building design and the local context without causing an adverse impact on the built environment. The first step started by exploring a theoretical framework of solar envelopes through a systematic review study. It aims not only to identify scopes and knowledge gaps of existing studies but also to highlight future directions of computational solar envelopes based on 3D contextual models, climatic properties, and geometric configurations. For example, existing methods for establishing solar envelopes provide an insufficient understanding of site characteristic information in the simulation model. In most cases, their 3D contextual model merely focuses on the building-oriented context so relevant parameters such as vegetation, materials, and other site properties are often neglected during the simulation process. To some extent, this can lead to a misinterpretation of simulation results and fewer comprehensive conclusions for the research problem.

The review studies simultaneously provides a theoretical underpinning for new technologies relevant to the specific gaps mentioned above, such as point cloud data gathered from 3D laser scanning technology. A preliminary simulation of solar envelopes was performed using geometric properties of point cloud based on ALS datasets. In this regard, the contextual dataset of point clouds has successfully contributed to capturing various site elements (e.g., vegetation) to support the geometric simulation model but still lacks to further identify the performance behavior of the existing context surfaces. On the other hand, attribute information stored in TLS datasets shows great relevance to deal with material properties of the existing context. Through geometric and radiometric information (i.e., XYZ, RGB, I), thermal (i.e., albedo and emissivity) and optical properties (i.e., reflectivity, translucency) of existing surface's dataset were computed in parallel with the development of environmental features such as glare analysis, surface material database, and solar radiation simulation based on existing context datasets.

As a result, two computational models for solar geometry (i.e., SOLEN, SHADEN) were developed based on the integration of the environmental performance features from the point cloud and subtractive design principles of solar envelopes. Although these models consisted of similar environmental features, they were different in terms of design objectives, computational workflow, geographic and climatic

properties, and case studies. The SOLEN model was applied in Delft, The Netherlands based on temperate climates and performance assessment features to the final geometry of subtractive solar envelopes. Meanwhile, the SHADEN model was applied in Bandung, Indonesia based on a tropical climate and performance generation features to produce the final geometry of subtractive shading envelopes.

After applying the models developed to each corresponding case study, this research confirmed positive results regarding the simulation workflow and the feasibility of applying them into architectural design practice. The proposed method of solar geometry not only allows architects to receives the environmental performance responses from surrounding buildings to the proposed design, but also delivers feedback from the new building to the existing context. In order to gain technical and conceptual feedback comprehensively, interviews with design practitioners in section 6.2 were conducted. Within this section, oral (i.e., presentation) and written descriptions of the proposed model (i.e., a booklet containing detailed computational steps) were presented in a comprehensive manner. In summary, the method developed has contributed significantly to the following particular facets:

- To better understand the existing context before starting design exploration. Thanks to the attribute information stored in the point cloud data, comprehensive site analysis can be performed based on real contextual datasets, not only identifying microclimatic conditions of the existing site but also highlighting potential impacts that may happen between new buildings and local contexts.
- To assist architects in making better design decisions during the conceptual design stage. The exploration of material properties and insolation analysis play a crucial part as design features, functioning not only as a performance evaluator assessing environmental impacts on the final geometry of solar envelopes, but also as a form generator producing the final configuration of the solar envelopes geometry.
- To organize a collaborative and interdisciplinary approach between the field of remote sensing and architectural design. This brings further opportunities for the two disciplines to explore relevant new features such as expanding the functional properties of the 3D point cloud data to contribute not only to 3D reconstruction but also to the architectural design process through environmental performance simulations.

6.4 Limitation of the research

Although the study presented a comprehensive answer to each research question and some considerations regarding the practical implementation of the proposed method mentioned in section 6.3, this research acknowledges some critical aspects that can benefit from further considerations. For example, as a collaborative research project that integrates technical knowledge from remote sensing with computational design in architecture, this study realized that dataset collection cannot be conducted and treated as individual surveys by architects. This is not only due to the technical operation of the 3D scanner but is also a matter of processing the dataset which requires a prerequisite knowledge in selecting the relevant information for the project. In addition, the tool cost affordability may also be a major constraint for many architectural design practices in using these support tools during the conceptual design stage, especially for small architectural firms.

This study also identified several limitations regarding the computational workflow of the models developed during the simulation. First, the currently proposed workflow predominantly employed an angle of incidence's (α) parameter as the main feature for the raw dataset correction, especially for the reflection intensity (I) attribute. As for the range (R) which is also included as part of acquisition geometric parameters, it is yet to be further explored due to technical preparation issues during scanning such as the reflector's placement in specific areas and the noises dataset created by the object's movement on site.

Second, the number of solar vectors used during the ray tracing analysis was on average limited due to the use of very dense datasets. As a consequence, high computation time and cost continue to cause major problems during the simulation process. This study solves this issue by selecting a small range of simulation periods with the subsampling dataset but on the other hand, this may affect the robustness of simulation results.

Third, this research mostly applied case studies based on specific building functions with simple urban forms. Therefore, the exploratory section regarding the geometric analysis between a new building and its local context was limited. In fact, the existing site located in urban areas usually consist of high and complex urban density.

With further developments to future research, the scope, objectives and research findings can be extended to explore new features and approach of solar geometries. The following aspects are highlighted according to the thesis framework:

Theoretical framework

As discussed in chapter 2, this study investigates a literature review that focuses on understanding computational methods for establishing solar envelopes based on design parameters, tools, and case studies, taking into account the selected references timeframe from 1960 to 2019. While this study has developed a new computational method, future reviews could be of further benefit to include additional methods from this study by performing quantitative analysis of various methods and using a similar set of parameters. In addition, a review that specifically investigates the shading envelope may complement a comprehensive approach database for solar envelopes.

Further developments on material properties also play a crucial role as part of the theoretical framework of this study. Chapter 4 presented a computational workflow for developing a material properties database from existing contexts, computed from attribute information of point cloud data. Additional integrated devices such as thermal cameras enable one to calibrate and verify temperature parameters when calculating reflectance values of the dataset. This will improve the identification of solar potential collection on existing datasets.

Methodological framework

The methodological section generally consists of two developed models, namely SOLEN model (see chapter 5-Part A) and SHADEN model (see chapter 5-Part B). At least three potential features were identified for future study development. First is the geometric configuration. As a voxel-based generative design, this study explored geometric shapes for solar and shading envelopes based on environmental performance criteria (i.e., direct sun access). In principle, the final geometric configuration represents the best possible compromise between the volumetric envelope and the selected environmental parameters. Since the option to select the optimized envelope was not yet available in this study, further consideration of multi-

objective optimization may allow for the generation of optimized solutions regarding geometric envelopes with the best environmental performance values. This may attract new environmental variables for future optimization research such as thermal comfort, wind, and acoustics.

Second is the ray-tracing analysis. Due to the heavy simulation dataset consuming high computational costs, this study can benefit further by investigating machine learning approaches, especially AI (Artificial Intelligence) to simulate ray tracing analysis between 3D polyhedra of the proposed building and the dataset of existing contexts. This can not only reduce the simulation time and increase the accuracy of result but also identify new configurations of performance analysis such as solar radiation and material performance behaviors.

Third are the computational design tools. This study developed an integrated computational method involving the interoperability of various digital tools and workflows from 3D laser scanning technology to environmental performance analysis. When it comes to design practice, this part becomes critical to help architects in making design decisions during the conceptual design process. Accordingly, during the reflection part, architects highlighted two specific points that may contribute to the future model of this study, namely the familiarity of design tools and the level of computational skills. Due to the lack of these factors in most common architectural practices, the development of customized design tools that specifically perform the model proposed in this study can be a potential option for future research. This tool can be components or plugins that embed into the existing digital tools environment or a new digital platform that can be used independently. In this regard, architects only need to input parameters that are relevant to their context and projects.

Case studies

This study applied the developed method into two case studies based on the context and climatic properties. The case study predominantly focuses on specific urban forms and building functions. By exploring more urban properties in densely populated areas with a variety of existing building functions enables one to understand in depth the complexities of existing urban forms in relation to new buildings. In addition, incorporating functional utility into the shading envelope simulation, especially during the ray tracing analysis makes it possible to integrate the geometric shape of each resulting voxel with a specific function meeting predefined criteria. This will significantly help architects to configure the 3D space layout of a new building based on the environmental performance simulations generated in the early phase of architectural design.

References

- [1] M. Leidi and A. Schlüter, "Exploring urban space Volumetric site analysis for conceptual design in the urban context," *International journal of architectural computing*, pp. 157–182, 2013.
- [2] J. Da Veiga and P. La Roche, "A Computer Solar Analysis Tool for the Design and Manufacturing of Complex Architectural Envelopes: EvSurf," in *Proceedings of the 6th Iberoamerican Congress of Digital Graphics* [SIGraDi 2002], Caracas, 2012.
- [3] P. Littlefair, "Passive solar urban design: ensuring the penetration of solar energy into the city," *Renewable and Sustainable Energy Reviews*, vol. 2, no. 3, pp. 303-326, 1998.
- [4] S. Freita, C. M. Catita, P. Redweik and M. Brito, "Modelling solar potential in the urban environment: State-ofthe-art-review," *Renewable and sustainable energy reviews*, vol. 41, no. -, pp. 915-931, 2015.
- [5] G. Lobaccaro, F. Frontini, G. Masera and T. Poli, "SolarPW: A new solar design tool to exploit solar potential in existing urban areas," *Energy Procedia*, vol. 30, pp. 1173-1183, 2012.
- [6] M. E. Charlton, S. J. Coveney and T. McCarthy, "Issues in Laser Scanning," in Laser Scanning for the Environmental Sciences, West Sussex, Wiley-Blackwell, 2009, pp. 35-48.
- [7] S. D. Bartolo and R. Salvini, "Multitemporal Terrestrial Laser Scanning for Marble Extraction Assessment in an Underground Quarry of the Apuan Alps (Italy)," *Sensors*, vol. 19, no. 3, pp. 1-10, 2019.

Appendix 6A

- 1 What is your educational and professional background?
- 2 What is your current position at the office?
- 3 What are your computational skills? What computational tools do you use in your daily practice?
- 4 How do you deal with performance-based architecture during the conceptual design process?
- 5 What do you think about the interdisciplinary (i.e., engineering and architecture) design approach to design practice?
- 6 To what extent can the presented method be adopted in your daily practice?
- 7 What are the pros and cons of using the method presented?
- 8 What are the future recommendations for the method presented ?

Appendix 6B

Procedural steps of the computational workflow

Step 1



Step 2



Step 3



Step 4



Step 4b



Step 5



Step 6



Step 7



Step 8


Step 9



Step 10



Step 11



Step 12



Step 13



Step 14







Step 13

Combine the envelopes generated from the upper floor and self-shading envelopes (step 12). This geometry becomes the final envelopes for shading envelopes

Curriculum Vitae



Miktha Farid Alkadri

Date of Birth: 24 September 1990 Place of Birth: Sinjai, Indonesia Email: mikthafarid@gmail.com

Education

09.2008 – 07.2012	Bachelor of Architecture University of Indonesia, Indonesia
09.2012 – 07.2013	Master of Architecture University of Indonesia, Indonesia

Professional Experience

02.2017 - 02.2018	Researcher "Re-printing Architectural Heritage" Selected grants from Light House projects 2017 – 4TU Bouw
10.2014 – 02.2015	Junior Architect SHAU Rotterdam, the Netherlands
03.2013 – 08.2014	Junior Architect PT Relife, Indonesia
07.2012 – 07.2013	Teaching and Research Assistant Department of Architecture, University of Indonesia, Indonesia

Selected Academic Activities

Since 02.2021	Reviewer in Journal of Building Research and Information
Since 10.2020	Reviewer in CAADRIA – PROJECTIONS 2021 The Chinese University of Hong Kong
Since 05.2020	Reviewer in Journal of Renewable and Sustainable Energy Reviews
Since 05.2020	Reviewer in Journal of Information Processing in Agriculture
08.2019	Workshop Coordinator Digital fabrication: Robotic weaving using bio-composites materials University of Indonesia, Indonesia
11.2019	Presenter In International LDE-Heritage Conference Delft, the Netherlands
01.2019	Presenter In Conference Child in the City, Children in the Sustainable City Antwerp, Belgium
01.2018	Workshop Organizer Computational method of solar envelopes and solar collection Chair of Design Informatics, TU Delft

List of Publications

Journal papers

Alkadri, M. F., De Luca, F., Turrin, M. and Sariyildiz, S., 2020. A computational workflow for generating a voxel-based design approach based on subtractive shading envelopes and attribute information of poitcloud data. *Remote sensing*, 12(16), pp. 2561

Alkadri, M. F., De Luca, F., Turrin, M. and Sariyildiz, S., 2020. Understanding computational design methods of solar envelopes based on design parameters, tools, and case studies. *Energies*, 13(13), pp. 3302

Alkadri, M. F., De Luca, F., Turrin, M. and Sariyildiz, S., 2020. An integrated approach to subtractive solar envelopes based on attribute information from point cloud data. *Renewable and Sustainable Energy Reviews,* 123, pp. 109742

Alkadri, M.F., Turrin, M. and Sariyildiz, S., 2019. A computational workflow to analyze material properties and solar radiation of existing contexts from attribute information of point cloud data. *Building and Environment*, 155, pp. 268-282

Koorstra, P., Reinstra, A., Dellebeke, A., Vanhove, D., Vlasblom, D., Bekooy, J., Teew, R., Vanhecke, V., Oostveen, W., 2019. Re-printing architectural heritage: Exploring current 3D printing and scanning technologies. **Spool**, 6(2), pp. 33-36

Alkadri, M. F., Turrin, M. and Sariyildiz, S., 2018. The use and potential applications of point clouds in simulation of solar radiation for solar access in urban contexts. *Advances in computational design*, 3(4), pp. 319-338

Conference papers

Alkadri, M.F., Turrin, De Luca, F., Turrin, M. and Sariyildiz, S., 2019. Making use of point cloud for generating subtractive solar envelopes. *The eCAADe* +*SIGraDi* 2019, pp. 633-640

Alkadri, M.F., Turrin, M., and Sariyildiz, S., 2018. Identifying the surface materials of existing environments through point cloud data attributes. *Proceeding of the Symposium for Architecture and Urban Design (SimAUD 2018)*, pp. 323-330

Alkadri, M.F., De Luca, F., Turrin, M. and Sariyildiz, S., 2018. Toward an environmental database. *Proceedings of the 36*th *eCAADe Conference*, pp. 263-270

Alkadri, M.F., Turrin, M. and Sariyildiz, S., 2017. Investigating solar envelopes by making use of point cloud data. *Proceedings of the 24th EG-ICE International Workshop on Intelligent Computing in Engineering (EG-ICE 2017)*, pp. 106-115

21#13

Solar Geometry in Performance of the Built Environment

An Integrated Computational Design Method for High-Performance Building Massing Based on Attribute Point Cloud Information

Miktha Farid Alkadri

As part of the passive design strategy, the development of computational solar envelopes plays a major role to construct a cooperative environmental performance exchange between new buildings and their local contexts. However, the state-of-the-art computational solar envelopes pose a great challenge in understanding site characteristics from a given context. Existing methods predominantly construct 3D context models based on basic architectural geometric shapes, which are often isolated from the surrounding properties of local contexts (i.e., vegetation, materials). Thus, they only focus on context-oriented buildings and energy quantities that unfortunately lack a contextual solar performance analysis. It is clear that this condition may result in a fragmented understanding of the local context during the design and simulation process.

With the potential application of attribute point cloud information, it is necessary to consider relevant parameters such as surface and material properties of existing contexts during the simulation of solar geometries, which are currently absent in computational frameworks. As such, the new method is required to enable architects not only to measure specific performances of the local context but also to identify vulnerable areas that may affect the proposed design. This research focuses on exploring an integrated computational design method for solar geometry based on solar and shading envelopes, and geometric and radiometric information from point cloud data. In particular, two computational models consisting of SOLEN (Subtractive Solar Envelopes) and SHADEN (Subtractive Shading Envelopes) are developed, which are applied to temperate and tropical climates, respectively. In design practice, these models help architects to produce informed-design decisions towards high-performed building massing.

A+BE | Architecture and the Built Environment | TU Delft BK